SPECTRAL ANALYSIS OF MORSE-SMALE GRADIENT FLOWS

NGUYEN VIET DANG AND GABRIEL RIVIÈRE

ABSTRACT. On a smooth, compact and oriented manifold without boundary, we give a complete description of the correlation function of a Morse-Smale gradient flow satisfying a certain nonresonance assumption. This is done by analyzing precisely the spectrum of the generator of such a flow acting on certain anisotropic spaces of currents. In particular, we prove that this dynamical spectrum is given by linear combinations with integer coefficients of the Lyapunov exponents at the critical points of the Morse function. Via this spectral analysis and in analogy with Hodge-de Rham theory, we give an interpretation of the Morse complex as the image of the de Rham complex under the spectral projector on the kernel of the generator of the flow. This allows us to recover classical results from differential topology such as the Morse inequalities and Poincaré duality.

1. Introduction

Consider a smooth (\mathcal{C}^{∞}) flow $(\varphi^t)_{t\in\mathbb{R}}$ acting on a smooth, compact, oriented manifold M which has no boundary and which is of dimension $n \geq 1$. A natural question to raise is whether the limit

$$\lim_{t \to +\infty} \varphi^{-t*}(\psi)$$

exists for any smooth function ψ defined on M. This is of course very unlikely to happen in general, and a natural setting where one may expect some convergence is the class of dynamical systems with hyperbolic behaviour and for a nice enough reference measure. For instance, if φ^t is a topologically transitive Anosov flow [1] and if we study the weak limit with respect to a so-called Gibbs measure, it is known from the works of Bowen, Ruelle and Sinai that such a limit exists and is equal to the average of ψ with respect to the Gibbs measure¹ [67, 13]. If one is able to show that this equilibrium state exists, a second natural question to raise is: can one describe the fluctuations? For instance, what is the rate of convergence to this state?

These problems are naturally related to the study of the operator generating the flow:

$$\mathcal{L}: \psi \in \mathcal{C}^{\infty}(M) \mapsto -\frac{d}{dt} \left(\varphi^{-t*}(\psi) \right) |_{t=0} \in \mathcal{C}^{\infty}(M).$$

Note that, by duality, this operator acts on the space of distributions $\mathcal{D}'(M)$. In recent years, many progresses have been made in the study of such operators acting on suitable Banach spaces of distributions when the flow φ^t enjoys the Anosov property. In [55], Liverani defined Banach spaces of distributions with "anisotropic Hölder regularity" for

¹Recall that a well-known example is the Liouville measure for the geodesic flow on a negatively curved manifold.

which he could make a precise spectral analysis of \mathcal{L} . For instance, he proved the existence of a spectral gap for contact Anosov flows from which one can deduce that, for every $t \geq 0$ and for every ψ_1, ψ_2 in $\mathcal{C}^{\infty}(M)$,

$$(1) \qquad C_{\psi_1,\psi_2}(t):=\int_{M}\varphi^{-t*}(\psi_1)\psi_2d\mathrm{vol}_g=\int_{M}\psi_1d\mathrm{vol}_g\int_{M}\psi_2d\mathrm{vol}_g+\mathcal{O}_{\psi_1,\psi_2}(e^{-\Lambda t}),$$

where $\Lambda > 0$ is some fixed positive constant related to the spectral gap of \mathcal{L} and where vol_q is the Riemannian volume. His construction was inspired by similar results for diffeomorphisms [11] and by a proof of Dolgopyat which holds in the 2-dimensional case [25]. Introducing Banach spaces inside $\mathcal{D}'(M)$ contrasts with earlier approaches to these questions where symbolic coding of Anosov flows was used to describe the weak convergence of $\varphi^{-t*}(\psi)$. For more general Anosov flows, Butterley and Liverani also showed how this direct approach allows to make a meromorphic extension for the Laplace transform of the correlation function $C_{\psi_1,\psi_2}(t)$ to the entire half plane [19]. This extended earlier works of Pollicott [62] and Ruelle [64] which were also based on the use of symbolic dynamics. Such poles (and their corresponding eigenstates) describe in some sense the fine structure of the long time dynamics and are often called *Pollicott-Ruelle resonances*. Pushing further this direct approach [40], Giulietti, Liverani and Pollicott extended this spectral analysis to anisotropic spaces of currents and they proved that, for any smooth Anosov flow, the Ruelle zeta function has a meromorphic extension to \mathbb{C} . In the case of Anosov geodesic flows satisfying certain pinching assumptions, they also showed that (1) also holds for the Bowen-Margulis measure (and not only with respect to the Riemannian volume).

In parallel to this approach via spaces of anisotropic Hölder distributions, it was observed that the spectral analysis of Anosov flows can in fact be understood as a semiclassical problem which fits naturally in the theory of semiclassical resonances [48, 29]. Building on earlier works for Anosov diffeomorphisms by Baladi-Tsujii [6, 7] and Faure-Roy-Sjöstrand [31] involving microlocal tools, this kind of approach to Pollicott-Ruelle resonances was initiated for Anosov flows by Tsujii [73, 74], Faure-Sjöstrand [32] and Faure-Tsujii [34, 33]. Thanks to this microlocal approach, several new results or alternative proofs were obtained in the last few years: upper bounds on the counting function in the high frequency limit near the imaginary axis [32, 23], explicit bounds on the size of the spectral gap [73, 74, 61, 33], meromorphic extension of the Ruelle zeta function [28], band structure of the resonances [34, 33, 26]. We refer to the survey article of Gouëzel [42] for a recent account on these progresses.

Regarding the important steps made in the Anosov case, it is natural to understand to what extent these methods can be adapted to more general dynamical systems satisfying weaker chaotic features. A natural extension to consider is the class of hyperbolic flows with discontinuities. Results in this direction were obtained recently by Baladi-Liverani [5] and Baladi-Demers-Liverani [4]. For instance, they proved exponential decay of correlations for the 2-dimensional finite horizon Sinai billiard flow [4]. We also refer to [71] for results on open billiard flows and to [24] for billiard maps. Another natural direction is to consider the case of open systems which was recently studied by Dyatlov and Guillarmou via microlocal techniques [27] – see also [2, 35] for similar problems concerning expanding maps. Their

framework is in fact related to the so-called Axiom A flows [70, 13]. In particular, they show that Pollicott-Ruelle resonances can be defined *locally* on a small neighborhood of any basic set of a given Axiom A flow with no critical points – see also [64, 6, 7, 41] for earlier results in the case of Axiom A diffeomorphisms.

1.1. Morse-Smale gradient flows and a question from Bowen's notebook. Given a Morse function f and a Riemannian metric g, one can define a gradient flow φ_f^t and hence give the simplest example of an Axiom A flow in the sense of Smale [70, p. 803]. Such dynamical systems of course do not satisfy the Anosov property and hyperbolicity is in some sense concentrated at the critical points of the Morse function. In his notebook [14, Problem 1], Bowen raised the following natural question concerning these flows: To what extent does the gradient flow near a critical point depend on the metric?

In the present article, we aim at extending the microlocal approach of Faure-Sjöstrand-Tsujii for this kind of dynamical systems. As a consequence, it will allow us to give some insights on that question as we will obtain a rather complete description of the asymptotics of the global "correlation function" for generic gradient flows. More precisely, we can in some sense track the dependence on the metric in the terms of our asymptotic expansion. For instance, our analysis will among other things show that all the decay rates will be given by integer linear combinations of the Lyapunov exponents which depend only on the "germ" of the metric at the critical points of f – see equation (3) below.

Near the critical points of the Morse function, this kind of dynamical system has in fact a very simple structure and it behaves like the gradient flow induced by a hyperbolic matrix acting on \mathbb{R}^n thanks to the Grobman-Hartman Theorem [44]. We note that hyperbolic linear models on \mathbb{R}^n were already analyzed via microlocal techniques by Faure and Tsujii in [34, Ch. 3-4] where it also served as a local picture for their global model. Here we will make use of similar ideas and one of the main task of the present article will be to understand how these local models can be patched together in our dynamical framework in order to give a global description of the dynamics. As we shall see, it will allow us to give a rather precise description of the spectrum of \mathcal{L} and of the long time dynamics of the flow φ_f^t acting on smooth differential forms of any degree.

1.2. From dynamics to differential topology. Recall that gradient flows associated with a Morse function are also interesting because of their deep connections with differential topology which first appeared in the pioneering works of Thom [72] and Smale [68, 69]. Our microlocal approach to the properties of gradient flows will allow among other things to give new (spectral) interpretations of some results of Laudenbach [53, 54] and Harvey-Lawson [45, 46] and to recover some classical facts from differential topology such as the finiteness of the Betti numbers, the Morse inequalities and Poincaré duality. The works of Witten [76], Helffer-Sjöstrand [47], Bismut-Zhang [10, 77], Burghelea et al. [17, 16, 18] developed the relationship between Morse theory, Hodge theory and Ray-Singer analytic torsion. For an introduction to this spectral approach to Morse theory, we refer the reader to the survey of Henniart [49]. To compare with our results, recall that Witten's approach applies for instance to the operator $d_s = d + si_V$ where s > 0, d is the usual coboundary

operator and i_V is the contraction operator induced by the vector field V of φ^t . More precisely, he studied the *elliptic* Hamiltonian $(d_s + d_s^*)^2$ [76, Sect. 3] and gave, in the limit $s \to +\infty$, a topological interpretation of its kernel in terms of the zero set of V. Here, instead of $(d_s + d_s^*)^2$, we will study the operator $s\mathcal{L} = d_s^2$ in the specific case of Morse-Smale vector fields. We will show that, even if \mathcal{L} is not elliptic, its spectrum on convenient Sobolev spaces has a very nice structure which also carries some topological informations on the manifold. For a more detailed exposition on these relations between dynamical systems, topology and spectral theory, we refer to the classical survey article of Bott [12].

2. Statement of the main results

2.1. **Dynamical framework.** We fix f to be a smooth (\mathcal{C}^{∞}) Morse function, meaning that f has only finitely many critical points and that these points are non degenerate. We denote by $\operatorname{Crit}(f)$ the set of critical points. For simplicity, we shall always assume that f is excellent in the sense that, given $a \neq b$ in $\operatorname{Crit}(f)$, one has $f(a) \neq f(b)$. If we consider a smooth Riemannian metric g on M, we can define a vector field V_f as follows

(2)
$$\forall (x, v) \in TM, \ d_x f(v) = \langle V_f(x), v \rangle_{g(x)}.$$

This vector field generates a complete flow on M [54, Ch. 6] that we denote by φ_f^t . Given any point a in Crit(f), we can define its stable (resp. unstable) manifold, i.e.

$$W^{s/u}(a) := \left\{ x \in M : \lim_{t \to +/-\infty} \varphi_f^t(x) = a \right\}.$$

One can show that $W^s(a)$ (resp. $W^u(a)$) is an embedded submanifold in M of dimension $0 \le r \le n$ (resp. n-r) where r is the index of the critical point [75]. Note also that $W^u(a) \cap W^s(a) = \{a\}$. A remarkable property of these submanifolds is that they form a partition of the manifold M [72], i.e.

$$M = \bigcup_{a \in \operatorname{Crit}(f)} W^s(a), \text{ and } \forall a \neq b, \ W^s(a) \cap W^s(b) = \emptyset.$$

The above property also holds true for the unstable manifolds. This partition in stable (and unstable) leaves will play a central role in our analysis. Among these Morse gradient flows, Smale introduced a particular family of flows [68]. Namely, given any a and b in Crit(f), he required that $W^s(a)$ and $W^u(b)$ intersect transversally whenever they intersect. This assumption also turns out to be a crucial ingredient to make our proofs work. We will use the terminology Morse-Smale for any gradient flow enjoying the above properties. Finally, for any point a in Crit(f), we define $L_f(a)$ as the only matrix satisfying

(3)
$$\forall \xi, \eta \in T_a M, \ d_a^2 f(\xi, \eta) = g_a(L_f(a)\xi, \eta).$$

As a is a nondegenerate critical point, $L_f(a)$ is symmetric with respect to g_a and invertible. Its eigenvalues are called the **Lyapunov exponents** at the point a and we write them as

$$\chi_1(a) \le \ldots \le \chi_r(a) < 0 < \chi_{r+1}(a) \le \ldots \le \chi_n(a),$$

where r is the index of the critical point a. All along the article, we will often make the assumption that (f,g) is a *smooth Morse pair* inducing a Morse-Smale gradient flow. By

smooth Morse pair, we roughly mean that there is a smooth linearizing chart for V_f near any critical point. By the Sternberg-Chen Theorem [20], this is for instance satisfied when, for every critical point a, the Lyapunov exponents $(\chi_j(a))_{1 \leq j \leq n}$ are rationally independent - see paragraph 3.5 for more details.

- Remark 2.1. Let us fix some conventions. We will denote by \mathbb{N}^* the set of positive integers $\{1,2,\ldots\}$ while N will be the set of nonnegative integers $\{0,1,2,\ldots\}$. We will use $\alpha=$ $(\alpha_1,\ldots,\alpha_n)$ for a multi-index in \mathbb{N}^n . Given any critical point a of f, we denote by $|\chi(a)|$ the vector $(|\chi_1(a)|, \dots, |\chi_n(a)|)$. For any $0 \le k \le n$, $\Omega^k(M)$ will be the space of smooth differential forms of degree k and $\mathcal{D}^{\prime,k}(M)$ will be the topological dual of $\Omega^{n-k}(M)$, i.e. the space of currents of degree k (or of dimension n-k). For an introduction to the theory of currents, we refer to [63, 66].
- 2.2. Correlation function. The main concern of the article will be to perform a spectral analysis of the operator \mathcal{L}_{V_f} acting on appropriate spaces of currents. As an application of our analysis, we will prove the following result on the asymptotic behaviour of $\varphi_f^{-t*}(\psi)$:

Theorem 2.2. Let φ_f^t be a Morse-Smale gradient flow all of whose Lyapunov exponents are rationally independent. Let $0 \le k \le n$.

Then, for every a in Crit(f) and for every α in \mathbb{N}^n , there exists a continuous linear operator,

$$\pi_{a,k}^{(\alpha)}: \Omega^k(M) \to \mathcal{D}'^{,k}(M),$$

such that for every $\Lambda > 0$, for every ψ_1 in $\Omega^k(M)$, for every ψ_2 in $\Omega^{n-k}(M)$, and for every $t \geq 0$,

$$\int_{M} \varphi_{f}^{-t*}(\psi_{1}) \wedge \psi_{2} = \sum_{a \in \operatorname{Crit}(f)} \sum_{\alpha \in \mathbb{N}^{n}: \alpha. |\chi(a)| < \Lambda} e^{-t\alpha. |\chi(a)|} \int_{M} \pi_{a,k}^{(\alpha)}(\psi_{1}) \wedge \psi_{2} + \mathcal{O}_{\psi_{1},\psi_{2}}(e^{-\Lambda t})$$

where $\alpha.|\chi(a)| = \sum_{i=1}^{n} \alpha_i |\chi_i(a)|$. Moreover, for every a in Crit(f) and for every α in \mathbb{N}^n , one has²

- for every ψ_1 in $\Omega^k(M)$, the support of $\pi_{a,k}^{(\alpha)}(\psi_1)$ is contained in $\overline{W^u(a)}$,
- $0 \le \operatorname{rk}(\pi_{a,k}^{(\alpha)}) \le 2^n$,
- $\operatorname{rk}(\pi_{a,k}^{(0)}) = \delta_{k,r}$ where r is the index of a, $\operatorname{rk}(\pi_{a,k}^{(\alpha)}) = \frac{n!}{k!(n-k)!}$ for every $\alpha \in (\mathbb{N}^*)^n$.

This Theorem gives us an asymptotic expansion at any order of the correlation function³ associated with a Morse-Smale gradient flow and it is a consequence of Proposition 7.8 which is in fact slightly more precise. As we shall see in this Proposition, we will also provide a more or less explicit expression of the operator $\pi_{a,k}^{(\alpha)}$ near the critical point a. These operators of course depend on the choice of the Riemannian metric used to define

²We will in fact give a (rather combinatorial) explicit expression of $rk(\pi_{a,k}^{(\alpha)})$ in the proof – see Remark 7.6.

³Here we make a small abuse of terminology as correlation functions are usually concerned with invariant measures.

the gradient flow as well as the Lyapunov exponents appearing in the above asymptotic expansion. In that sense, this result gives a partial answer to the question mentionned in the introduction concerning the dependence of gradient flows on the underlying metric [14, Problem 1].

Each term appearing in the sum looks also very much like the expansion obtained by Faure and Tsujii in the case of linear models acting on \mathbb{R}^n [34, Ch. 3-4]. Here, one of the main difficulty will be to understand how these local models can be glued together in order to obtain a result valid on the whole manifold. We also mention that similar expansions appear via techniques from complex analysis in the case of analytic expanding circle maps arising from finite Blaschke products [8, 9]. Even if our flows are in some sense degenerate Axiom A flows, this result is also closely related to the recent results of Dyatlov and Guillarmou on Pollicott-Ruelle resonances for open systems [27]. Maybe the main difference with this reference is that Theorem 2.2 holds globally on M and not only in a neighborhood of the critical points⁴. If we consider the time 1 map of the flow, the induced diffeomorphism $h = \varphi_f^{-1}$ is probably one of the simplest example of an Axiom A (but not Anosov) diffeomorphism [70], even if it is a kind of "trivial" example as all the basic sets are reduced to fixed points. Resonances of general Axiom A diffeomorphisms were studied by Ruelle in [65] via methods of symbolic dynamics while the direct functional approach of these dynamical systems was developed by Baladi-Tsujii [6, 7] and by Gouëzel-Liverani [41]. As for the case of Axiom A flows treated in [27], these results focused on the dynamics near a convenient basic set. In our case, it would correspond to restrict ourselves to ψ_1 and ψ_2 supported in a neighborhood of a fixed critical point. Here, due to the simple structure of the diffeomorphism, the asymptotic expansion of the correlation function can be made in the entire complex plane without restrictions on ψ_1 and ψ_2 and with discrete poles on the real axis which are completely determined by the Lyapunov exponents on the basic sets of the diffeomorphism (i.e. the critical points of f). Moreover, the corresponding resonant states can also be made explicit.

We made a kind of (global) nonresonance assumption in the statement of Theorem 2.2. This assumption ensures that the generator of the flow does not have any Jordan blocks. Regarding this point, we should mention that, in [41, p.436], Gouëzel and Liverani raised the question of the existence of Jordan blocks for nontopologically transitive Axiom A diffeomorphisms. As far as we know, the above Theorem gives the first example of such diffeomorphisms where all the Jordan blocks in the spectrum are trivial. In [36], Frenkel, Losev and Nekrasov were lead to similar problems in the context of quantum field theory. In particular, as a by product of their analysis, they obtain the complete asymptotic for the correlation function of the flow associated with the height function on the 2-sphere endowed with its canonical metric. In that case, the Lyapunov exponents are all equal to ± 1 (hence resonant) and they prove that there are indeed infinitely many polynomial factors (hence non trivial Jordan blocks) in the asymptotic expansion. In the general case (including the case of [36]), we still obtain an asymptotic expansion for the correlation function which may involve polynomial factors in t except for the peripheral eigenvalue

⁴We also note that [27] made the assumption that the vector field does not vanish.

 $\lambda = 0$. We refer to paragraph 5.3 for a complete expression in the general setting. We just mention here the *leading term of the asymptotics* in the general case:

Theorem 2.3. Let (f,g) be a smooth Morse pair generating a Morse-Smale gradient flow φ_f^t . Let $0 \le k \le n$.

Then, for every a in Crit(f) of index k, there exist

- U_a in $\mathcal{D}^{\prime,k}(M)$ whose support is equal to $\overline{W^u(a)}$,
- S_a in $\mathcal{D}^{\prime,n-k}(M)$ whose support is equal to $\overline{W^s(a)}$,

such that, for every

$$0 < \Lambda < \min \{ |\chi_j(b)| : 1 \le j \le n, b \in \operatorname{Crit}(f) \},$$

for every (ψ_1, ψ_2) in $\Omega^k(M) \times \Omega^{n-k}(M)$, and for every $t \geq 0$,

$$\int_{M} \varphi_f^{-t*}(\psi_1) \wedge \psi_2 = \sum_{a \in \operatorname{Crit}(f): \operatorname{ind}(a) = k} \int_{M} S_a \wedge \psi_1 \int_{M} U_a \wedge \psi_2 + \mathcal{O}_{\psi_1, \psi_2}(e^{-\Lambda t}).$$

Except for the remainder term, this result was first proved by Harvey and Lawson via techniques from geometric measure theory and under slightly more restrictive assumptions on the gradient flows [45, 46]. This Theorem follows from Proposition 6.9 which is again slightly more precise. To our knowledge, the currents U_a and S_a appearing in the leading term of the asymptotic expansion were first constructed by Laudenbach [53, 54] in the case of a "locally flat metric adapted to the Morse coordinates" - see paragraph 3.5 for the definition. In the following, we shall refer to them as Laudenbach's currents. The difficulty is that the submanifolds $W^u(a)$ and $W^s(a)$ are not a priori properly embedded and one has to justify that the currents of integration are well defined. Precisely, one can integrate on $W^{u}(a)$ a differential form ψ whose support is included in a compact part of $W^{u}(a)$ but integration of a general form whose support may intersect the boundary needs to be justified. This can solved by analyzing the mass of the currents near the boundary of the unstable (resp. stable manifold) and this requires a careful description of the structure of the boundary of $W^{u}(a)$ [53, 54]. Even if it is in a different manner, similar difficulties involving the boundary will of course occur at some point in our analysis and we shall deal with this problem via dynamical techniques following the works of Smale [68] and Weber [75] – see for instance Lemmas 3.6 and 3.8.

After properly defining the spectral framework of our problem, we will recover the existence of these currents as a consequence of our spectral analysis. They correspond to the kernel of the operator \mathcal{L}_{V_f} acting on appropriate anisotropic spaces of currents. The advantage of this approach is that it allows to treat more general families of gradient flows and that it sheds a new (spectral) light on these natural dynamical objects. A difficulty may be that it relies on microlocal techniques which are maybe not as well-known as the geometric measure theory used by Harvey and Lawson in [45, 46] to give an interpretation of these currents as a limit of the correlation function under the assumption of finite volume – see also [58] for generalizations of this result. In any case, Theorem 2.2 generalizes this type of result in the sense that it does not only give the existence of the limit but

also a rate of convergence to this equilibrium state and the full asymptotic expansion as $t \to +\infty$.

2.3. Topological interpretation of the leading term. One of the main application of this dynamical approach to Morse theory is that the partition of the manifold into unstable components has beautiful topological implications [72, 68, 46, 54]. In section 8, we will explain how to recover some classical results from differential topology (e.g. finiteness of Betti numbers, Poincaré duality, Morse inequalities) via our spectral approach and via some analogies with Hodge-de Rham theory. For that purpose, we can set, for every $0 \le k \le n$,

$$C^{k}(f) := \operatorname{span} \{U_{a} : \operatorname{ind}(a) = k\}.$$

In analogy with Hodge-de Rham theory where one uses the formula $\Delta = d \circ d^* + d^* \circ d$, we can write the Cartan formula

$$\mathcal{L}_{V_f} = d \circ i_{V_f} + i_{V_f} \circ d,$$

where d is the coboundary operator and i_{V_f} is the contraction by the vector field V_f . We will verify that $(C^*(f), d)$ induces a cohomological complex while $(C^*(f), i_{V_f})$ induces a homological complex. The first complex is known as the Morse complex (also sometimes called the Thom-Smale-Witten complex) and we will call the second one the **Morse-Koszul** complex. Using our analysis of Morse-Smale gradient flows, we will give a purely spectral proof of the following results:

Theorem 2.4. Let (f,g) be a smooth Morse pair generating a Morse-Smale gradient flow φ_f^t . Then, the following holds.

(1) The maps

$$\mathbb{P}^{(k)}: \psi \in \Omega^k(M) \mapsto \sum_{a \in \operatorname{Crit}(f): \operatorname{ind}(a) = k} \left(\int_M S_a \wedge \psi \right) U_a \in C^k(f)$$

induce a quasi-isomorphism between the cohomology of the de Rham complex $(\Omega^*(M), d)$ and the cohomology of the Morse complex $(C^*(f), d)$.

(2) The Euler characteristic of M is equal to the Euler characteristic of the Morse-Koszul complex $(C^*(f), i_{V_f})$.

The first part of the Theorem is due to Laudenbach in the case of a "locally flat metric adapted to the Morse coordinates" [53, 54]. It recovers the classical fact that the de Rham complex $(\Omega^*(M), d)$ is quasi-isomorphic to the Morse complex. From this, it is classical to deduce the finiteness of the Betti numbers and the so-called Morse inequalities – see section 8 for more details. The second part seems new and it shows that the Morse-Koszul complex has nice topological content. As we shall see, the dimension of the homology of degree k of the Morse-Koszul complex is equal to the number of critical points of index k for every $0 \le k \le n$. On the other hand, we will verify that the dimension of the homology of degree k of the classical Koszul complex $(\Omega^*(M), i_{V_f})$ is equal to 0 for every $1 \le k \le n$ while the 0-th homology has dimension $|\operatorname{Crit}(f)|$. This may sound a little bit surprising as it shows that the Morse-Koszul complex $(C^*(f), i_{V_f})$ is not quasi-isomorphic to the classical

Koszul complex $(\Omega^*(M), i_{V_f})$ via the collection of maps $\mathbb{P}^k, 0 \leq k \leq n$. This is due to the fact that, contrary to the coboundary operator d, i_{V_f} is sensitive to the regularity of the elements of the complex.

2.4. About the proof: spectral analysis of \mathcal{L}_{V_f} . As was already alluded, understanding the asymptotic properties of φ_f^{-t*} is related to the spectral analysis of its generator which is nothing else but the Lie derivative \mathcal{L}_{V_f} along the vector field V_f . One of the main difficulty one encounters when trying to describe this spectrum is to find good Banach spaces containing $\Omega^*(M)$ and where \mathcal{L}_{V_f} has nice spectral properties such as discrete spectrum. We already explained that important progresses were made recently concerning this problem especially in the case of Anosov vector fields. Among the type of spaces that can be considered, we can very roughly distinguish two types: (1) anisotropic Hölder spaces like in [55], (2) anisotropic Sobolev spaces obtained via microlocal techniques like in [32]. Here, we will in fact closely follow the construction of Faure-Sjöstrand in [32] (see also [28, 33]) in the case of currents) and explain how to adapt it to our dynamical framework. One of the main issue we have to deal with is the asymptotic behaviour of the Hamiltonian lift of φ_f^t . In particular, we have to verify that the attractor and the repeller of the normalized Hamiltonian flow are compact subsets – see Lemmas 3.6 and 3.8. This is one of the first place where we will strongly use our extra assumptions on the flow, namely the Smale transversality and the (smooth) linearizing property near every critical point.

After setting properly this dynamical framework and its asymptotic properties, we can closely follow the construction from [32] which requires minor (but necessary) modifications and which will be described in section 4 – see also appendices A and B. Given any $\Lambda > 0$ and any $0 \le k \le n$, this procedure allows us to construct an anisotropic Sobolev space $\mathcal{H}_k^{m_{\Lambda}}(M) \subset \mathcal{D}'^{,k}(M)$ such that

$$-\mathcal{L}_{V_f}^{(k)}:\mathcal{H}_k^{m_{\Lambda}}(M)\to\mathcal{H}_k^{m_{\Lambda}}(M),$$

and such that the operator has only discrete spectrum with finite multiplicity in the half plane $\{\text{Re}(z) > -\Lambda\}$. According to [32, Th. 1.5], these values are independent of the choice of our anisotropic space. These complex numbers are called the Pollicott-Ruelle resonances of $-\mathcal{L}_{V_f}^{(k)}$ [62, 64], and they correspond to the poles of the meromorphic extension of $(-\mathcal{L}_{V_f}^{(k)} - z)^{-1}: \Omega^k(M) \to \mathcal{D}'^{,k}(M)$ to the complex plane. We denote these poles by $\mathcal{R}_k(f,g)$.

In the case where (f, g) is a smooth Morse pair inducing a Morse-Smale gradient flow, we will obtain several results on their structure that we will now describe:

- (1) Any element in $\mathcal{R}_k(f,g)$ is contained in $(-\infty,0]$ and is a linear combination with integer coefficients of the Lyapunov exponents at a fixed critical point a (Proposition 5.1).
- (2) Up to its multiplicity, we can determine exactly which linear combination appears in the spectrum (Proposition 5.1).

- (3) The dimension of the kernel is equal to the number of critical points of index k, and we have an expression for any element of the basis in a neighborhood of any critical point of index a (Propositions 6.7 and 6.8).
- (4) If all the Lyapunov exponents are rationally independent, we can determine the multiplicity of every element in $\mathcal{R}_k(f,g)$ and the local expression of the eigenmodes near the associated critical point (Propositions 7.5 and 7.7).
- (5) The algebraic multiplicity of an eigenvalue is always equal to its geometric multiplicity (Propositions 6.1 and 7.2).
- (6) In particular, we can determine Weyl asymptotics in terms of the Lyapunov exponents (Proposition 7.9) and we recover the classical Lefschetz trace formula (Proposition 7.10).

The combination of all these results allows to prove Theorem 2.2 and Theorem 2.3– see Propositions 7.8 and 6.9. The proofs of these different spectral results will heavily rely on the construction of the spaces $\mathcal{H}_k^{m_{\Lambda}}(M)$. In particular, we will deduce from that construction that we have a very strong information on the wave front set of our eigenmodes [50], namely it is contained in the unstable direction of the gradient flow. Combining this a priori information to the dynamical structure of our flow and to results from distribution theory and microlocal analysis, we will be able to reconstruct in some sense a lot of informations on the Pollicott-Ruelle resonances and their corresponding eigenmodes. As a by-product of our spectral analysis (see paragraph 6.4.5), we observe that our description of the spectrum of $\mathcal{L}_{V_f}^{(k)}$ implicitly shows the following result:

Corollary 2.5. Let (f,g) be a smooth Morse pair generating a Morse-Smale gradient flow φ_f^t and let $0 \le k \le n$. Then, for every ψ in $\Omega^k(M)$, the cohomological equation

$$\mathcal{L}_{V_f}^{(k)}u = \psi$$

admits a solution in $\mathcal{H}_k^{m_{\Lambda}}(M)$ if and only if, for every critical point a of index k, one has

$$\langle S_a, \psi \rangle = 0.$$

From the construction, one knows that the anisotropic Sobolev space $\mathcal{H}_k^{m_\Lambda}(M)$ is contained in the standard Sobolev space $\mathcal{H}_k^{-N}(M)$ for some large positive constant N. In the case of a strongly non-resonant Morse-Smale gradient flow, the argument of paragraph 6.4.5 could be adapted to find criteria under which the cohomological equation $(\mathcal{L}_{V_f}^{(k)} + z)u = \psi$ admits solution in $\mathcal{H}_k^{-N}(M)$ for N > 0 large enough depending on the real part of z.

2.5. Witten's approach to Morse theory and supersymmetric quantum mechanics. In [76], Witten recovered the classical Morse inequalities by studying the operator $d_{\hbar} := \hbar e^{-\frac{f}{\hbar}} de^{\frac{f}{\hbar}}$ and its adjoint d_{\hbar}^* in the semiclassical limit $\hbar \to 0$. More precisely, he obtained this result by considering the bottom of the spectrum of the twisted Laplacian $d_{\hbar}^* d_{\hbar} + d_{\hbar} d_{\hbar}^*$. Even if the operator \mathcal{L}_{V_f} is of different nature, we would like to emphasize the following common point with Witten's approach. Like in [32, 28], the strategy is to consider the action of $\mathcal{L}_{V_f}^{(k)}$ on certain anisotropic Sobolev spaces $\mathcal{H}_k^{m_{\Lambda}}(M)$. As we shall see in

Appendix B, this is equivalent to conjugating the operator by a certain elliptic pseudifferential operator $\operatorname{Op}(\mathbf{A}_{m_{\Lambda}}^{(k)})$ with variable order and to consider the action of the conjugated operator on the Hilbert space $L^2(M, \Lambda^k(T^*M))$. Compared with Witten, we deform the operator \mathcal{L}_{V_f} (instead of d and d^*) by conjugating it by $\operatorname{Op}(\mathbf{A}_{m_{\Lambda}}^{(k)})$ (instead of $e^{\frac{f}{h}}$). Following the works of Faure-Sjöstrand, the effect of this conjugation is to unveil some dynamical resonances of the operator $\mathcal{L}_{V_f}^{(k)}$, and, like in Witten's approach, this dynamical spectrum carries some topological informations related to the Morse complex – see section 8 for more details.

Recall that Witten's theory was inspired by supersymmetric (SUSY) quantum mechanics. In some sense, this framework allows to compare our approach with Hodge theory (and also with Witten's work). This unifying point of view is discussed from the physical perspective in inspiring works of Losev et al. [36, 37, 52, 56, 57]. In order to make this parallel between Hodge theory and the present work, we observe that the (N = 1) supersymmetric quantum mechanical system associated with our problem is given by:

- a \mathbb{Z}_2 -graded state space $\mathcal{H}^{m_{\Lambda}}(M) = \mathcal{H}^{m_{\Lambda}}_{+}(M) \oplus \mathcal{H}^{m_{\Lambda}}_{-}(M)$ of anisotropic Sobolev currents with bosonic (resp fermionic) states $\mathcal{H}^{m_{\Lambda}}_{+}(M)$ (resp $\mathcal{H}^{m_{\Lambda}}_{-}(M)$) of even (resp odd) degree.
- a fermion operator $(-1)^F$ whose eigenspaces are $(\mathcal{H}^{m_{\Lambda}}_{+}(M), \mathcal{H}^{m_{\Lambda}}_{-}(M))$ with eigenvalue ± 1 depending on the parity of the state,
- two operators of symmetry, i.e. $Q_1 = d + i_{V_f}$ and $Q_2 = i(d i_{V_f})$ which satisfies $Q_1Q_2 + Q_2Q_1 = 0$ (i_{V_f} plays the role of d^* in Hodge theory),
- a Hamiltonian which is given by the Lie derivative \mathcal{L}_{V_f} and which verifies $\mathcal{L}_{V_f} = Q_1^2 = Q_2^2 = di_{V_f} + i_{V_f}d$ from Cartan's formula.

As we shall see, the currents U_a appearing in Corollary 2.3 can be interpreted as Lagrangian states whose wave front set lies in conormals of unstable manifolds. From the point of view of supersymmetry, they span the vacuum $C^*(f)$ (generated by eigenstates of minimal energy) of our quantum theory.

2.6. Organization of the article. In section 3, we gather some crucial dynamical preliminaries and introduce some notations that will be used all along the article. In section 4, we make use of our dynamical assumptions in order to construct anisotropic spaces of currents which are adapted to our problem. As our construction is very close to the one in [32], we mostly focus on the differences, namely the construction of the escape function whose detailed proof is postponed to appendix A. In section 5, we make use of the regularity properties of the eigenmodes to determine exactly the Pollicott-Ruelle spectrum (up to its multiplicities). In section 6, we study the case of the eigenvalue 0. Section 7 is devoted to describe the local structure of the eigenmodes under some nonresonance assumptions. From that, we deduce some results on the multiplicity of the eigenvalues. We explain in section 8 how to deduce some classical results of differential topology from the results obtained in the previous sections. In appendix A, we give the complete proof of the construction of the escape function. Appendix B is devoted to a brief reminder of [32] concerning the

proof of Proposition 4.2. Finally, appendix C collects some facts on asymptotic expansions that we use at several stages of our work.

Acknowledgements. We would like to thank Frédéric Faure for many explanations about his recent works on transfer operators and for generously sharing his knowledge on these questions with us. We warmly thank Livio Flaminio, Damien Gayet, Colin Guillarmou, Patrick Popescu Pampu, Nicolas Vichery and Tobias Weich for useful discussions related to this work. Finally, we are grateful to Serguei Barannikov, Lie Fu, Camille Laurent-Gengoux and Joan Millès for explanations and comments related to the content of section 8 and more specifically to Koszul complexes. The second author is partially supported by the Agence Nationale de la Recherche through the Labex CEMPI (ANR-11-LABX-0007-01) and the ANR project GERASIC (ANR-13-BS01-0007-01).

3. Morse-Smale gradient flows

In this section, we briefly collect some facts on the dynamical properties of Morse-Smale gradient flows. The main new results of this section are Lemmas 3.6 and 3.8 which are related to earlier works of Weber [75] on the space of connecting orbits for gradient flows. These two lemmas are the crucial ingredients to develop the machinery of anisotropic Sobolev spaces of Faure and Sjöstrand [32]. We also fix some conventions that we will use all along the article. For the well-known results, we follow the lines of [54, 75] and we refer to these references for a more detailed exposition. Recall that, in all the article, M denotes a smooth (\mathcal{C}^{∞}) , oriented, compact manifold without boundary.

3.1. **Gradient flows.** Let $f: M \to \mathbb{R}$ be a smooth function on M. If we fix a Riemannian metric g on M (compatible with our orientation), then we can define the corresponding gradient vector field as follows:

$$\forall x \in M, \ \langle \operatorname{grad} f(x), . \rangle_{g(x)} := df(x).$$

In local coordinates, this can we written as

$$V_f(x) := \operatorname{grad} f(x) = \sum_{i,j=1}^n g^{ij}(x) \partial_{x_j} f \partial_{x_j},$$

where $(g^{ij}(x))_{1 \le i,j \le n}$ is the induced Riemannian metric on T_x^*M . Under our geometric assumptions (compactness of the manifold), one knows that the gradient vector field induces a complete flow that we denote by

$$\varphi_f^t:M\to M.$$

If it does not create any particular confusion, we will sometimes use the convention $x(t) = \varphi_f^t(x_0)$ for a fixed x_0 in M. Note that, for any integral curve $t \mapsto x(t)$ of the gradient vector field, one has

(4)
$$\forall t_1, t_2 \in \mathbb{R}, \ f(x(t_2)) - f(x(t_1)) = \int_{t_1}^{t_2} \|\operatorname{grad} f(x(t))\|_{g(x(t))}^2 dt.$$

In other words, f can only increase along the flow lines. Suppose now that f is a Morse function. We denote by Critf its critical points. For the sake of simplicity, we will also assume our Morse function f to be **excellent** which means that all critical values are distinct. Recall that such functions are dense in the topological space $\mathcal{C}^{\infty}(M,\mathbb{R})$. The Morse Lemma tells us

Lemma 3.1 (Morse Lemma). Let f be a Morse function on a Riemannian manifold (M, g). Then, near any critical point a, there is a system of coordinates $(z_i)_i$ such that the point a is given by z = 0, and such that

$$f(x) = f(a) - \frac{z_1^2}{2} - \dots - \frac{z_r^2}{2} + \frac{z_{r+1}^2}{2} + \dots + \frac{z_n^2}{2},$$

for some $0 \le r \le n$. The integer r is called the **index** of the critical point a. We will either denote it by r(a) or ind(a).

An important property of the dynamical system $\varphi_f^t: M \to M$ is that, for any given x_0 in M, there exist two points x_- and x_+ in Crit f such that

$$\lim_{t \to +\infty} \varphi_f^t(x_0) = x_{\pm}.$$

3.2. **Stable and unstable manifolds.** Let a be a critical point of f. The stable (resp. unstable) manifold $W^s(a)$ (resp. $W^u(a)$) is defined as the set of points x in M satisfying $\varphi_f^t(x) \to a$ as $t \to +\infty$ (resp. $t \to -\infty$). From [75, Th. 2.7], one knows that $W^s(a)$ (resp. $W^u(a)$) is a *smooth submanifold* of dimension r(a) (resp. n-r(a)) where $0 \le r(a) \le n$ is the index of the critical point a. Note that, for more general vector fields with an hyperbolic point, the stable (resp. unstable) manifold is a priori only injectively immersed in M. The fact that we consider a gradient flow allows to show that it is also embedded [75, Th. 2.7], even if it is not a priori properly embedded.

We will say that the gradient flow φ_f^t satisfies the **Morse-Smale assumption** if for every pair of critical points (a,b), the submanifolds $W^s(a)$ and $W^u(b)$ are transversal. Note that, in the case where a=b, the intersection of the tangent spaces is in fact reduced to $\{0\}$. When $a \neq b$ and $W^s(a) \cap W^u(b)$ is non empty, then this assumption ensures that r(b) < r(a).

Let us now fix some conventions. Given any x_0 in M, there exists a unique pair of critical points (x_-, x_+) such that x_0 belongs to $W^u(x_-) \cap W^s(x_+)$. We define

$$E^{u}(x_0) := T_{x_0}W^{u}(x_{-})$$
 and $E^{s}(x_0) := T_{x_0}W^{s}(x_{+})$.

Note that, whenever x_0 is not a critical point of f, the intersection of these two subspaces is not reduced to 0 as they both contains the direction of the flow. From our transversality assumption, one has $T_{x_0}M = E^u(x_0) + E^s(x_0)$. We refer to paragraph 3.3 for a more detailed description of these subspaces at the critical points of f. We can also introduce the dual spaces $E_u^*(x_0)$ and $E_s^*(x_0)$ which are defined as the annihilators of these unstable and stable spaces, i.e. $E_{u/s}^*(x_0)(E^{u/s}(x_0)) = 0$. From the Morse-Smale transversality assumption, one can verify that, for any x_0 in M,

(6)
$$E_u^*(x_0) \cap E_s^*(x_0) = \{0\}.$$

3.3. Lyapunov exponents. Given every point x in M, we define $L_f(x)$ as the unique matrix satisfying

$$\forall \xi, \eta \in T_x M, \ \langle L_f(x)\xi, \eta \rangle_{q(x)} = d_x^2 f(\xi, \eta).$$

Let a be an element in Crit(f). The matrix $L_f(a)$ corresponds to the linearization of V_f at the point a. It can be shown [75, Lemma 2.5] that

$$\forall t \in \mathbb{R}, \ d\varphi_f^t(a) = \exp(tL_f(a)).$$

Moreover, from the definition and from the Morse assumption, one can verify that $L_f(a)$ is an invertible matrix which is symmetric with respect to the Riemannian metric g. In particular, it is diagonalizable and we denote by $(\chi_j(a))_{j=1,\dots,n}$ its eigenvalues. These nonzero real numbers are called the **Lyapunov exponents** of the system. They depend both on f and on the metric g. We will always suppose that $\chi_i(a) < 0$ for $1 \le i \le r$ and $\chi_i(a) > 0$ for $r+1 \le i \le n$. Moreover, there exists a basis of eigenvectors which is orthonormal with respect to the metric g. According to [75, Th. 2.7], the stable (resp. unstable) space is in fact equal to the direct sum of eigenspaces corresponding to the negative (resp. positive) eigenvalues of $L_f(a)$.

3.4. Lift to the cotangent space. We will now explain how one can lift this gradient flow to the cotangent space T^*M . We associate to the vector field V_f an Hamiltonian function H_f which can be written as follows:

$$\forall (x,\xi) \in T^*M, \ H_f(x,\xi) := \xi \left(V_f(x) \right).$$

This Hamiltonian function also induces an Hamiltonian flow that we denote by $\Phi_f^t: T^*M \to T^*M$. We note that, by construction,

$$\Phi_f^t(x,\xi) := \left(\varphi_f^t(x), \left(d\varphi_f^t(x)^T\right)^{-1}\xi\right),\,$$

and that this flow induces a diffeomorphism between $T^*M - \{0\}$ and $T^*M - \{0\}$. When it does not lead to any confusion, we will also write $\Phi_f^t(x,\xi) = (x(t),\xi(t))$. We note that this flow induces a smooth flow on the unit cotangent bundle S^*M , i.e.

$$\forall t \in \mathbb{R}, \ \forall (x,\xi) \in S^*M, \ \tilde{\Phi}_f^t(x,\xi) = \left(\varphi_f^t(x), \frac{\left(d\varphi_f^t(x)^T\right)^{-1}\xi}{\left\|\left(d\varphi_f^t(x)^T\right)^{-1}\xi\right\|}\right).$$

We denote by \tilde{X}_{H_f} the induced smooth vector field on S^*M .

3.5. Adapted coordinates. We will say in the following that (f, g) is a smooth Morse pair if, given any critical point a of f, one can find an open neighborhood V_a of a and a system of smooth (meaning C^{∞}) local coordinate charts $(z_j)_{j=1,\dots,n} = (x,y)$ such that, in this coordinate chart, the vector field V_f reads

(7)
$$V_f := \sum_{j=1}^n \chi_j(a) z_j \partial_{z_j} = -\sum_{j=1}^r |\chi_j(a)| x_j \partial_{x_j} + \sum_{j=r+1}^n |\chi_j(a)| y_j \partial_{y_j}.$$

The key point for us is that this change of coordinates is smooth which will allow us to take as many derivatives as we want in the following sections where we aim at using microlocal techniques. Requiring that there exists a smooth change of coordinates for which the gradient vector field can be linearized may a priori look as a strong assumption. We will briefly discuss below two situations where this assumption is satisfied, the second one being in some sense rather general.

When this assumption is satisfied, we shall say that we have an **adapted system** of coordinates. We note that the function f may not have a nice expression in these coordinates, meaning that f may a priori not have a Morse-type expression. In such a coordinate chart, the gradient flow reads

$$\varphi_f^t(z) = (e^{t\chi_1(a)}z_1, \dots, e^{t\chi_n(a)}z_n)
= (e^{-|\chi_1(a)|t}x_1, \dots, e^{-|\chi_r(a)|t}x_r, e^{|\chi_{r+1}(a)|t}y_{r+1}, \dots, e^{|\chi_n(a)|t}y_n).$$

Let us fix some conventions that we will use in the following. For every critical point a, we denote the change of coordinates by

$$\kappa_a: w \in V_a \subset M \to (x,y) \in W_a = (-\delta_a, \delta_a)^n \subset \mathbb{R}^n,$$

where $\delta_a > 0$ is some small enough parameter.

Remark 3.2. Let $(U, \kappa) = (u^i)$ be local coordinates on M. Whenever the chart is of class \mathcal{C}^1 , one can lift in a canonical way these coordinates into coordinates (u^i, v_j) on the cotangent space T^*M by using (\mathcal{K}, T^*U) , where $\mathcal{K}(x, \xi) = (\kappa(x), (d\kappa(x)^T)^{-1}\xi)$. When we make a change of coordinates $(\tilde{u}^i, \tilde{v}_j)$, one can verify that \tilde{v} is the image of v under a linear transformation (depending only on the coordinate charts (u^i) and (\tilde{u}^i)).

We can also write the expression of the Hamiltonian flow in the corresponding adapted coordinate chart near a critical point a. In such a chart, one can write

$$H_f(z,\zeta) = \sum_{j=1}^n \chi_j(a) z_j \zeta_j = -\sum_{j=1}^r |\chi_j(a)| x_j \xi_j + \sum_{j=r+1}^n |\chi_j(a)| y_j \eta_j.$$

In particular, the map Φ_f^t can be written in this local coordinate chart as

$$\Phi_f^t(z,\zeta) = (e^{\chi_1(a)t}z_1, \dots e^{\chi_n(a)t}z_n; e^{-\chi_1(a)t}\zeta_1, \dots e^{-\chi_n(a)t}\zeta_n).$$

3.5.1. Locally flat metrics. The vector field V_f is a priori not linear in the chart of the Morse Lemma. In fact, there might be no Morse chart in which $g_{ij}(0)$ is diagonal and the vector field linear. We will call the metric g locally flat with respect to f if, for any critical point of f, there exists a smooth Morse chart (z_i) in which the vector field V_f has the linear form

$$V_f(z) = \sum_{j=1}^n \chi_j(a) z_j \partial_{z_j}.$$

Such flows are also sometimes referred as tame flows. This type of locally flat metrics appears for instance in [45, 54]. It is shown in [45] that, given any Morse function, one can

find an adapted metric g such that the Morse-Smale transversality assumption is satisfied. Moreover, they proved that this property is more or less generic among such metrics.

3.5.2. Sternberg-Chen Theorem. We would like to justify that asking for a smooth change of coordinates which linarizes the gradient flow is in some sense generic. For that purpose, we just recall Sternberg-Chen's Theorem on the linearization of vector fields near hyperbolic critical points [20] (see also [59, Th. 9, p.50]):

Theorem 3.3 (Sternberg-Chen). Let $X(x) = \sum_j a_j(x) \partial_{x_j}$ be a smooth vector field defined in a neighborhood of 0 in \mathbb{R}^n . Suppose that X(x) = 0. Denote by (χ_j) the eigenvalues of $L := (\partial_{x_k} a_j(0))_{k,j}$. Suppose that the eigenvalues satisfy the **non resonant assumption**,

$$\forall k_1, \dots, k_n \in \mathbb{Z} \text{ s.t } k_1, \dots k_n \ge 2, \ \forall \ 1 \le i \le n, \ \chi_i \ne \sum_{j=1}^n k_j \chi_j.$$

Then, there exists a smooth diffeomorphism h which is defined in a neighborhood of 0 such that

$$X \circ h(x) = dh \circ (Lx.\partial_x).$$

The classical Grobman-Hartman Theorem [44] ensures the existence of a conjugating homeomorphism. The crucial point of the Sternberg-Chen Theorem is that the conjugating map is smooth provided some non resonance assumption is made. Applying this Theorem locally near the critical points of f allows to show the existence of a smooth and adapted system of coordinates. Note that this non-resonant assumption on the eigenvalues is for instance satisfied if, for every a in Crit(f), the Lyapunov exponents $(\chi_j(a))_{j=1,\dots,n}$ are rationally independent. In section 7, we will in fact make the assumption that all the Lyapunov exponents are rationally independent.

3.6. Attractor and repeller of the Hamiltonian flow. We now introduce the following subsets of T^*M :

$$\Gamma_{+} = \bigcup_{x \in M} E_{s}^{*}(x), \ \Gamma_{-} = \bigcup_{x \in M} E_{u}^{*}(x), \text{ and } \Gamma = M \times \{0\}.$$

We have then

Lemma 3.4. Suppose that (f,g) is a smooth Morse pair which generates a Morse-Smale gradient flow φ_f^t . One has, for every (x,ξ) in T^*M with $\xi \neq 0$,

$$(x,\xi) \in \Gamma_{\pm} \implies \lim_{t \to +\infty} ||\xi(t)||_{x(t)} = 0,$$

and

$$(x,\xi) \notin \Gamma_{\pm} \implies \lim_{t \to \pm \infty} \|\xi(t)\|_{x(t)} = +\infty,$$

where
$$(x(t), \xi(t)) = \Phi_f^t(x, \xi)$$
.

This Lemma tells us that the trapped set of the Hamiltonian flow is reduced to the zero section of T^*M . The proof of this Lemma will crucially use the fact that we made the Morse-Smale assumption and that we have a smooth (at least C^1) change of coordinates which linearizes the vector field.

Proof. We only consider the case where $t \to +\infty$ (the other case can be obtained by replacing f by -f). Let (x,ξ) be an element in T^*M with $\xi \neq 0$. There exists a critical point x_+ of f such that $\lim_{t\to +\infty} \varphi_f^t(x) = x_+$. In particular, for t>0 large enough, $\varphi_f^t(x)$ belongs to the adapted chart around x_+ which was defined in paragraph 3.5. Up to a translation of time, one can write that, in this system of adapted coordinates and for every $t \geq 0$

$$\Phi_f^t(x,\xi) = (e^{-|\chi_1|t}x_1, \dots, e^{-|\chi_r|t}x_r, 0, \dots, 0, e^{|\chi_1|t}\xi_1, \dots, e^{|\chi_r|t}\xi_r, e^{-|\chi_{r+1}|t}\eta_{r+1}, \dots, e^{-|\chi_n|t}\eta_n).$$

Hence, as all the norms can be made uniformly equivalent to the Euclidean norm in a small neighborhood of x_+ , one can find two positive constants $0 < C_1 < C_2$ such that

$$C_1\left(\sum_{j=1}^r e^{2|\chi_j|t}\xi_j^2 + \sum_{j=r+1}^n e^{-2|\chi_j|t}\eta_j^2\right) \le \|\xi(t)\|_{x(t)}^2 \le C_2\left(\sum_{j=1}^r e^{2|\chi_j|t}\xi_j^2 + \sum_{j=r+1}^n e^{-2|\chi_j|t}\eta_j^2\right).$$

The fact that (x, ξ) belongs to Γ_+ is exactly equivalent to the fact that $\xi_1 = \ldots = \xi_r = 0$ from which one can easily conclude the expected property.

Introduce now the two following disjoint subsets of S^*M :

$$\Sigma_u := S^*M \cap \Gamma_+$$
, and $\Sigma_s := S^*M \cap \Gamma_-$.

Then, one has:

Lemma 3.5. Suppose that (f,g) is a smooth Morse pair which generates a Morse-Smale gradient flow φ_f^t . One has

(8)
$$\forall (x,\xi) \in S^*M - \Sigma_s, \lim_{t \to -\infty} d_{S^*M} \left(\tilde{\Phi}_f^t(x,\xi), \Sigma_u \right) = 0,$$

and

(9)
$$\forall (x,\xi) \in S^*M - \Sigma_u, \lim_{t \to +\infty} d_{S^*M} \left(\tilde{\Phi}_f^t(x,\xi), \Sigma_s \right) = 0.$$

This lemma tells us that Σ_u and Σ_s are in a certain weak sense repeller and attractor of the flow $\tilde{\Phi}_f^t$. A stronger version of this fact will be given in Lemma 3.8.

Proof. We proceed as in the proof of Lemma 3.4, and we just treat the case where $t \to +\infty$. Let (x,ξ) be an element in $S^*M - \Sigma_u$. In other words, (x,ξ) does not belong Γ_+ . Using the notations of the proof of Lemma 3.4, it means that

$$\Phi_f^t(x,\xi) = (e^{-|\chi_1|t}x_1, \dots, e^{-|\chi_r|t}x_r, 0, \dots, 0, e^{|\chi_1|t}\xi_1, \dots, e^{|\chi_r|t}\xi_r, e^{-|\chi_{r+1}|t}\eta_{r+1}, \dots, e^{-|\chi_n|t}\eta_n),$$

with $\xi_j \neq 0$ for some $1 \leq j \leq r$. By letting $t \to +\infty$, we find that that any accumulation point of $\tilde{\Phi}_f^t(x,\xi)$ is of the form $(0,\ldots,0,\tilde{\xi}_1,\ldots,\tilde{\xi}_r,0,\ldots,0)$. Equivalently, every accumulation point belongs to Σ_s .

3.7. Compactness. In order to make the machinery of anisotropic Sobolev space work, it will first be important for us that Σ_u and Σ_s are compact subsets of S^*M . This assumption is verified for gradient flow satisfying the Morse-Smale assumption:

Lemma 3.6. Suppose that (f,g) is a smooth Morse pair which induces a gradient flow with the Morse-Smale property. Then, the subsets Σ_u and Σ_s are compact in S^*M .

This property combined with Lemma 3.8 will be crucial in our construction of anisotropic Sobolev spaces. In particular, they are necessary to prove Lemma 2.1 from [32] which is at the heart of this construction. We note that the proof of Lemma 3.6 requires both the Morse-Smale assumption and the existence of a (at least C^1) linearizing chart for the flow.

Remark 3.7. Even if this Lemma sounds natural, the proof is a little bit subtle and it is related to the so-called Whitney regularity condition [60, Ch. 7] – see also the appendix of [54] for related results in the case of locally flat metrics. Here, we are aiming at weaker results than in these references and we shall give a proof of our Lemma which is based on purely "dynamical arguments". Our argument is in fact very close to the proof of the compactness of the space of connecting orbits of Weber in [75, Th. 3.8] – see also [68] for earlier related results of Smale. In this reference, it was proved that the space of connecting orbits between two critical points a and b is "compact up to broken orbits". It means that, for a fixed sequence $(x_p)_{p\geq 1}$ in $W^u(a)\cap W^s(b)$, there exists (up to extraction) a sequence of critical points $a=b_0,b_1,\ldots,b_l=b$ and a finite sequence of points u_m in $W^u(b_m)\cap W^s(b_{m+1})$ such that

$$\forall \epsilon_1 > 0, \exists p_0, \forall p \ge p_0, \ d\left(\mathcal{O}(x_p), \bigcup_{0 \le m \le l-1} \mathcal{O}(u_m)\right) < \epsilon_1,$$

where $\mathcal{O}(x)$ denotes the orbit of x under the flow φ_f^t . The key "dynamical argument" in the proof of Weber was to use the Grobman-Hartman linearization Theorem around the critical points of f. Here, we want to prove something slightly stronger in the sense that we will have to keep track of the behaviour of the cotangent vectors in the phase space S^*M and not only of the points in the position space M. For that purpose, we will crucially makes use of the fact that we have a smooth (at least \mathcal{C}^1) chart where the vector field can be linearized. In some sense, our compactness statement on S^*M requires the Sternberg-Chen's Theorem while Weber's compactness statement on M only required the Grobman-Hartman's Theorem.

Proof. We only treat the case of Σ_s as the case of Σ_u can be obtained by replacing f by -f. In order to prove compactness, we will just prove that Σ_s is closed (as S^*M is compact). Consider (z_m, ζ_m) a converging subsequence in Σ_s . We denote by (z_∞, ζ_∞) its accumulation point which belongs to S^*M . We will proceed by contradiction and suppose that (z_∞, ζ_∞) does not belong to Σ_s . Without loss of generality, we can suppose that, for every $m \geq 1$, z_m belongs to the unstable manifold $W^u(a)$ of some critical point a. Observe also that z_∞ belongs to $W^u(b_1)$ for some b_1 in $\operatorname{Crit}(f)$. If $a = b_1$, we can apply the backward flow Φ^{-T} for T large enough to the sequence $(z_m; \zeta_m)_m$ and its limit (z_∞, ζ_∞) so that we may assume that the sequence and its limit are in the linearized chart near a and go directly to the final step of our proof.

Suppose now that $a \neq b_1$. In that case, we will show that there exists a converging subsequence $(z_m^{(1)}, \zeta_m^{(1)})$ in $\Sigma_s \cap W^u(a)$ such that the accumulation point $(z_\infty^{(1)}, \zeta_\infty^{(1)})$ belongs to $W^s(b_1) - \{b_1\}$ but not to Σ_s . For that purpose, we note that $z_\infty \in W^u(b_1) \Longrightarrow \lim_{T \to +\infty} \varphi^{-T}(z_\infty) = b_1$. Hence by continuity of $\varphi_f^{-T}: M \mapsto M$, applying the flow $\tilde{\Phi}_f^{-T}$ for some large T > 0 and up to another extraction, we can suppose that, for every $m \geq 1$, z_m belongs to the neighborhood of b_1 with adapted coordinates as defined in paragraph 3.5. In other words, we can write z_m as a point $(x_{m,1}, \ldots, x_{m,r}, y_{m,r+1}, \ldots, y_{m,n})$ such that,

$$\sum_{j=1}^{r} |x_{m,j}| + \sum_{j=r+1}^{n} |y_{m,j}| \le \frac{\delta_{b_1}}{2},$$

where $\delta_{b_1} > 0$ is the size of the chart of adapted coordinates defined in paragraph 3.5. As $a \neq b_1$, we also know that, for every $m \geq 1$, there exists $1 \leq j \leq r$ such that $x_{m,j} \neq 0$. Otherwise the sequence x_m would belong to $W^u(b_1)$ which is given in local coordinates near b_1 by the equations $\{x_j = 0, 1 \leq j \leq r\}$. In particular, for every $m \geq 1$, there exists some $T_m > 0$ such that

$$z_m^{(1)} = \varphi_f^{-T_m}(z_m) = (e^{T_m} x_{m,1}, \dots, e^{T_m} x_{m,r}, e^{-T_m} y_{m,r+1}, \dots, e^{T_m} y_{m,n}) = (x_{m,1}^{(1)}, \dots, x_{m,r}^{(1)}, y_{m,r+1}^{(1)}, \dots, y_{m,n}^{(1)})$$
 verifies

$$\exists 1 \le j \le r \text{ s.t. } |x_{m,j}^{(1)}| \ge \frac{\delta_{b_1}}{2}.$$

One can verify that any accumulation point $z_{\infty}^{(1)}$ belongs to $W^s(b_1) - \{b_1\}$. Up to this point, we did not use our assumption on ζ_{∞} and we did not track what happens for the covector ζ_m under application of the flow $\tilde{\Phi}_f^{-T_m}$. From our assumption that $(z_{\infty}, \zeta_{\infty})$ does not belong to Σ_s and from property (8), we find that $d\left(\tilde{\Phi}_f^{-T}(z_{\infty}, \zeta_{\infty}), \Sigma_u\right) \to 0$ when $T \to +\infty$. Fix now $p \geq 1$. Since Σ_u is locally given by $\{\xi_1 = \dots = \xi_r = 0\}$, we observe that if we pick T > 0 larger in the first step (in a way that depends on $p \geq 1$) and up to another extraction, we suppose that one can find $m(p) \geq 1$ such that $\zeta_{m(p)}$ belongs to a small conic neighborhood of size 1/p of $\{\xi_1 = \dots = \xi_r = 0\}$. If the conic neighborhood is chosen small enough (i.e. if $p \geq 1$ is large enough), then it is stable under the action of $\tilde{\Phi}_f^{-T_m}$. Hence, if we set $(z_m^{(1)}, \zeta_m^{(1)}) = \tilde{\Phi}_f^{-T_m}(z_m, \zeta_m)$, then, for every p large enough, $\zeta_{m(p)}^{(1)}$ still belongs to a conic neighborhood of the stable direction of size 1/p. Note that, as in the proof of Lemma 3.4, this property follows from the expression we have for the Hamiltonian flow Φ_f^t in the adapted system of coordinates.

Up to another extraction, we can thus suppose that the limit $(z_{\infty}^{(1)}, \zeta_{\infty}^{(1)})$ satisfies

$$z_{\infty}^{(1)} \in W^s(b_1) - \{b_1\}, \text{ and } \zeta_{\infty}^{(1)} \in E_s^*(z_{\infty}^{(1)}),$$

which means in fact that $(z_{\infty}^{(1)}, \zeta_{\infty}^{(1)}) \in \Sigma_u$. In particular, as $E_u^* \cap E_s^* = \{0\}$ from the Morse-Smale assumption, we can deduce that $(z_{\infty}^{(1)}, \zeta_{\infty}^{(1)})$ does not belong to Σ_s which implies the expected property.

Now, there exists another critical point b_2 such that $z_{\infty}^{(1)}$ belongs to $W^u(b_2)$. If $b_2 \neq a$, we can apply the same procedure one more time. As there are only finitely many critical points,

this iteration should terminate with a convergent sequence $(z_m^{(l)}, \zeta_m^{(l)})_m$ in $\Sigma_s \cap S_{W^u(a)}^*M$ such that its limit $(z_\infty^{(l)}, \zeta_\infty^{(l)}) \in S_{W^u(a)}^*M$ does not belong to Σ_s by construction. We can iterate the flow $\tilde{\Phi}_f^t$ in backward times once again. Thus, we can suppose that $(z_\infty^{(l)}, \zeta_\infty^{(l)})$ belongs to the adapted chart around a. As $W^u(a)$ is a smoothly embedded manifold around the point a, we can finally get the contradiction as any sequence in Σ_s belonging to this small neighborhood should then converge to a point in Σ_s .

3.8. Invariant neighborhoods. We conclude this dynamical section with the following Lemma which states that Σ_u and Σ_s are repeller and attractor in a slightly stronger sense than in Lemma 3.5.

Lemma 3.8. Let (f,g) be a smooth Morse pair that induces a Morse-Smale gradient flow. Let $\epsilon > 0$. Then, there exists an open neighborhood V^s of Σ_s which is of size $\leq \epsilon$ and such that

$$\forall t \ge 0, \ \tilde{\Phi}_f^t(V^s) \subset V^s.$$

The same holds for Σ_u in backward times.

One more time, the proof makes use of the existence of a (at least C^1) linearizing chart for the flow. We also make use of the fact that the critical values of f are distinct.

Proof. Again, we only treat the case of Σ_s (the case of Σ_u can be obtained by replacing f by -f). Recall that we assumed our Morse function f to be excellent, meaning that all its critical values are distinct. Thus, we can define the following **total order** relation between critical points. We say that a < b if f(a) < f(b). This relation allows to order the critical points as $a_1 < a_2 < \cdots < a_K$.

The proof of this lemma requires one more time a delicate analysis of the flow. We construct the neighborhood in a progressive manner. First, we build a small neighborhood of the projection of Σ_s on M which is equal to $\bigcup_{1 \leq j \leq K: r(a_j) \neq 0} W^u(a_j)$. Then, we adjust the construction to be able to lift this open set into a small open neighborhood of Σ_s inside S^*M . In order to construct the neighborhood in M, we fix, for every $1 \leq j \leq K$ for which a_j has positive Morse index (i.e. $r(a_j) > 0$), the following open neighborhood of a_j inside M:

$$R(a_j, \epsilon_j, \epsilon_j') := \{(x, y) : \forall j, |x_j| < \epsilon_j', |y_j| < \epsilon_j\},$$

where $\epsilon_j, \epsilon_j' > 0$ are small enough parameters to ensure that this defines an ϵ neighborhood of a_j . We will now adjust the values of these parameters to construct a neighborhood of the projection of Σ_s which is invariant by φ_f^t for every $t \geq 0$. For that purpose, we proceed by induction starting from the largest values of f.

First, we observe that every point whose trajectory enters $R(a_j, \epsilon_j, \epsilon'_j)$ will either stay in this open set for every $t \geq 0$, or escape this open set (in a maybe arbitrarly large time) by crossing the following subset of M:

$$F(a_j, \epsilon_j, \epsilon_j') := \{(x, y) : \forall j, |x_j| < \epsilon_j', \exists j, |y_j| = \epsilon_j\}$$

which is one of the face of the boundary of $R(a_j, \epsilon_j, \epsilon'_j)$. We will inductively construct from the maximum of f a system of open neighborhoods $R(a_j, \epsilon_j, \epsilon'_j)_j, r(a_j) > 0$ such that for every face $F(a_j, \epsilon_j, \epsilon'_j)$ of $R(a_j, \epsilon_j, \epsilon'_j)$, there exists a finite time $T_j > 0$, such that for every $x \in F(a_j, \epsilon_j, \epsilon'_j)$, the trajectory $t \mapsto \varphi_f^t(x)$ meets $\bigcup_{j < i} R(a_i, \epsilon_i, \epsilon'_i)$ for some $t \in (0, T_j)$. For j = K, one can verify that the neighborhood is invariant by the flow in positive time provided that we pick $\epsilon_K = \epsilon'_K > 0$ small enough to ensure that we are in the neighborhood of adapted coordinates defined in paragraph 3.5. Suppose now that we have fixed the values of ϵ_i and ϵ'_i for every i > j with $r(a_i) \neq 0$ and that $r(a_j) \neq 0$. We will explain how to fix the value of ϵ_j and ϵ'_j . We claim that the forward trajectory of every point inside $F(a_j, \epsilon_j, \epsilon'_j)$ will reach

$$\bigcup_{i>j:r(a_i)\neq 0} R(a_i,\epsilon_i,\epsilon_i')$$

in a finite time $0 < t < T_j$ where T_j depends only on ϵ_i, ϵ_i' with i > j satisfying $r(a_i) \neq 0$ and on ϵ_j . In particular, this time can be made uniform in ϵ_j' . Assume by contradiction that, for every m > 0 and for every T > 0, there exists $x_{m,T}$ in $F(a_j, \epsilon_j, 1/m)$ such that the orbit $t \in [0, T] \mapsto \varphi_f^t(x_{m,T})$ does not meet the subset $\bigcup_{i>j:r(a_i)\neq 0} R(a_i, \epsilon_i, \epsilon_i')$. We fix T > 0, and, by compactness, one can extract a subsequence such that $x_{m,T} \to x_{\infty,T}$ as $m \to +\infty$ where $x_{\infty,T}$ belongs to $W^u(a_j)$ is at distance $> \mathcal{O}(\epsilon_j)$ of a_j . We now extract another subsequence (as $T \to +\infty$) and we obtain a point $x_\infty \neq a_j$ in $W^u(a_j)$ that would not reach $\bigcup_{i>j:r(a_i)\neq 0} R(a_i,\epsilon_i,\epsilon_i')$ in finite time. This contradicts the fact that $\lim_{t\to +\infty} \varphi_f^t(x_\infty)$ is equal to a_i for some i > j satisfying $r(a_i) \neq 0$.

Recall now that the distance between two trajectories can grow at most exponentially under the flow [78, Lemma 11.11]. Hence, if we choose $\epsilon'_j > 0$ small enough, we can ensure that, the forward trajectory of every point inside $F(a_j, \epsilon_j, \epsilon'_j)$ will remain ϵ close to $W^u(a_j)$ up the finite time $t \leq T_j$ where it will enter one of the neighborhood $R(a_i, \epsilon_i, \epsilon'_i)$ with i > j and $r(a_i) > 0$. This construction defines a family of open neighborhood of the critical points a_j of index > 0 whose forward trajectory under the flow will remain within a distance $\epsilon > 0$ of $\bigcup_{1 \leq j \leq K: r(a_j) \neq 0} W^u(a_j)$ which is exactly the projection of Σ_s on M. Then, we set

$$\mathcal{N} := \bigcup_{t \ge 0} \bigcup_{1 \le j \le K: r(a_j) \ne 0} \varphi_f^t(R(a_j, \epsilon_j, \epsilon_j')).$$

By construction, this set is invariant by φ_f^t . Moreover, it defines a neighborhood of $\bigcup_{1 \leq j \leq K: r(a_i) \neq 0} W^u(a_j)$ which is of size $\leq \epsilon$.

It now remains to verify that we can lift this neighborhood into a neighborhood of size ϵ of Σ_s . For that purpose, we rely on the fact that, our smooth system of coordinate chart allows to linearize also the Hamiltonian flow Φ_f^t . Hence, we fix another positive parameter $\epsilon_j''>0$ and we consider above each neighborhood $R(a_j, \epsilon_j, \epsilon_j')$ an open neighborhood $\tilde{R}(a_j, \epsilon_j, \epsilon_j', \epsilon_j'')$ in S^*M made of unit covectors which are within a distance $<\epsilon_j''$ of $\xi_1 = \ldots = \xi_r = 0$. For every fixed choice of $\epsilon_j > 0$ and $\epsilon_j'' > 0$, we can use the compactness of Σ_s to fix $\epsilon_j' > 0$ small enough to ensure that this defines indeed a neighborhood of size $<\epsilon$ of $\Sigma_s \cap S^*R(a_j, \epsilon_j, \epsilon_j')$. Using the fact that the distance between two trajectories can grow at most exponentially

under the flow $\tilde{\Phi}_f^t$, we can argue by induction as in the case of M. More precisely, at each step of the induction, we can fix $\epsilon_j''>0$ small enough in a way that depends only on the values of ϵ_j and of $\epsilon_i^{(*)}$ with i>j and $r(a_i)>0$ and such that

$$\tilde{\mathcal{N}} := \bigcup_{t \geq 0} \bigcup_{1 \leq j \leq K: r(a_j) \neq 0} \tilde{\Phi}_f^t (\tilde{R}(a_j, \epsilon_j, \epsilon_j', \epsilon_j''))$$

defines a forward invariant open neighborhood of Σ_s of size $< \epsilon$.

4. Spectral properties of the transfer operator acting on currents

In [32], Faure and Sjöstrand constructed some anisotropic Sobolev spaces adapted to the spectral study of transfer operators for Anosov flows – see also [31] for the case of Anosov diffeomorphisms. The extension of this microlocal approach to anisotropic spaces of currents was done by Dyatlov and Zworski in [28, Sect. 3] by a slightly different approach. The definition of these spaces is based on the construction of a nice escape function adapted to the dynamics of the flow, namely a function that is strictly decreasing along the flow. This type of construction appears in fact naturally in the study of semiclassical resonances – see e.g. the appendix of [38]. In our framework, a natural candidate to pick regarding (4) is the opposite of the function f. However, near the critical points of f, the derivative along the flow vanishes, and we have to find an appropriate candidate for this part of phase space. This can be done by using the hyperbolicity of the flow at these points and mimicking the construction of [32].

This section is organized as follows. First, we state the existence of a nice escape function enjoying the dynamical features of [32, 28]. This allows us to define some Sobolev spaces of anisotropic currents following these references. Finally, we recall the spectral properties of $-\mathcal{L}_{V_f}^{(k)}$ acting on these spaces. The main difference with the above references is the construction of the escape function which requires modifications compared with the setting from [32, Lemma 2.1] where the authors made use of the Anosov property. Lemmas 3.6 and 3.8 will in fact ensure that the construction of Faure and Sjöstrand can be extended to our framework.

From this point on, we will always assume that (f, g) is a smooth Morse pair generating a Morse-Smale gradient flow.

4.1. Construction of anisotropic Sobolev spaces.

4.1.1. Escape function. The key ingredient in the construction of [32] is the following Lemma which will allow us to define appropriate Sobolev spaces where the operator \mathcal{L}_{V_f} has nice spectral properties.

Lemma 4.1 (Escape function). Let $N_0, N_1 > 4||f||_{\mathcal{C}^0}$ be two elements in \mathbb{R} . Then, there exist $c_0 > 0$ (depending on (M, g) but not on N_0 and N_1) and a smooth function $m(x, \xi)$ in $\mathcal{C}^{\infty}(T^*M)$ with bounded derivatives and which

- takes values in $[-2N_0, 2N_1]$,
- is 0 homogeneous for $\|\xi\|_x \ge 1$,

- $is \leq -\frac{N_0}{2}$ on a conic neighborhood of Γ_- (for $\|\xi\|_x \geq 1$), $is \geq \frac{N_1}{2}$ on a conic neighborhood of Γ_+ (for $\|\xi\|_x \geq 1$),

and such that there exists $R_0 > 0$ for which the escape function

$$G_m(x,\xi) := m(x,\xi)\log(1+\|\xi\|_x^2)$$

verifies, for every (x,ξ) in T^*M with $\|\xi\|_x \geq R_0$,

$$X_{H_f}(G_m)(x,\xi) \le -C_N := -c_0 \min\{N_0, N_1\}.$$

Now that we have settled the dynamical framework precisely in section 3, the construction of the function G_m closely follows the one from [32]. For the sake of exposition, we postpone the detailed proof of this result to appendix A, and we just mention the key ingredients: (1) f is strictly decreasing along the flow, (2) there exists a \mathcal{C}^1 chart of adapted coordinates (see paragraph 3.5), (3) the attractor and repeller of the Hamiltonian flow (Lemmas 3.6 and 3.8) are compact. Lemma 4.1 is in fact the only step in the construction of the anisotropic Sobolev space where one uses the dynamical properties of the flow under consideration.

4.1.2. Anisotropic Sobolev spaces. Let us now define the corresponding anisotropic Sobolev spaces. We fix $N_0, N_1 > 4||f||_{\mathcal{C}^0}$ large and we set

(10)
$$A_m(x,\xi) := \exp G_m(x,\xi),$$

where $G_m(x,\xi)$ is given by Lemma 4.1. Following paragraph 1.1.2 in [32], one can define the following anisotropic Sobolev space

$$H^m(M) := \operatorname{Op}(A_m)^{-1}(L^2(M)),$$

where $Op(A_m)$ is a pseudodifferential operator⁵ with principal symbol A_m .

We now briefly collect some facts concerning these spaces and we refer to [31, Sect. 3.2] for more properties of these spaces. The space $H^m(M)$ is endowed with a Hilbert structure induced by the Hilbert structure on $L^2(M)$. The space

$$H^{-m}(M) = \operatorname{Op}(A_m)L^2(M)$$

is the topological dual of $H^m(M)$. The anisotropic Sobolev space $H^m(M)$ is a reflexive space. Finally, one has

$$C^{\infty}(M) \subset H^m(M) \subset \mathcal{D}'(M),$$

where the injections are continuous.

⁵Note that this requires to deal with symbols of variable orders whose symbolic calculus was described in Appendix A of [31]. This can be done as the symbol $m(x,\xi)$ belongs to the standard class of symbols $S^{0}(T^{*}M).$

4.1.3. Anisotropic Sobolev spaces of currents. Let $0 \le k \le n$. We consider the vector bundle $\Lambda^k(T^*M)$ of exterior k forms. We define $\mathbf{A}_m^{(k)}(x,\xi) := A_m(x,\xi)\mathbf{Id}$ belonging to $\mathrm{Hom}(\Lambda^k(T^*M))$. We fix the inner product $\langle,\rangle_{g^*}^{(k)}$ on $\Lambda^k(T^*M)$ which is induced by the metric g on M.

This allows to define the Hilbert space $L^2(M, \Lambda^k(T^*M))$ and to introduce an anisotropic Sobolev space of currents by setting

$$\mathcal{H}_k^m(M) = \operatorname{Op}(\mathbf{A}_m^{(k)})^{-1} L^2(M, \Lambda^k(T^*M)),$$

where $\operatorname{Op}(\mathbf{A}_m^{(k)})$ is a pseudodifferential operator with principal symbol $\mathbf{A}_m^{(k)}$. We refer to [28, App. C.1] for a brief reminder of pseudodifferential operators with values in vector bundles. In particular, adapting the proof of [31, Cor. 4] to the vector bundle valued framework, one can verify that $\mathbf{A}_m^{(k)}$ is an elliptic symbol, and thus $\operatorname{Op}(\mathbf{A}_m^{(k)})$ can be chosen to be invertible. Mimicking the proofs of [31], we can deduce some properties of these spaces of currents. First of all, they are endowed with a Hilbert structure inherited from the L^2 -structure on M. The space

$$\mathcal{H}_k^m(M)' = \operatorname{Op}(\mathbf{A}_m^{(k)}) L^2(M, \Lambda^k(T^*M))$$

is the topological dual of $\mathcal{H}_k^m(M)$ which is in fact reflexive. We also note that the space $\mathcal{H}_k^m(M)$ can be identified with $H^m(M) \otimes_{\mathcal{C}^{\infty}(M)} \Omega^k(M)$. Finally, one has

$$\Omega^k(M) \subset \mathcal{H}_k^m(M) \subset \mathcal{D}'^{k}(M)$$

where the injections are continuous.

4.2. **Identifying the dual.** Recall that the Hodge star operator is the unique isomorphism $\star_k : \Lambda^k(T^*M) \to \Lambda^{n-k}(T^*M)$ such that, for every ψ_1 in $\Omega^k(M)$ and ψ_2 in $\Omega^{n-k}(M)$,

$$\int_{M} \psi_1 \wedge \psi_2 = \int_{M} \langle \psi_1, \star_k^{-1} \psi_2 \rangle_{g^*(x)}^{(k)} \omega_g(x),$$

where $\langle ., . \rangle_{g^*(x)}^{(k)}$ is the induced Riemannian metric on $\Lambda^k(T^*M)$. In particular, \star_k induces an isomorphism from $\mathcal{H}_k^m(M)'$ to $\mathcal{H}_{n-k}^{-m}(M)$, whose Hilbert structure is given by the scalar product

$$(\psi_1, \psi_2) \in \mathcal{H}_{n-k}^{-m}(M)^2 \mapsto \langle \star_k^{-1} \psi_1, \star_k^{-1} \psi_2 \rangle_{\mathcal{H}_k^m(M)'}.$$

Thus, the topological dual of $\mathcal{H}_k^m(M)$ can be identified with $\mathcal{H}_{n-k}^{-m}(M)$, where, for every ψ_1 in $\Omega^k(M)$ and ψ_2 in $\Omega^{n-k}(M)$, one has the following duality relation:

$$\langle \psi_1, \psi_2 \rangle_{\mathcal{H}_k^m \times \mathcal{H}_{n-k}^{-m}} = \int_M \psi_1 \wedge \psi_2 = \langle \operatorname{Op}(\mathbf{A}_m^{(k)}) \psi_1, \operatorname{Op}(\mathbf{A}_m^{(k)})^{-1} \star_k^{-1} \overline{\psi_2} \rangle_{L^2} = \langle \psi_1, \star_k^{-1} \psi_2 \rangle_{\mathcal{H}_k^m \times (\mathcal{H}_k^m)'}.$$

4.3. **Discrete spectrum.** The main result on the spectral properties of $-\mathcal{L}_{V_f}^{(k)}$ acting on these anisotropic spaces is the following Proposition:

Proposition 4.2 (Discrete spectrum). The operator $-\mathcal{L}_{V_f}^{(k)}$ defines a maximal closed unbounded operator on $\mathcal{H}_k^m(M)$,

$$-\mathcal{L}_{V_f}^{(k)}: \mathcal{H}_k^m(M) \to \mathcal{H}_k^m(M),$$

with domain given by $\mathcal{D}(-\mathcal{L}_{V_f}^{(k)}) := \{u \in \mathcal{H}_k^m(M) : -\mathcal{L}_{V_f}^{(k)}u \in \mathcal{H}_k^m(M)\}$. It coincides with the closure of $-\mathcal{L}_{V_f}^{(k)} : \Omega^k(M) \to \Omega^k(M)$ in the graph norm for operators. Moreover, there exists a constant C_0 in \mathbb{R} (that depends on the choice of the order function $m(x,\xi)$) such that $-\mathcal{L}_{V_f}^{(k)}$ has empty spectrum for $\text{Re}(z) > C_0$. Finally, the operator

$$-\mathcal{L}_{V_f}^{(k)}: \mathcal{H}_k^m(M) \to \mathcal{H}_k^m(M),$$

has a discrete spectrum with finite multiplicity in the domain

$$\operatorname{Re}(z) > -C_N + C,$$

where C > 0 depends only the choice of the metric (and the local coordinate charts) and $C_N > 0$ is the constant from Lemma 4.1.

The second part on the discrete spectrum is obtained by showing that the operator is $(-\mathcal{L}_{V_f}^{(k)} - z)$ is a Fredholm operator of index 0 depending analytically on z in the corresponding half plane [48, 78]. In the case of Anosov flows, the proof of this result was given by Faure-Sjöstrand in [32, Sect. 3] for k = 0 while the extension to the case of currents was done by Dyatlov-Zworski in [28, Sect. 3]. Note that the proofs in both references are of slightly different nature but they both crucially rely on the properties of the escape function used to define the anisotropic space $\mathcal{H}_k^m(M)$. Up to some adaptations to deal with the case of currents, we can in fact follow the proof of [32] for the extension to the case of currents – see appendix B for a brief account on the proof of Faure and Sjöstrand.

We now list some properties of this spectrum. As in [32, Th. 1.5], we can show that the eigenvalues (counted with their algebraic multiplicity) and the eigenspaces of $-\mathcal{L}_{V_f}^{(k)}$: $\mathcal{H}_k^m(M) \to \mathcal{H}_k^m(M)$ are in fact independent of the choice of escape function. For every $0 \le k \le n$, we call the eigenvalues the **Pollicott-Ruelle resonances of index** k. For later use, we will write

$$\mathcal{R}_k(f,g) := \{ \text{Pollicott-Ruelle resonances of index } k \} \subset \mathbb{C}.$$

In other words, these complex numbers are the poles of the meromorphic extension of the resolvent

$$\left(-\mathcal{L}_{V_f}^{(k)}-z\right)^{-1}:\Omega^k(M)\to\mathcal{D}'^{,k}(M).$$

Finally, we note that, by duality, the same spectral properties holds for the dual operator

$$(11) \qquad (-\mathcal{L}_{V_f}^{(k)})^{\dagger} = -\mathcal{L}_{V_{-f}}^{(n-k)} : \mathcal{H}_{n-k}^{-m}(M) \to \mathcal{H}_{n-k}^{-m}(M).$$

5. Computing the spectrum

In sections 5 to 7, we aim at describing the eigenvalues and the eigenmodes in great details. In order to explain the idea in a simpler manner, we start with the simpler question of determining the Pollicott-Ruelle resonances of $-\mathcal{L}_{V_f}^{(k)}$ without describing their corresponding eigenmodes and/or their algebraic multiplicity. In sections 6 and 7, we will refine the arguments presented here in order to obtain a complete description of the

spectrum. Yet, in order to emphasize the main ideas of the proof, we start with this simpler question, and the main result of this section reads:

Proposition 5.1. Suppose that (f,g) is a smooth Morse pair inducing a Morse-Smale gradient flow. Then, one has, for every $0 \le k \le n$,

$$\mathcal{R}_k(f,g) = \mathcal{I}_k := \bigcup_{a \in \text{Crit}(f)} \mathcal{I}_k(a),$$

where, for every a in Crit(f) of index $0 \le r \le n$,

(12)
$$\mathcal{I}_k(a) := \bigcup_{I \subset \{1, \dots, r\}, \ J \subset \{r+1, \dots, n\}: |J|-|I|=k-r} \left\{ -\sum_{j=1}^n \alpha_j |\chi_j(a)|: \ \forall j \in I \cup J, \ \alpha_j \ge 1 \right\}.$$

In other words, up to the multiplicities, we are able to determine exactly the Pollicott-Ruelle spectrum. Building on the construction of section 4, we shall now give the proof of this Proposition. We emphasize that the existence of a smooth linearizing chart is crucial in our proof. With lower regularity (say \mathcal{C}^1), we would not a priori obtain such a precise description. We proceed in several steps. First, we start with a simple propagation Lemma which will be used at several stages of our proof. Then, we use our a priori information on the regularity of the resonant states to show the inclusion $\mathcal{R}_k(f,g) \subset \mathcal{I}_k$. Finally, by considering appropriate test functions, we obtain the converse inclusion $\mathcal{I}_k \subset \mathcal{R}_k(f,g)$.

5.1. **Propagation Lemma.** In our proof, we intend to use a simple propagation Lemma that we will now prove.

Lemma 5.2 (Propagation Lemma). Let $0 \le k \le n$, let z in \mathbb{C} and let $u \in \mathcal{D}'^{,k}(M)$ be a solution of $\mathcal{L}_{V_f}^{(k)}u = zu$. If $u|_U = 0$ where $U \subset M$ is some open subset then u vanishes on the larger open subset $\bigcup_{t \in \mathbb{R}} \varphi_f^t(U)$.

Proof. We shall establish the result by a duality argument. First, we note that, for every ψ in $\Omega^{n-k}(M)$,

$$\frac{d}{dt} \left\langle \varphi_f^{t*} u, \psi \right\rangle = \left\langle \varphi_f^{t*} \mathcal{L}_{V_f}^{(k)} u, \psi \right\rangle = z \left\langle \varphi_f^{t*} u, \psi \right\rangle.$$

Hence, solving the ODE yields

$$\varphi_f^{t*}u = e^{tz}u, \ \forall t \in \mathbb{R}.$$

Assume now $x \in \bigcup_{t \in \mathbb{R}} \varphi_f^t(U)$. It means that there is some $t_0 \in \mathbb{R}$ such that $x \in \varphi_f^{t_0}(U)$ which is an open subset of M. Let ψ be an element in $\Omega^{n-k}(\varphi_f^{t_0}(U))$. We have the identity

$$\langle u, \psi \rangle = \langle (\varphi_f^{t_0})^* u, (\varphi_f^{-t_0})_* \psi \rangle = \langle e^{t_0 z} u, (\varphi_f^{-t_0})_* \psi \rangle = 0$$

since supp $((\varphi_f^{-t_0})_*\psi) \subset U$ and $u|_U = 0$.

- 5.2. **Proving** $\mathcal{R}_k(f,g) \subset \mathcal{I}_k$. We let $0 \leq k \leq n$. Let $z = \lambda + i\gamma$ be an element in $\mathcal{R}_k(f,g)$. We will prove that z belongs to \mathcal{I}_k via a kind of induction argument on the set of critical points. For that purpose, we order this set $a_1 < a_2 < \ldots < a_K$ by increasing order as in the proof of Lemma 3.8. In order to clarify the main idea, we will explain first the argument in the case of critical points of index 0, i.e. minima of f. Then, we will treat the general case using the pull-back Theorem of Hörmander [50].
- 5.2.1. Smoothness near minima. From the definition of $\mathcal{R}_k(f,g)$, one knows that there exists u in $\mathcal{H}_k^m(M)$ such that $\mathcal{L}_{V_f}^{(k)}u = -zu$. From [32, Th. 1.5], we know that the spectrum and the generalized eigenvalues are intrinsic, i.e. they do not depend on the choice of the order function $m(x,\xi)$. Thus, for any given N, we know from Sobolev injections that, if we pick N_1 large enough in the definition of m, u is of class \mathcal{C}^N in a small neighborhood of any minimum. The neighborhood may depend on the choice of N_1 . Yet, by a propagation argument, this remains true on the image of this neighborhood under the flow φ_f^t . In other words, u is smooth in the neighborhood of any local minimum of f, i.e. close to every point a of index r(a) = 0. We have then the following alternative
 - \bullet either u identically vanishes in a neighborhood of every point of index 0,
 - or there exists w_0 in a small neighborhood of a point a of index 0 such that w_0 belongs to the neighborhood of adapted coordinates of paragraph 3.5 and $u(w_0) \neq 0$.

Suppose that we are in the second case. Following the proof of the propagation Lemma, we write that, for every $t \ge 0$,

$$u \circ \varphi_f^{-t}(w_0) = e^{(\lambda + i\gamma)t} u(w_0).$$

Equivalently, if we set $t = -\ln s$ for $0 < s \le 1$, one has

(13)
$$u \circ \varphi_f^{\ln s}(w_0) = s^{-(\lambda + i\gamma)} u(w_0).$$

Recall that, if we integrate the flow of $\varphi_f^{\ln s}$ in the adapted coordinates $w=(y_i)_{1\leq i\leq n}$ of paragraph 3.5, we get

$$\varphi_f^{\ln s}(w) = (s^{\chi_i(a)}y_i)_{i=1,\dots,n}.$$

Hence, when we look at equality (13), the flow acts on the $(y_i)_{1 \le i \le n}$ variables via scalings with **different speeds** depending on the various Lyapunov exponents. We would now like to conclude that equality (13) combined with $u(w_0) \ne 0$ implies that z belongs to $\mathcal{I}_k(a)$.

For that purpose, we will use Taylor expansion of u in both even and odd variables as u is a **smooth form** near a. Indeed, we think of exterior forms $u \in \Omega^k(M)$ as superfunctions, i.e. as elements

$$u(y_1,\ldots,y_n;dy_1,\ldots,dy_n)$$

of the Grassmann algebra $C^{\infty}(M)[dy_1,\ldots,dy_n]$ over the algebra $C^{\infty}(M)$. Hence, as u is smooth, it admits a Taylor expansion in both **even** (y_1,\ldots,y_n) and **odd** variables (dy_1,\ldots,dy_n) . The Taylor formula reads :

$$u(y; dy) \sim \sum_{(\alpha, \beta) \leq N} u_{\alpha, \beta} y^{\alpha} dy^{\beta}$$

where (α, β) are multi-indices and the polynomials are of bounded degree in the odd variables dy. Note that this Taylor expansion in the adapted coordinate charts makes sense because we are considering smooth change of variables. We now introduce n scaling variables $(s_1, \ldots, s_n) \in \mathbb{R}^n$. Then, with the conventions of paragraph 3.5, we consider the map $\Phi: (s_j, y_j)_{1 \le j \le n} \in (-1, 1)^n \times W_a \mapsto (s_j y_j)_{1 \le j \le n} \in W_a$ which acts by pull-back as:

(14)
$$\Phi^* u(s, x, dx) = u(s_j y_j, s_j dy_j)$$

which implies that $\Phi^*u(s, y, dy)$ is **smooth** in $s = (s_1, \ldots, s_n) \in \mathbb{R}^n$. However our goal is to study the asymptotic behaviour of $u(s^{\chi_j(a)}y_j, s^{\chi_j(a)}dy_j)$ near s = 0. Thanks to Lemma C.3, we can deduce that this expression has a polyhomogeneous expansion in s. Combining this observation to equation (13) (with $u(w_0) \neq 0$) and to Lemma C.4, we deduce that

$$\gamma = 0$$
, and $\lambda = -\sum_{j=1}^{n} \alpha_j \chi_j(a)$,

for some $\alpha_j \in \mathbb{N}$. We also note that, as u belongs to $\Omega^k(M)$, at least k of the α_j must be ≥ 1 . This shows that z belongs to $\mathcal{I}_k(a)$ except if u was identically vanishing in a neighborhood of a. Our next step will be to generalize this argument near any critical point of f.

5.2.2. The general case. Assume now that, for all $i \leq j$, the eigenfunction u vanishes near a_i and assume the germ of u near a_{j+1} is non vanishing (see Lemma 5.3 below). In that case, we would like to prove that z is of the form:

$$z = -\sum_{l=1}^{n} \alpha_l |\chi_l(a_{j+1})|,$$

with some restrictions on the coefficients α_l . For that purpose, we start with the following central observation:

Lemma 5.3. Let $u \in \mathcal{D}'^{,k}(M)$ be some eigencurrent of $-\mathcal{L}_{V_f}^{(k)}$ acting on $\mathcal{H}_k^m(M)$. If u vanishes in some neighborhood of all a_i for $i \leq j$, then u restricted to the level $f^{-1}(< f(a_{j+1}))$ vanishes. Moreover, if the germ $u|_{V_{a_{j+1}}} \neq 0$ (for the adapted chart $\kappa_{a_{j+1}}$: $V_{a_{j+1}} \to W_{a_{j+1}}$ defined in paragraph 3.5), then the germ $u|_{V_{a_{j+1}}}$ is supported in the germ of unstable manifold $W^u(a_{j+1}) \cap V_{a_{j+1}}$.

Remark 5.4. A first consequence of this Lemma is that there is necessarily a critical point a in a neighborhood of which u does not vanish.

Proof. Assume without loss of generality that $f(a_{j+1}) = 0$. The level $f^{-1}(<0)$ contains only the critical points $\{a_1, \ldots, a_j\}$. Moreover, since the value of the potential f is monotonic along the flow it follows that the level $f^{-1}(<0)$ is contained in the union of unstable manifolds $\bigcup_{i \leq j} W^u(a_i)$. Hence, by the propagation Lemma, $u|_{\{f^{-1}(<0)\}} = 0$. Now consider some open set V in $V_{a_{j+1}}$ which does not interset $W^u(a_{j+1})$. Using the facts that f is excellent that $W^u(a_{j+1})$ is an embedded submanifold and that f must increase along the flow, one knows that, for every x in V, $x_- = \lim_{t \to -\infty} \varphi_f^t(x)$ belongs to $\{a_1, \ldots, a_j\}$. Using

the propagation Lemma one more time, one can then deduce that $u_{|V} = 0$. This is valid for any open set $V \subset V_{a_{j+1}}$ which does not intersect $W^u(a_{j+1})$. In other words, $u_{|U}$ is supported in the germ of unstable manifold $W^u(a_{j+1}) \cap V_{a_{j+1}}$, which concludes the proof of the Lemma.

We want to proceed as in the first step of the induction. In other words, we let ψ be a smooth test form in $\Omega^{n-k}(M)$ which is compactly supported in $V_{a_{j+1}}$. We can then write, for every s in (0,1],

$$\langle (\varphi_f^{\ln s})^* u, \psi \rangle = s^{-(\lambda + i\gamma)} \langle u, \psi \rangle.$$

As we made the assumption that u is not identically vanishing in the neighborhood of a_{j+1} , we can choose ψ compactly supported near a_{j+1} such that $\langle u, \psi \rangle \neq 0$. Then, we would like to prove that the left-hand side of the equality admits a polyhomogeneous expansion in s which is indexed by the set $\mathcal{I}_k(a_{j+1})$. Combining this to Lemma C.4, we would then deduce that z is of the expected form. Thus, our last task is to prove that $\langle (\varphi_f^{\ln s})^* u, \psi \rangle$ admits a polyhomogeneous expansion indexed by $\mathcal{I}_k(a_{j+1})$. For that purpose, we shall work using the local coordinates (x,y) defined in paragraph 3.5. We denote by \tilde{u} the image of u in this chart. From Lemma 5.3, this defines a current which is carried in $(-\delta_a, \delta_a)^n \cap \{x = 0\}$. This can be extended into a current defined on $\tilde{W}_a := \mathbb{R}^{r(a)} \times (-\delta_a, \delta_a)^{n-r(a)}$ by setting $\tilde{u} = 0$ outside $(-\delta_a, \delta_a)^n$. Then, we introduce the following map

$$\Phi: (s_1, \dots, s_n; x, y) \in (-1, 1)^n \times \tilde{W}_a \mapsto (s_i^{-1} x_i, s_j x_j) \in \tilde{W}_a.$$

Note that this is well defined except if $s_i = 0$ for some $1 \le i \le r(a)$. We also define the partial maps:

$$\Phi^1: (s_1, \dots, s_n; x, y) \in (-1, 1)^n \times \tilde{W}_a \mapsto (x_j, s_j y_j) \in \tilde{W}_a,$$

and

$$\Phi^2: (s_1, \dots, s_n; x, y) \in (-1, 1)^n \times \tilde{W}_a \mapsto (s_i x_i, y_i) \in \tilde{W}_a.$$

Contrary to Φ , these two maps are well defined for **s** belonging to the whole set $(-1,1)^n$. Let **s** be a point in $(-1,1)^n$ with all entries which are non vanishing. In that case, we can write

(15)
$$\langle \Phi(\mathbf{s})^* \tilde{u}, \tilde{\psi} \rangle = \langle \Phi^2(\mathbf{s})_* \Phi^1(\mathbf{s})^* \tilde{u}, \tilde{\psi} \rangle = \langle \Phi^1(\mathbf{s})^* \tilde{u}, \Phi^2(\mathbf{s})^* \tilde{\psi} \rangle.$$

This is valid as long as $s_j \neq 0$ for every $1 \leq j \leq n$. Our next step is to show that this extends as a smooth function on $(-1,1)^n$. From the previous expression, one can observe that the main concern is to be able to study the smoothness of $\Phi^1(\mathbf{s})^*\tilde{u}$ in the variable $(s_j)_j \in (-1,1)^n$. Recall that u is an eigenvector of $-\mathcal{L}_{V_f}^{(k)}$ acting on a certain anisotropic Sobolev space $\mathcal{H}_k^m(M)$. According to [32, Th. 1.5], the eigenmodes are independent of the choice of the order function $m(x,\xi)$ satisfying the assumptions of Lemma 4.1. As in the case of minima, letting $N_1 \to +\infty$ in this Lemma and using Lemma 5.3, one finds that the wave front set $WF(\tilde{u})$ of \tilde{u} satisfies the following

(16)
$$WF(\tilde{u}) \subset \left\{ (0, y, \xi, 0) \in T^* \tilde{W}_a : \xi \neq 0 \right\}.$$

We would now like to define the pull-back of \tilde{u} under the map Φ^1 , and, for that purpose, we shall apply Hörmander's pullback Theorem [50, Th. 8.2.4] – see also [15]. Hence, we have to compute the normal $N_{\Phi^1}^* \subset T^* \tilde{W}_a$ of the map Φ^1 ,

$$\begin{split} N_{\Phi^1}^* &= \left\{ (x, s_j y_j; \xi, \eta) \text{ such that } (0, 0, 0) = (\xi, \eta) \circ d_{(\mathbf{s}, x, y)} \Phi, (\xi, \eta) \neq (0, 0) \right\} \\ &= \left\{ (x, s_j y_j; \xi, \eta) \text{ such that } (0, 0, 0) = (\xi, \eta) \circ \begin{pmatrix} 0 & 1 & 0 \\ y & 0 & (s_j)_j \end{pmatrix}, (\xi, \eta) \neq (0, 0) \right\} \\ &= \left\{ (x, s_j y_j; \xi, \eta) \text{ such that } \left(\sum_j y_j \eta^j, \xi, \sum_j s_j \eta^j \right) = (0, 0, 0), (\xi, \eta) \neq (0, 0) \right\} \\ &= \left\{ (x, s_j y_j; 0, \eta) \text{ such that } \eta \neq 0, \sum_j s_j \eta^j = \sum_j y_j \eta^j = 0 \right\}. \end{split}$$

In particular, from (16), $N_{\Phi^1}^* \cap WF(\tilde{u})$ is empty. Hence, we can apply Hörmander's pullback Theorem, i.e. $(\Phi^1)^*\tilde{u}$ is well defined and its wave front set is contained in

$$(\Phi^{1})^{*}WF(\tilde{u}) = \left\{ \left(\mathbf{s}, x, y; \sum_{j} y_{j} \eta^{j}, \xi, \sum_{j} s_{j} \eta^{j} \right) \text{ such that } \left(x, \sum_{j} s_{j} y_{j}; \xi, \eta \right) \in WF(\tilde{u}) \right\}$$

$$\subset \left\{ (\mathbf{s}, 0, y; 0, \xi, 0) \text{ such that } \xi \neq 0 \right\}.$$

As $\tilde{\psi}$ is a smooth test form, we note that $(\Phi^1)^*\tilde{u} \wedge (\Phi^2)^*\tilde{\psi}$ is a current of degree n on $(-1,1)^n \times \tilde{W}_a$ whose wave front set is included in $(\Phi^1)^*WF(\tilde{u})$. Consider now the push-forward of this current under the map

$$\mathbf{p}:(x,y,\mathbf{s})\in \tilde{W}_a\times (-1,1)^n\mapsto \mathbf{s}\in (-1,1)^n$$

By the push-forward Theorem [50, 15], the wave front set of the pushforward distribution is included in

$$\mathbf{p}_*\left((\Phi^1)^*WF(\tilde{u})\right) = \left\{ (\mathbf{s}; \sigma) \text{ such that } (\mathbf{s}, x, y; \sigma, 0, 0) \in (\Phi^1)^*WF(\tilde{u}), \ \sigma \neq 0 \right\} = \emptyset.$$

In other words, the pushforward distribution is a smooth function in the variable $\mathbf{s} \in (-1,1)^n$. In particular, according to (15), $\langle \Phi(\mathbf{s})^* \tilde{u}, \tilde{\psi} \rangle$ has a well-defined Taylor expansion in \mathbf{s} around 0. Then, we can combine Lemma C.3 to the fact that, in our system of adapted coordinates, the reparametrized flow $\varphi_f^{\ln s}$ can be written $(s^{\chi_j(a)}x_j, s^{\chi_j(a)}y_j)$. From that, we deduce the expected property, i.e. $\langle (\varphi_f^{\ln s})^* u, \psi \rangle$ admits a polyhomogeneous expansion indexed by $(\alpha.|\chi(a)|)_{\alpha \in \mathbb{N}^n}$. In order to conclude the proof, we should observe that u(x,y,dx,dy) is of degree k and $\psi(x,y,dx,dy)$ of degree n-k. This forces that some of the α_j do not vanish when we express z as a combination of the Lyapunov exponents, i.e. z must in fact belong to the set $\mathcal{I}_k(a_{j+1})$. This concludes the proof of the first inclusion, i.e. $\mathcal{R}_k(f,g) \subset \mathcal{I}_k$.

Remark 5.5. We shall use this kind of arguments several times in the following. We observe that we have just been able to prove that $\langle \varphi_f^{\ln s*} u, \psi \rangle$ has a polyhomogeneous expansion indexed by the set $\mathcal{I}_k(a_{j+1})$ and that our proof only made use of the facts that the support

of u near a_{j+1} was included in $W^u(a_{j+1})$ and that its wave front is included in E_u^* near a_{j+1} .

5.3. Asymptotic expansion of the correlation function. Before proving the converse inclusion, let us draw some consequences of the fact that $\mathcal{R}_k(f,g) \subset \mathcal{I}_k$ following the lines of [31]. From (50) in the appendix, we know that $(\varphi_f^{-t})^*$ generates also a strongly continuous semi-group from $\mathcal{H}_k^m(M)$ to $\mathcal{H}_k^m(M)$ for every $1 \leq k \leq n$ whose norm is bounded by e^{tC_0} . Fix now $\Lambda > 0$. Suppose without loss of generality that $-\Lambda$ does not belong to \mathcal{I}_k . From Proposition 4.2, we now observe that, for every $\Lambda > 0$, one can find a weight function $m(x, \xi)$ such that the operator

$$-\mathcal{L}_{V_f}^{(k)}:\mathcal{H}_k^m(M)\to\mathcal{H}_k^m(M)$$

has only discrete spectrum with finite multiplicity in the half plane $\text{Re}(\lambda) > -\Lambda$. Moreover, from the fact that $\mathcal{R}_k(f,g) \subset \mathcal{I}_k$, the operator has only finitely many eigenvalues in this region which are real and nonpositive. We denote by $-\lambda_i^{(k)}$ the eigenvalues of $-\mathcal{L}_{V_f}^{(k)}$ (counted with their algebraic multiplicities). Note that each eigenvalue may a priori be associated with a Jordan block of size $d_i^{(k)} \geq 1$. Following [48, App. A], we fix a Jordan path in \mathbb{C} which separates the eigenvalues in the half plane $\text{Re}(\lambda) > -\Lambda$ from the rest of the spectrum. Then, according to this reference, the spectral projector associated with this finite part of the spectrum can be written as

$$\Pi_{\Lambda}^{(k)} := \int_{\gamma} (-\mathcal{L}_{V_f}^{(k)} - z)^{-1} dz.$$

We can then split the operator $-\mathcal{L}_{V_{\epsilon}}^{(k)}$ as follows:

$$-\mathcal{L}_{V_f}^{(k)} := \Pi_{\Lambda}^{(k)} \circ (-\mathcal{L}_{V_f}^{(k)}) \circ \Pi_{\Lambda}^{(k)} + (\operatorname{Id} - \Pi_{\Lambda}^{(k)}) \circ (-\mathcal{L}_{V_f}^{(k)}) \circ (\operatorname{Id} - \Pi_{\Lambda}^{(k)}).$$

According to [30, p. 244-246], the spectrum of the operator $(\operatorname{Id} - \Pi_{\Lambda}^{(k)}) \circ (-\mathcal{L}_{V_f}^{(k)}) \circ (\operatorname{Id} - \Pi_{\Lambda}^{(k)})$ is contained in the half plane $\operatorname{Re}(\lambda) < -\Lambda$ while the finite rank part can be written as

$$(17) \qquad \Pi_{\Lambda}^{(k)} \circ (-\mathcal{L}_{V_f}^{(k)}) \circ \Pi_{\Lambda}^{(k)} = \sum_{i:\lambda_i^{(k)} \le \Lambda} \left(\sum_{l=1}^{d_i^{(k)}} -\lambda_i^{(k)} |u_{i,l}^{(k)}\rangle \langle v_{i,l}^{(k)}| + \sum_{l=1}^{d_i^{(k)}-1} |u_{i,l}^{(k)}\rangle \langle v_{i,l+1}^{(k)}| \right)$$

where

- $(u_{i,l}^{(k)})_{\lambda_i^{(k)} \leq \Lambda, l=1,\dots d_i^{(k)}}$ belongs to $\mathcal{H}_k^m(M) \subset \mathcal{D}'^{,k}(M)$,
- $(v_{i,l}^{(k)})_{\lambda_i^{(k)} \leq \Lambda, l=1,\dots d_i^{(k)}}$ belongs to $\mathcal{H}_{n-k}^{-m}(M) \subset \mathcal{D}'^{n-k}(M)$,
- $|u\rangle\langle v|$: $\psi \in \mathcal{H}_k^m(M) \mapsto \langle v, \psi\rangle u \in \mathcal{H}_k^m(M)$

Recall from [32, Th. 1.5] that these "generalized eigendistributions" are intrinsic and that they do not depend on the choice of the order function m. We also note that the vectors $v_*^{(k)}$ give rise to a Jordan basis for the spectral decomposition of the dual operator acting on $\mathcal{H}_{n-k}^{-m}(M)$. We now want to relate this spectral decomposition to the correlation function from the introduction:

Proposition 5.6. Let $0 \le k \le n$. Then, for every $i \ge 0$, there is an integer $d_i^{(k)} \ge 1$ s.t. for any $\Lambda > 0$, for every $(\psi_1, \psi_2) \in \Omega^k(M) \times \Omega^{n-k}(M)$ and for every $t \ge 0$,

(18)
$$\langle (\varphi_f^{-t})^* \psi_1, \psi_2 \rangle = \sum_{i: \lambda_i^{(k)} < \Lambda} e^{-\lambda_i^{(k)} t} \sum_{l=0}^{d_i^{(k)} - 1} \frac{t^l}{l!} \sum_{j=l+1}^{d_i^{(k)}} \langle u_{i,j}^{(k)}, \psi_2 \rangle \langle v_{i,j+l}^{(k)}, \psi_1 \rangle + \mathcal{O}_{\psi_1, \psi_2, \Lambda}(e^{-\Lambda t}).$$

In fact, the result also holds for any ψ_1 in $\mathcal{H}_k^m(M)$ provided the parameters (N_0, N_1) involved in the definition of m are large enough.

Note that the sum is finite and that all the quantities involved in the sum are independent of the choice of the order function m. This expression gives us an asymptotic expansion for the correlation function at any order of precision. As was already explained, all the $\lambda_i^{(k)}$ appearing in the sum belong to the set $-\mathcal{I}_k \subset \mathbb{R}_+$. The rest of the article is devoted to a more precise understanding of the terms appearing in this asymptotic expansion. Yet, before that, let us prove this Proposition.

Proof. Fix $q \geq 1$. We first follow the arguments of [31, Th. 1] applied to the hyperbolic diffeomorphism $\varphi_q := \varphi_f^{-\frac{1}{q}}$ rather than to the generator $-\mathcal{L}_{V_f}^{(k)}$. Precisely, following this reference, we can verify that the order function m_{N_0,N_1} from Lemma A.1 satisfies the assumptions of [31, Lemma 2]. Then, following almost verbatim [31, section 3.2], we can deduce that the transfer operator

$$\varphi_q^* : \psi \in \mathcal{H}_k^m(M) \to \varphi_f^{-\frac{1}{q}*} \psi \in \mathcal{H}_k^m(M)$$

defines a bounded operator on the anisotropic space $\mathcal{H}_k^m(M)$ which can decomposed as

(19)
$$\varphi_q^* = \hat{r}_{m,q} + \hat{c}_{m,q},$$

where $\hat{c}_{m,q}$ is a compact operator and the remainder $\hat{r}_{m,q}$ has small operator norm bounded as : $\|\hat{r}_{m,q}\| \leq e^{\frac{C-\frac{C_N}{q}}{3}}$ (for some uniform C that may be slightly larger than the one from Proposition 4.2). Note that, for every $q \in \mathbb{N}$, we can make $\|\hat{r}_{m,q}\|$ arbitrarily small

from Proposition 4.2). Note that, for every $q \in \mathbb{N}$, we can make $\|\hat{r}_{m,q}\|$ arbitrarily small by choosing N large enough. The proof follows similar lines as for the definition of the spectrum of $-\mathcal{L}_{V_f}^{(k)}$ except that we deal with the propagator at discrete times instead of the generator. Again, we can verify that the spectrum is intrinsic, i.e. independent of the choice of order function. This is because the eigenvalues and associated spectral projectors correspond to the poles and residues of a discrete resolvent defined as an operator from $\Omega^k(M)$ to $\mathcal{D}'^{,k}(M)$ as follows: consider the series $\sum_{l=0}^{+\infty} e^{-lz} \varphi_q^{l*}$. Then, by the direct bound:

$$\|\sum_{l=0}^{+\infty} e^{-lz} \varphi_q^{l*} \psi\|_{\mathcal{H}_k^m(M)} \leqslant \sum_{l=0}^{+\infty} e^{-l\operatorname{Re}(z)} \|\varphi_q^*\|^l \|\psi\|_{\mathcal{H}_k^m(M)},$$

we deduce that, for $\operatorname{Re}(z)$ large enough, the series $\sum_{l=0}^{+\infty} e^{-lz} \varphi_q^{l*} \psi$ converges absolutely in $\mathcal{H}_k^m(M)$ for every test form $\psi \in \Omega^k(M)$. Hence, by the continuous injections $\Omega^k(M) \hookrightarrow$

 $\mathcal{H}_{k}^{m_{N_0,N_1}}(M) \hookrightarrow \mathcal{D}'^{,k}(M)$, the identity

$$(\mathrm{Id} - e^{-z}\varphi_q^*)^{-1} := \sum_{l=0}^{+\infty} e^{-lz}\varphi_q^{l*} : \Omega^k(M) \to \mathcal{D}'^{,k}(M)$$

holds true for Re(z) large enough. A consequence of the decomposition (19) is that the resolvent of φ_q^*

$$(\lambda - \varphi_a^*)^{-1} : \Omega^k(M) \to \mathcal{D}^{\prime,k}(M)$$

has a meromorphic extension from $|\lambda| > e^{C_0}$ to $\lambda \in \mathbb{C}$ with poles of finite multiplicity which correspond to the eigenvalues of the operator φ_q^* [31, Corollary 1]. In other words, $(\mathrm{Id} - e^{-z}\varphi_q^*)^{-1}: \Omega^k(M) \to \mathcal{D}'^{,k}(M)$ has a meromorphic extension from $\mathrm{Re}(z) > C_0$ (with $C_0 > 0$ large enough) to $z \in \mathbb{C}$ with poles of finite multiplicity. Denote by $\tilde{\pi}_{\lambda,q}^{(k)}$ the spectral projector of φ_q^* associated to the eigenvalue λ which is obtained from the contour integral formula:

$$\tilde{\pi}_{\lambda,q}^{(k)} = \frac{1}{2i\pi} \int_{\gamma} \left(\mu - \varphi_q^*\right)^{-1} d\mu$$

where γ is a small circle around λ . This corresponds to the residues of the discrete resolvent at $e^z = \lambda$. As φ_q^* commutes with $-\mathcal{L}_{V_f}^{(k)}$, we can deduce that the range of $\tilde{\pi}_{\lambda,q}^{(k)}$ is preserved by $-\mathcal{L}_{V_f}^{(k)}$. In particular, any eigenvalue z_0 of $-\mathcal{L}_{V_f}^{(k)}$ on that space must verify $e^{\frac{z_0}{q}} = \lambda$. As we know that any resonance of $-\mathcal{L}_{V_f}^{(k)}$ is real, we can deduce that the poles of $(\mathrm{Id} - e^{-z/q}\varphi_q^*)^{-1}$ belong to $\mathcal{R}_k(f,g) \subset \mathbb{R}$ modulo $2i\pi\mathbb{Z}$. Take now z_0 in $\mathcal{R}_k(f,g)$. We want to show that

(20)
$$\tilde{\pi}_{e^{z_0},1}^{(k)} = \pi_{z_0}^{(k)},$$

where $\pi_{z_0}^{(k)}$ is the spectral projector of $-\mathcal{L}_{V_f}^{(k)}$ associated to the eigenvalue z_0 . Equivalently, the spectral projectors are the same for both problems. Once it will be done, the proposition is just a consequence of decomposition (19) for q=1 when t is an integer as a consequence of [31, Corollary 1]. When t is a positive real number, we can conclude by writing $\varphi^{-t} = \varphi^{-[t]} \varphi^{-t+[t]}$ and by using (50) from the appendix.

In order to show (20), we first observe that $\tilde{\pi}_{e^{z_0/q},q}^{(k)} = \tilde{\pi}_{e^{z_0},1}^{(k)}$ and we decompose the resolvent $(z + \mathcal{L}_{V_f}^{(k)})^{-1}$ as follows:

$$(z + \mathcal{L}_{V_f}^{(k)})^{-1} = \sum_{l=0}^{+\infty} e^{-\frac{z}{q}} \varphi_q^* \int_0^{\frac{1}{q}} e^{-zt} \varphi_f^{-t*} dt = (\operatorname{Id} - e^{-\frac{z}{q}} \varphi_q^*)^{-1} \int_0^{\frac{1}{q}} e^{-zt} \varphi_f^{-t*} dt.$$

For $\operatorname{Re}(z)$ large enough, this expression makes sense viewed as an operator from $\Omega^k(M)$ to $\mathcal{D}'^{,k}(M)$. We have seen that it can be meromorphically continued to \mathbb{C} by using the fact that we have built a proper spectral framework and that we may pick N_0 and N_1 arbitrarily large. Consider now a small contour γ around z_0 containing no other elements

of $\mathcal{R}_k(f,g)$. Integrating over this contour tells us that, for every $q \geq 1$:

$$\pi_{z_0}^{(k)} = \tilde{\pi}_{e^{z_0/q}, q}^{(k)} q \int_0^{\frac{1}{q}} e^{-z_0 t} \varphi_f^{-t*} dt = \tilde{\pi}_{e^{z_0}, 1}^{(k)} \int_0^1 e^{-t \frac{z_0}{q}} \varphi_f^{-\frac{t}{q}*} dt.$$

As an operator on $\Omega^k(M)$, we can observe that $\int_0^1 e^{-t\frac{z_0}{q}} \varphi_f^{-\frac{t}{q}*} dt$ converges to the identity as $q \to +\infty$. Hence, $\pi_{z_0}^{(k)} = \tilde{\pi}_{e^{z_0},1}^{(k)}$ as expected.

5.4. **Proving** $\mathcal{I}_k \subset \mathcal{R}_k(f,g)$. Note that the formula we obtained for the correlation function in (18) shows that the function $s \in (0,1] \mapsto \langle \varphi^{\ln(s)*}\psi_1, \psi_2 \rangle$ admits an asymptotic expansion in $(\ln s)^l s^{\lambda}$ where $-\lambda$ belongs to the spectrum of $\mathcal{L}_{V_f}^{(k)}$ acting on the anisotropic space $\mathcal{H}_k^m(M)$. We also observe that this asymptotic expansion is valid for ψ_1 in $\mathcal{H}_k^m(M)$ and ψ_2 in $\Omega^{n-k}(M)$.

This implies the following **criteria**: If there is some pair $(\psi_1, \psi_2) \in \mathcal{H}_k^m(M) \times \Omega^{n-k}(M)$ such that s^{λ} shows up in the asymptotic expansion of $\langle \psi_1, \varphi^{\ln(s)*} \psi_2 \rangle$ near s = 0 then λ belongs to the spectrum of $\mathcal{L}_{V_f}^{(k)}$. Hence, for every fixed $0 \leq k \leq n$ and every $-\lambda \in \mathcal{I}_k$, we would show that $\lambda \in \mathcal{R}_k(f,g)$ if we could find some pair $(\psi_1, \psi_2) \in \mathcal{H}_k^m(M) \times \Omega^{n-k}(M)$ such that $\langle \varphi^{\log(s)*} \psi_1, \psi_2 \rangle$ has a polyhomogeneous expansion in s with s^{λ} appearing in the asymptotic expansion.

For that purpose, we fix a critical point a of index r and $0 \le k \le n$. We work with the adapted coordinates (x, y) introduced in paragraph 3.5. We let $0 \le \theta_1(x) \le 1$ (resp. $\theta_2(y)$) be a smooth function which is compactly supported in $(-\delta_a, \delta_a)^r$ (resp. $(-\delta_a, \delta_a)^{n-r}$) which is equal to 1 in a neighborhood of 0. We fix a multi-index $(\alpha, \beta) = (\alpha_1, \ldots, \alpha_r, \beta_{r+1}, \ldots, \beta_n)$, and two sets

$$I \subset \{1, \dots, r\}, \text{ and } J \subset \{r+1, \dots, n\},$$

such that |J| - |I| = k - r. In these coordinates, we define

$$\psi_1(x, y, dx, dy) = \theta_1(x)\theta_2(y)\delta_0^{(\alpha)}(x)y^{\beta} \left(\wedge_{i \notin I} dx_i \right) \wedge \left(\wedge_{i \in J} dy_i \right),$$

and

$$\psi_2(x, y, dx, dy) = \theta_1(x)\theta_2(y) \left(\wedge_{i \in I} dx_i \right) \wedge \left(\wedge_{i \notin J} dy_i \right).$$

Note that ψ_1 (more precisely its image in $\mathcal{D}'^{,k}(M)$) belongs to the anisotropic Sobolev space $\mathcal{H}_k^m(M)$ provided that we choose N_0 large enough in the definition of the order function $m(x,\xi)$. We can then write, for every $0 < s \le 1$,

(21)
$$\langle (\varphi_f^{\ln s})^* \psi_1, \psi_2 \rangle = \pm s^{\lambda_{\alpha,\beta,I,J}} \langle \delta_0(x) y^{\beta}, \theta_1(x) \theta_2(y) (\partial^{\alpha} \theta_1) ((s^{|\chi_i(a)|} x_i)_i) \theta_2 ((s^{|\chi_i(a)|} y_i)_i) \rangle$$
. where

$$\lambda_{\alpha,\beta,I,J} := \sum_{i=1}^{r} \alpha_i |\chi_i(a)| + \sum_{i=r+1}^{n} \beta_i |\chi_i(a)| + \sum_{1 \le i \le r: i \in I} |\chi_i(a)| + \sum_{r+1 \le i \le n: i \in J} |\chi_i(a)|.$$

Writing the Taylor expansion of (21), one finds that

$$\langle (\varphi_f^{\ln s})^* \psi_1, \psi_2 \rangle = \pm s^{\lambda_{\alpha,\beta,I,J}} \theta_1(0) \theta_2(0) \left\langle \delta_0^{(\alpha)}(x) y^{\beta}, \theta_1(x) \theta_2(y) \right\rangle (1 + o(1)).$$

Hence, if we choose θ_1 and θ_2 such that the leading term is not zero in this asymptotic expansion, we find that $-\lambda_{\alpha,\beta,I,J}$ belongs to $\mathcal{R}_k(f,g)$. We shall come back to this construction in section 7 where we will construct a family of linearly independent eigenmodes for each eigenspace.

6. Description of the Kernel

In the previous section, we found the Pollicott-Ruelle spectrum of any degree $0 \le k \le n$. Yet, we did not discuss the multiplicity or the existence of Jordan blocks in the spectrum. We will now consider this question in sections 6 and 7. The argument presented in these two sections is a refinement of the argument of section 5 and it relies on a Theorem of Schwartz [66] which describes the distributions carried by a smooth submanifold inside M. The general case is slightly involved combinatorially and it requires some nonresonance assumption. Hence, we will first expose the proof under the simplifying assumption that $\lambda = 0$. The case $\lambda \ne 0$ will follow similar lines (up to the resonance assumption) and it will be discussed in section 7. Precisely, we will show that there is no Jordan blocks in the kernel (Proposition 6.1) and that there exists a "canonical" basis of the kernel which is carried by the closure of the unstable manifolds $W^u(a)$ (Proposition 6.7 and 6.8).

6.1. **Jordan blocks.** Let us first show the absence of Jordan blocks in the kernel:

Proposition 6.1. Suppose that (f,g) is a smooth Morse pair which induces a Morse-Smale gradient flow. Let $0 \le k \le n$. Then, when acting on a convenient⁶ anisotropic space $\mathcal{H}_k^m(M)$, one has

$$\operatorname{Ker}(\mathcal{L}_{V_f}^{(k)}) = \operatorname{Ker}((\mathcal{L}_{V_f}^{(k)})^2).$$

In other words, there is no Jordan blocks in the kernel.

We start with the following Lemma:

Lemma 6.2. Let $1 \le j \le K$. Then, there exists an open neighborhood V_{a_j} of a_j such that, for every i < j with $r(a_i) \ge r(a_j)$, one has

$$W^u(a_i) \cap V_{a_i} = \emptyset.$$

Proof. Let $1 \leq j \leq K$. Let i < j satisfying $r(a_i) \geq r(a_j)$. The closure of $W^u(a_i)$ is a compact subset. In order to prove this Lemma, we suppose by contradiction that a_j belongs to $\overline{W^u(a_i)}$. It means that there exists a sequence $(x_p)_{p\geq 1}$ in $W^u(a_i)$ such that x_p converges to a_j as $p \to +\infty$. Without loss of generality, we can suppose that there exists a unique b in $\operatorname{Crit}(f)$ such that, for every $p \geq 1$, $x_p \in W^s(b)$. Using the conventions of Remark 3.7 for our fixed sequence $(x_p)_{p\geq 1}$, there exists a sequence of critical points $a_i = b_0, b_1, \ldots, b_l = b$ and a finite sequence of points u_m in $W^u(b_m) \cap W^s(b_{m+1})$ such that

$$\forall \epsilon_1 > 0, \exists p_0, \forall p \ge p_0, \ d\left(\mathcal{O}(x_p), \cup_{0 \le m \le l-1} \mathcal{O}(b_m)\right) < \epsilon_1,$$

⁶It means that there is a discrete spectrum for $Re(\lambda) < C$ if C > 0.

where $\mathcal{O}(x)$ denotes the orbit of x under the flow φ_f^t . In particular, as $x_p \to a_j$, this implies that $a_j = b_m$ for some $0 < m \le l$. By the Morse-Smale transversality assumption, this implies that $r(a_j) > r(a_i) = r(b_0)$ which gives the contradiction.

We can now give the proof of Proposition 6.1. Suppose by contradiction that there exists a Jordan block associated to the eigenvalue 0 for a certain degree of currents k. Then, it means that there exists $u_0 \neq 0$ and $u_1 \neq 0$ in our anisotropic Sobolev space of currents $\mathcal{H}_k^m(M)$ such that

$$\mathcal{L}_{V_f}^{(k)} u_0 = 0 \text{ and } \mathcal{L}_{V_f}^{(k)} u_1 = u_0.$$

Integrating these expressions, we find that, for all t in \mathbb{R} ,

$$(\varphi_f^t)^* u_0 = u_0 \text{ and } (\varphi_f^t)^* u_1 = u_1 + t u_0.$$

As in our computation of the spectrum, we let $t = \ln s$ with $0 < s \le 1$,

(22)
$$(\varphi_f^{\ln s})^* u_1 - u_1 = (\ln s) u_0.$$

As above, we order our critical points $a_1 < a_2 < \ldots < a_K$ using the fact that the critical values of f are distinct.

We now use this Lemma to get the expected contradiction. We fix $j \geq 0$. We suppose that u_0 is vanishing in a neighborhood of any critical point $(a_i)_{i\leq j}$ and that it does not vanish in a neighborhood of a_{j+1} . According to Lemma 5.3, we can deduce that $\sup(u_0)\cap V_{a_{j+1}}$ is included in $W^u(a_{j+1})$. Arguing as in paragraph 5.2.2 (i.e. via the pull-back Theorem of Hörmander), we can verify that $\langle (\varphi_f^{\ln s})^* u_0, \psi \rangle = \langle u_0, \psi \rangle$ has a bounded asymptotic expansion in s for ψ a smooth test form compactly supported in $V_{a_{j+1}}$. Moreover, we can choose ψ such that the right hand side of the equality does not vanish. Hence, the leading order of this expansion must be of degree 0. This implies that a_{j+1} is a critical point of index $r(a_{j+1}) = k$.

We would now like to prove that, near a_{j+1} , u_1 is also supported in $W^u(a_{j+1})$. We fix V an open subset of $V_{a_{j+1}}$ which does not intersect $W^u(a_{j+1})$. From Lemma 6.2, one knows that, for every x in V, there exists $i \leq j$ such that $a_i = \lim_{t \to -\infty} \varphi_f^t(x)$ and $r(a_i) < r(a_{j+1})$. Hence, we would conclude that $\sup(u_1) \cap V_{a_{j+1}}$ is included in $W^u(a_{j+1})$ if we could show that, for every $i \leq j$ with $r(a_i) < k$, u_1 identically vanishes in an open neighborhood of a_i .

Let $i_0 \leq j$ be an index with $r(a_{i_0}) < k$. Then, either $\operatorname{supp}(u_1) \cap V_{a_{i_0}}$ is included in $W^u(a_{i_0})$, or, as $\mathcal{L}_{V_f}u_1 = u_0 = 0$ on $f^{-1}(< f(a_{j+1}))$, we can deduce by propagation that there exists a critical point a of smaller index such that u_1 does not vanish in a neighborhood of a. Hence, without loss of generality, we can suppose that $\operatorname{supp}(u_1) \cap V_{a_{i_0}}$ is included in $W^u(a_{i_0})$. As $\mathcal{L}_{V_f}u_1 = 0$ in $V_{a_{i_0}}$, we can argue one more time as in paragraph 5.2.2. From that, we deduce that

$$\langle (\varphi_f^{\ln s})^* u_1, \psi \rangle = \langle u_1, \psi \rangle$$

has a bounded asymptotic expansion in s for every choice of ψ compactly supported in $V_{a_{i_0}}$. Using the fact that $r(a_{i_0}) < k$, we conclude that the left hand side must go to 0 as $s \to 0^+$. Thus, one has $\langle u_1, \psi \rangle = 0$ as expected from which we deduce that $\sup(u_1) \cap V_{a_{j+1}}$ is included in $W^u(a_{j+1})$.

Thanks to the fact that $\sup(u_1) \cap V_{a_{j+1}}$ is included in $W^u(a_{j+1})$ and to the fact that u_1 belongs to our family of anisotropic spaces, we can argue one more time as in paragraph 5.2.2. We find then that $\langle (\varphi_f^{\ln s})^* u_1, \psi \rangle$ has a bounded asymptotic expansion as $s \to 0^+$ for any smooth test function ψ supported near a_{j+1} . Using then that u_1 verifies equation (22), we can finally conclude that $\langle u_0, \psi \rangle = 0$ for every ψ supported near 0 which gives the contradiction to the fact that there exists a nontrivial Jordan block in the kernel.

Finally, let us conclude with the following fact which has been implicitely proved following the lines of paragraph 5.2.2:

Proposition 6.3. Suppose that (f,g) is a smooth Morse pair which induces a Morse-Smale gradient flow. Let $u \neq 0$ be an element of $\mathcal{H}_k^m(M)$ satisfying $\mathcal{L}_{V_f}^{(k)}u = 0$. Let a be the critical point of f satisfying the following properties:

- u does not vanish in any neighborhood of a,
- for every a' in Crit(f) satisfying f(a') < f(a), u identically vanishes near a'. Then, the Morse index of a equals k.
- 6.2. Exponential convergence of the correlation function. Before continuing our description of the kernel, let us draw a simple consequence of the fact that there is no Jordan blocks in the kernel of $\mathcal{L}_{V_f}^{(k)}$. Using the spectral expansion (18), one has, for every $\psi_1 \in \Omega^k(M)$ and every ψ_2 in $\Omega^{n-k}(M)$,

(23)
$$\forall t \ge 0, \ \langle (\varphi_f^{-t})^* \psi_1, \psi_2 \rangle = \sum_{i:\lambda_i^{(k)} = 0} \langle u_{i,1}^{(k)}, \psi_2 \rangle \langle v_{i,1}^{(k)}, \psi_1 \rangle + \mathcal{O}_{\psi_1, \psi_2}(e^{-\Lambda_+ t}),$$

for some $\Lambda_+ > 0$ which is independent of ψ_1 and ψ_2 . We now aim at describing more precisely the elements in the kernel. Note that this asymptotic expansion is valid when (f,g) is a smooth Morse pair inducing a Morse-Smale gradient flow.

Remark 6.4. To prove this property, we could probably use less regularity of the linearizing chart. In fact, our construction of the order function m only makes use of the fact that the linearizing chart is C^1 . Smoothness is only important if we want to determine exactly all the elements inside $\mathcal{R}_k(f,g)$.

6.3. Background material on currents. In order to describe the elements of the kernel, we start with some background material on the theory of currents which will also be useful in section 7. By a Theorem of Schwartz [66, Th. 37 p. 102] whose adaptation to the case of currents is straightforward, we first recall that:

Theorem 6.5 (Schwartz). Let u be a current of degree k supported by a smooth submanifold X embedded in M. Suppose that in a small neighbohood of $x \in X$, one has a system of coordinate functions $(x_i, y_j)_{1 \le i \le r, r+1 \le j \le n}$ where the coordinates $(x_i)_{1 \le i \le r}$ are transversal coordinates of X, i.e. the submanifold X is given by the equations $\{x_i = 0, 1 \le i \le r\}$. Then the current u reads locally as a finite sum:

(24)
$$u(x,y) = \sum_{\alpha,|I|+|J|=k} u_{\alpha,I,J}(y) \partial_x^{\alpha} \delta_{\{0\}}^{\mathbb{R}^r}(x) dx^I \wedge dy^J$$

where (α, I, J) are multi-indices, the $u_{\alpha,I,J}$ are distributions in $\mathcal{D}'(\mathbb{R}^{n-r})$.

If we denote by N^*X the conormal cycle of X, we also have the following property [21, Lemma 9.2]:

Lemma 6.6. Suppose that the assumptions of the previous Theorem hold and use the same notations. If $WF(u) \subset N^*(X)$, then the current u reads

(25)
$$u(x,y) = \sum_{\alpha,|I|+|J|=k} u_{\alpha,I,J}(y) \partial_x^{\alpha} \delta_{\{0\}}^{\mathbb{R}^r}(x) dx^I \wedge dy^J$$

where the $u_{\alpha,I,J}$ are smooth functions in $C^{\infty}(\mathbb{R}^{n-r})$.

6.4. Description of the eigenmodes. Let $0 \le k \le n$. In this paragraph, we will construct a "canonical" basis for the kernel of the operator $\mathcal{L}_{V_f}^{(k)}$ acting on the anisotropic space $\mathcal{H}_k^m(M)$. We proceed in three steps. First, we show that, near the "smallest" critical point a of index k, an element in the kernel must be proportional to the current of integration $[W^u(a)]$ on the unstable manifold. Then, we prove that the germ of current near to 0 can be extended into a current carried by $\overline{W^u(a)}$ that we denote by U_a . Finally, we show that these currents form indeed a basis of the kernel.

The construction of these currents U_a is known to be a delicate task and, to our knowledge, it was first made by Laudenbach under some tameness assumption on the Morse-Smale gradient flow [53, 54]. Here, we will show how our microlocal machinery allows to give an alternative proof of this construction for quite general Morse-Smale gradient flows.

6.4.1. Local form near the critical point. Let $u \neq 0$ be an element in $\mathcal{H}_k^m(M)$ such that $\mathcal{L}_{V_f}^{(k)}u = 0$. As above, we order the critical points $a_1 < \dots a_K$ according to the values of f. We denote by f the index such that, for every f is in a neighborhood of f and such that f does not vanish near f and f level from Lemma 5.3 that such a f exists and that f suppf is included in f in f for some small enough neighborhood f of f we have also shown in Proposition 6.3 that f must be of index f.

Using the above results, we deduce that, in the adapted coordinates of paragraph 3.5, the current u reads as a finite sum :

(26)
$$u(x,y) = \sum_{\alpha,|I|+|J|=k} u_{\alpha,I,J}(y) \partial_x^{\alpha} \delta_{\{0\}}^{\mathbb{R}^k}(x) dx^I \wedge dy^J$$

where the $u_{\alpha,I,J}$ are smooth functions in $C^{\infty}(\mathbb{R}^{n-k})$. The fact that $u_{\alpha,I,J}$ is smooth follows from Lemma 6.6 and from the fact that we can pick N_1 arbitrarly large in the definition of the order function m used to define our anisotropic Sobolev space. Let now ψ be a smooth test form of degree n-k supported near a_j . Using the fact that u satisfies $\mathcal{L}_{V_f}^{(k)}u=0$, one has, for every $0 < s \le 1$,

$$\langle u, \psi \rangle = \langle \varphi_f^{\ln s*} u, \psi \rangle.$$

Plugging the expression of the flow in these coordinates into the explicit form of u given by equation (26), one finds that u must be of the following form

$$u(x,y) = u_0(y)\delta_{\{0\}}^{\mathbb{R}^k}(x)dx_1 \wedge dx_2 \wedge \ldots \wedge dx_k,$$

where u_0 is a smooth function. Using one more time the invariance by the flow φ_f^t , one can deduce that u_0 is **constant** in a neighborhood of the origin. Using the result of Corollary D.4 in appendix D of [22], we recognize that $\delta_{\{0\}}^{\mathbb{R}^k}(x)dx_1 \wedge \ldots \wedge dx_k$ is the integral formula for the current of integration on the germ of submanifold $W^u(a_j) = \{x_i = 0, 1 \leq i \leq k\}$. Hence, our last statement means that locally near a_j , $u = c[W^u(a_j)]$ is a multiple of the current of integration on $W^u(a_j)$.

6.4.2. Extension of the local form to M. Our next step will be to prove that the current of integration on the germ of unstable submanifold near any critical point a of index r can be extended in a natural manner to M.

For that purpose, we still work with the adapted coordinates (x, y) introduced in paragraph 3.5. According to [22, Cor. D.4], the current of integration on $W^u(a)$ is well defined near a and it can be written locally as

$$[W^{u}(a)] = \delta_0^{\mathbb{R}^r}(x) dx_1 \wedge dx_2 \wedge \ldots \wedge dx_r.$$

Using the conventions of paragraph 3.5, we let $0 \le \theta \le 1$ be a smooth cutoff function which is equal to 1 on $(-\delta_a/2, \delta_a/2)^n$ and to 0 outside $(-\delta_a, \delta_a)^n$. Thanks to this cutoff function, we define the following element of $\mathcal{D}^{\prime,k}(M)$:

$$\tilde{U}_a := \theta(x, y) \delta_0^{\mathbb{R}^r}(x) dx_1 \wedge dx_2 \wedge \ldots \wedge dx_r.$$

By a straightforward calculation, \tilde{U}_a satisfies $\mathcal{L}_{V_f}\tilde{U}_a = 0$ on $(-\delta_a/4, \delta_a/4)^n$. One can also verify that, for every order function satisfying the properties of Lemma 4.1, \tilde{U}_a belongs to the space $\mathcal{H}_r^m(M)$. Recall that $(\varphi_f^{-t})^*$ generates a strongly continuous semigroup associated to the operator $-\mathcal{L}_{V_f}^{(r)}$ acting on $\mathcal{H}_r^m(M)$. As we have shown that there is no Jordan block in the kernel and that $-\mathcal{L}_{V_f}^{(r)}$ has a spectral gap, we can deduce that, in $\mathcal{H}_r^m(M)$ (hence in $\mathcal{D}'^{r}(M)$),

(27)
$$\lim_{t \to +\infty} \varphi_f^{-t*}(\tilde{U}_a) = \sum_{i:\lambda_i^{(r)} = 0} \langle \tilde{U}_a, v_{i,0}^{(r)} \rangle u_{i,0}^{(r)}.$$

We denote by U_a this limit current which satisfies $\mathcal{L}_{V_f}^{(r)}(U_a) = 0$. Let us now verify that U_a is an extension of $\tilde{U}_a|_{(-\delta_a/4,\delta_a/4)^n}$. For that purpose, we write the double inclusion

$$Y := \{(0, y) : \forall r + 1 < j < n, |y_i| \le \delta_a/4\} \subset X := \text{supp } (\tilde{U}_a) \subset W^u(a).$$

Set also $X_m = \varphi_f^{mt_0}(X)$, and $Y_m = \varphi_f^{mt_0}(Y)$ for some fixed $t_0 > 0$. Then, we have the following sequences of inclusions, for all $m \in \mathbb{N}$:

$$Y_m \subset X_m \subset W^u(a),$$

$$Y_m \subset Y_{m+1},$$

$$\operatorname{supp} \varphi_f^{-mt_0*}\left(\tilde{U}_a\right) = X_m \subset \overline{W^u(a)}.$$

These inclusions are simple consequences of the inclusion $(Y \subset X \subset W^u(a))$ and of the fact that $W^u(a)$ is stable by application of the one-parameter group $(\varphi_f^t)_t$. In particular, from (27), we have

(28)
$$\operatorname{supp}(U_a) \subset \overline{W^u(a)}.$$

In the local chart $(-\delta_a/4, \delta_a/4)^n$ around a, we can write

$$(\varphi_f^{-t_0})^*(\tilde{U}_a)(x, y, dx, dy) = \theta\left((e^{-t_0\chi_j(a)}x_j)_j, (e^{-t_0\chi_j(a)}y_j)_j\right) \delta_0^{\mathbb{R}^r}(x) dx_1 \wedge dx_2 \wedge \ldots \wedge dx_r.$$

As θ is constant and equal to 1 on $(-\delta_a/2, \delta_a/2)^n$, we have, provided that we pick $t_0 > 0$ small enough,

$$(\varphi_f^{-t_0})^*(\tilde{U}_a)(x,y,dx,dy) = \theta(x,y)\,\delta_0^{\mathbb{R}^r}(x)dx_1 \wedge dx_2 \wedge \ldots \wedge dx_r = \tilde{U}_a(x,y,dx,dy).$$

To summarize, we have

$$[W^{u}(a)]|_{(-\delta_a/4,\delta_a/4)^n} = \tilde{U}_a|_{(-\delta_a/4,\delta_a/4)^n} = \left((\varphi_f^{-t_0})^* (\tilde{U}_a) \right)|_{(-\delta_a/4,\delta_a/4)^n}.$$

From this, we can infer that, for every $m \geq 0$,

$$\left(\varphi_f^{-mt_0*}\tilde{U}_a\right)|_{\varphi_f^{mt_0}((-\delta_a/4,\delta_a/4)^n)} = \left(\varphi_f^{-(m+1)t_0*}\tilde{U}_a\right)|_{\varphi_f^{mt_0}((-\delta_a/4,\delta_a/4)^n)}.$$

Recall that the support of both currents is included in X_{m+1} . The above equality tells us that they coincide on the subset Y_m . From that, we can deduce that, for every $m \geq 0$,

$$\left(\varphi_f^{-mt_0*}\tilde{U}_a\right)|_{(-\delta_a/4,\delta_a/4)^n} = \left(\varphi_f^{-(m+1)t_0*}\tilde{U}_a\right)|_{(-\delta_a/4,\delta_a/4)^n}.$$

Hence, for every $m \geq 1$, one has

$$\left(\varphi_f^{-(m+1)t_0*}\tilde{U}_a\right)|_{(-\delta_a/4,\delta_a/4)^n} = \tilde{U}_a|_{(-\delta_a/4,\delta_a/4)^n},$$

which, by letting $m \to +\infty$ implies that

$$[W^u(a)]|_{(-\delta_a/4,\delta_a/4)^n} = U_a|_{(-\delta_a/4,\delta_a/4)^n}.$$

To summarize, we have proved the following result:

Proposition 6.7. Let (f,g) be a smooth Morse pair which induces a Morse-Smale gradient flow. Let a be a critical point of index r and let $0 \le \theta \le 1$ be a smooth cutoff function which is compactly supported in a small enough neighborhood V_a of a, and equal to 1 in an open neighborhood of a.

Then, there exists an open neighborhood $V_a \subset V_a$ of a such that the current

$$U_a := \sum_{i:\lambda_i^{(r)} = 0} \langle \theta[W^u(a)], v_{i,1}^{(r)} \rangle u_{i,1}^{(r)}$$

satisfies

- $\bullet \ U_a|_{\tilde{V}_a} = [W^u(a)]|_{\tilde{V}_a},$
- $\operatorname{supp}(U_a) = \overline{W^u(a)},$
- $\bullet \ \mathcal{L}_{V_{\varepsilon}}^{(r)}(U_a) = 0.$

We call these currents Laudenbach's currents of degree r.

6.4.3. The generation property. Even if the currents depend a priori on the choice of cutoff function, the following Proposition shows that they are in some sense "canonical":

Proposition 6.8. Let $0 \le k \le n$. The family of currents

$$\{U_a: a \in \operatorname{Crit}(f) \ and \ \operatorname{ind}(a) = k\}$$

forms a basis of the kernel of the operator

$$\mathcal{L}_{V_f}^{(k)}:\mathcal{H}_k^m(M)\to\mathcal{H}_k^m(M).$$

Proof. Let us first show that this family of currents is linearly independent. For that purpose, we suppose that

$$\sum_{a \in \operatorname{Crit}(f): \operatorname{ind}(a) = k} \alpha_a U_a = 0.$$

Let a be the "smallest" point of index k, in the sense that, for every other point $a' \neq a$ of index k, f(a') > f(a). We pick ψ a smooth form which is compactly supported near a and such that $\langle [W^u(a)], \psi \rangle \neq 0$. As the support of U_b is contained in $\overline{W^u(b)}$ for any critical point b of index k, we can deduce (provided that the support of ψ is small enough) that

$$0 = \sum_{a \in \operatorname{crit}(f): \operatorname{ind}(b) = k} \alpha_b \langle U_b, \psi \rangle = \alpha_a \langle U_a, \psi \rangle = \alpha_a \langle [W^u(a)], \psi \rangle.$$

From this, we deduce that $\alpha_a = 0$. By induction, we can conclude that the family contains only linearly independent currents.

It remains to verify that this family generates all the kernel. Let $u \neq 0$ be an element in $\mathcal{H}_k^m(M)$ in the kernel of $\mathcal{L}_{V_f}^{(k)}$. From the argument of paragraph 6.4.1, we know that u must be equal to $c_a[W^u(a)]$ in a neighborhood of a where a is the "smallest" critical point where u does not identically vanish and where c_a is a fixed constant. Set now

$$u_1 = u - c_a U_a.$$

We know that u_1 belongs to $\mathcal{H}_k^m(M)$ and that it satisfies $\mathcal{L}_{V_f}^{(k)}u_1 = 0$. Morever, by construction, we know that u_1 vanishes identically near every critical point a' with $a' \leq a$. Repeating the process a finite number of times, we finally get that

$$u = \sum_{a \in \operatorname{Crit}(f): \operatorname{ind}(a) = k} c_a U_a,$$

for some c_a in \mathbb{R} .

6.4.4. Back to the correlations. We now turn back to the correlation function. For that purpose, we fix $\Lambda > 0$ which is strictly smaller than $\min\{|\chi_j(a)| : 1 \le j \le n, a \in \operatorname{Crit}(f)\}$. As in paragraph 5.3, we write the spectral decomposition associated with the spectral projector of the eigenvalue $\lambda = 0$. We have shown that there is no Jordan blocks for this eigenvalue (Prop. 6.1), and that we can choose a basis of eigenmodes $(U_a)_a$ indexed by the critical points of index k. Moreover, all the elements in this basis can be chosen in such a

way that the support of U_a is equal to $\overline{W^u(a)}$. We denote by S_a the corresponding dual basis. Proceeding as in paragraph 5.3, we have then

$$\forall \psi_1 \in \mathcal{H}_k^m(M), \ \forall \psi_2 \in \mathcal{H}_{n-k}^{-m}(M), \ \langle \varphi_f^{-t*} \psi_1, \psi_2 \rangle = \sum_{a: \text{ind}(a) = k} \langle U_a, \psi_2 \rangle \langle S_a, \psi_1 \rangle + \mathcal{O}_{\psi_1, \psi_2}(e^{-\Lambda t}),$$

for every t>0. Applying the arguments of the previous paragraphs to the operator $\mathcal{L}_{V_{-f}}^{(n-k)}$ acting on the anisotropic space $\mathcal{H}_{n-k}^{-m}(M)$, we can construct a basis of the kernel that we denote by $(\overline{S}_a)_a$ indexed by the critical points of index k. Mimicking the above procedure, we can impose that \overline{S}_a has support contained in $\overline{W^s(a)}$ and that \overline{S}_a coincides with $[W^s(a)]$ in a neighborhood of the critical point a. In particular, as $\overline{W^s(a)} \cap \overline{W^u(a)} = \{a\}$, we can use our local adapted coordinates near a to find that $\langle \overline{S}_a, U_a \rangle = 1$. Consider now $a' \neq a$ of index k. If we are able to show that $\langle \overline{S}_a, U_{a'} \rangle = 0$ for every such a', then we will have that $\overline{S}_a = S_a$. To prove this, we just need to observe that $\overline{W^s(a)} \cap \overline{W^u(a')} = \emptyset$. In fact, according to Remark 3.7 applied to f and -f, we find that, if x belongs to $\overline{W^s(a)} \cap \overline{W^u(a')}$, then ind $(x_-) \geq k$ and ind $(x_+) \leq k$, where $x \in W^u(x_-) \cap W^s(x_+)$. In other words, from the Morse-Smale assumption, $x_- = x_+$. From Lemma 6.2, we would then have a = a' which gives the contradiction.

To summarize, we have shown:

Proposition 6.9. Let (f, g) be a smooth Morse pair which induces a Morse-Smale gradient flow. Let $\Lambda > 0$ be such that

$$\Lambda < \min\{|\chi_j(a)| : 1 \le j \le n, a \in \operatorname{Crit}(f)\}.$$

Then, for every $0 \le k \le n$, and, for every $a \in Crit(f)$ of index k, there exists

$$U_a \in \mathcal{H}_k^m(M)$$
 and $S_a \in \mathcal{H}_{n-k}^{-m}(M)$

whose support are respectively equal to $\overline{W^u(a)}$ and $\overline{W^s(a)}$ and satisfying

$$\forall t \ge 0, \ \varphi_f^{-t*} = \sum_{a: \text{ind}(a)=k} |U_a\rangle\langle S_a| + \mathcal{O}_{\mathcal{H}_k^m \to \mathcal{H}_k^m}(e^{-\Lambda t}).$$

This implies in particular Theorem 2.3 from the introduction. We also note that we recover Theorem 3.3 from [46] via our spectral analysis of gradient flows. Our result is slightly stronger in two ways: (1) it holds for quite general families of Morse-Smale gradient flows (not necessarily associated with a locally flat metric); (2) it gives a rate of convergence in the asymptotic. This kind of decomposition of the transfer operator into a product of stable and unstable distributions is also at the heart of the spectral analysis of Anosov vector fields performed by Faure and Tsujii in [34, 33]. As we shall see in section 7, we can in fact generalize this expansion at any order.

6.4.5. Solving the cohomohological equation $\mathcal{L}_{V_f}^{(k)}u = \psi$. Note that we can use our spectral decomposition to prove Corollary 2.5. In fact, given any $\psi \in \Omega^k(M)$, one can write

$$\psi = \mathbb{P}^{(k)}(\psi) + (\mathrm{Id} - \mathbb{P}^{(k)})(\psi),$$

for $\mathbb{P}^{(k)}$ associated with the eigenvalue 0 – see e.g. paragraph 8.4.2 below. As we can solve $\mathcal{L}_{V_f}^{(k)}(u) = (\mathrm{Id} - \mathbb{P}^{(k)})(\psi)$ from the spectral properties of $\mathcal{L}_{V_f}^{(k)}$, the problem

$$\mathcal{L}_{V_f}^{(k)}(u) = \psi$$

admits a solution in $\mathcal{H}_k^m(M)$ if and only if

$$\mathcal{L}_{V_f}^{(k)}(u) = \mathbb{P}^{(k)}(\psi) = \sum_{a \in \text{Crit}(f): \text{ind}(a) = k} \langle S_a, \psi \rangle U_a$$

has a solution in $\mathcal{H}_k^m(M)$. As the U_a forms a basis of the kernel of $\mathcal{L}_{V_f}^{(k)}$ which does not contain any Jordan block, this is equivalent to the fact that, for every a of index k, $\langle S_a, \psi \rangle = 0$.

7. The case
$$\lambda \neq 0$$

In this section, we describe some properties of the eigenmodes when $\lambda \neq 0$. Except for the use of the nonresonance assumption, the proof is the same as in the case $\lambda = 0$ but the arguments are somewhat more involved due to the possible multiplicity at every critical point. Precisely, we will one more time show that there exists a "canonical" basis of eigenmodes which are carried by the closure of the unstable manifolds $W^u(a)$ (Propositions 7.5 and 7.7) and that there are no Jordan blocks (Proposition 7.2). In all this section, we make the assumption that the Lyapunov exponents

$$\{\chi_j(a): 1 \le j \le n, \ a \in \operatorname{Crit}(f)\}$$

are rationally independent. From paragraph 3.5, we know that this implies in particular that (f,g) is a smooth Morse pair. Hence we can apply the results from section 5. We will crucially use this nonresonance assumption to ensure the absence of Jordan blocks. From this, we will be able to construct a more or less explicit basis of eigenmodes.

Remark 7.1. This (global) nonresonance assumption may sound a little bit surprising but it seems to us that removing this assumption may be a subtle issue. In fact, our proof works under the following assumptions:

- (f,g) is a smooth Morse pair generating a Morse-Smale gradient flow,
- If there exists $\alpha \neq 0$ and $\alpha' \neq 0$ in \mathbb{N}^n such that

$$\sum_{j=1}^{n} \alpha_{j} |\chi_{j}(a)| = \sum_{j=1}^{n} \alpha'_{j} |\chi_{j}(a')|,$$

then a = a'.

7.1. **Jordan blocks.** As for the kernel, we start by proving the absence of Jordan blocks. In other words, the algebraic multiplicity is always equal to the geometric multiplicity.

Proposition 7.2. Let φ_f^t be a Morse-Smale gradient flow all of whose Lyapunov exponents are rationally independent. Let $0 \le k \le n$. Then, when acting on a convenient anisotropic space $\mathcal{H}_k^m(M)$, one has, for every $\lambda \in \mathcal{R}_k(f,g)$,

$$\operatorname{Ker}((\mathcal{L}_{V_f}^{(k)} + \lambda)) = \operatorname{Ker}((\mathcal{L}_{V_f}^{(k)} + \lambda)^2).$$

Proof. Once again, the proof is very similar to the case of the kernel. We only treat the case $\lambda \neq 0$. Suppose by contradiction that there exists a Jordan block associated to the eigenvalue $\lambda > 0$ for a certain degree k. Once again, it means that there exists $u_0 \neq 0$ and $u_1 \neq 0$ in our anisotropic Sobolev space of currents $\mathcal{H}_k^m(M)$ such that

$$\mathcal{L}_{V_f}^{(k)} u_0 = \lambda u_0 \text{ and } \mathcal{L}_{V_f}^{(k)} u_1 = \lambda u_1 + u_0.$$

Integrating these expressions, we find that, for all t in \mathbb{R}_{-} ,

$$(\varphi_f^t)^* u_0 = e^{\lambda t} u_0 \text{ and } (\varphi_f^t)^* u_1 = e^{\lambda t} (u_1 + t u_0).$$

As in our computation of the spectrum, we let $t = \ln s$ with $0 < s \le 1$,

$$(29) \qquad (\varphi_f^{\ln s})^* u_1 - s^{\lambda} u_1 = s^{\lambda} (\ln s) u_0.$$

Following the proof of paragraph 5.2.2, we denote by j+1 the index point such that, for every $i \leq j$, u_0 vanishes in a neighborhood of a_i and such that u_0 does not vanish near a_{j+1} . This implies that $\sup(u_0) \cap V_{a_{j+1}}$ is included in $W^u(a_{j+1})$ and that λ is of the following form:

$$\lambda = \sum_{i=1}^{n} \alpha_i |\chi_i(a_{j+1})|.$$

As $\mathcal{L}_{V_f}^{(k)}u_1 = \lambda u_1 + u_0$, we know that $\mathcal{L}_{V_f}^{(k)}u_1 = \lambda u_1$ on the open set $f^{-1}(\langle f(a_{j+1})\rangle)$. Suppose now that there exists $i_0 \leq j$ such that u_1 does not identically vanish near a_{i_0} . Without loss of generality, we may suppose that i_0 is minimal. In such a neighborhood, one has $\mathcal{L}_{V_f}^{(k)}u_1 = \lambda u_1$ as u_0 vanishes near a_{i_0} . Arguing as in paragraph 5.2.2 one more time, we would find that

$$\lambda = \sum_{i=1}^{n} \alpha_i |\chi_i(a_{i_0})|.$$

As we supposed that $\lambda \neq 0$ and that the Lyapunov exponents are rationally independent, this would lead to a contradiction. Hence, u_1 vanishes near any critical point a_i with $i \leq j$. As u_0 is locally supported on $W^u(a_{j+1})$, we know that u_1 still satisfies the eigenvalue equation $\mathcal{L}_{V_f}^{(k)}u_1 = \lambda u_1$ near a_{j+1} and outside $W^u(a_{j+1})$. By propagation, we deduce that u_1 is locally supported on $W^u(a_{j+1})$. According to Remark 5.5, we are then able to infer that $\langle (\varphi_f^{\ln s})^* u_1, \psi \rangle$ has a (bounded) polyhomogeneous expansion in s as $s \to 0^+$ for every smooth test form ψ supported near a_{j+1} . From our assumption on j, one can find ψ such that $\langle u_0, \psi \rangle \neq 0$ which gives the expected contradiction when we write (29).

We conclude this paragraph by mentioning the following analogue of Proposition 6.3 which we used above and which was implicitely proved in paragraph 5.2.2:

Proposition 7.3. Suppose that (f,g) is a smooth Morse pair generating a Morse-Smale gradient flow. Let $u \neq 0$ be an element of $\mathcal{H}_k^m(M)$ satisfying $\mathcal{L}_{V_f}^{(k)}u = \lambda u$. Let a be the critical point of f satisfying the following properties:

- u does not vanish in any neighborhood of a,
- for every a' in Crit(f) satisfying f(a') < f(a), u identically vanishes near a'.

Then, $-\lambda$ belongs to the index set $\mathcal{I}_k(a)$.

Note that this part of the proof is based on the proof of paragraph 5.2.2 and it does not require the rational independence of the Lyapunov exponents.

7.2. **Description of the eigenmodes.** We now use similar arguments as at was done in section 6 in order to construct a basis of every eigenspace. Compared with the case $\lambda = 0$, there will be some combinatorial complications due to the possible multiplicity of the eigenvalue at a critical point a. In this paragraph, we fix $0 \le k \le n$ and λ an eigenvalue of the operator

$$\mathcal{L}_{V_f}^{(k)}:\mathcal{H}_k^m(M)\to\mathcal{H}_k^m(M).$$

Recall from Proposition 5.1 that λ must be of the form

$$\lambda = \sum_{j \in I \cup J} (\alpha_j + 1) |\chi_j(a)| + \sum_{j \in (I \cup J)^c} \alpha_j |\chi_j(a)|,$$

where

- a is a critical point of index r,
- for every $1 \leq j \leq n$, α_j is a nonnegative integer,
- $I \subset \{1, \ldots, r\}$ and $J \subset \{r+1, \ldots, n\}$ such that |J| |I| = k r.

We shall now proceed as in the case of the kernel. First, we determine the local form of an eigenmode near critical points. Then, we show how to extend these local models into currents defined on M. Finally, we show that these currents form a basis of the eigenspace associated to λ .

Remark 7.4. The case $\lambda = 0$ was already treated in the previous paragraph. Thus, we will always suppose $\lambda \neq 0$ in this paragraph.

7.2.1. Local form near the "smallest" critical point. Let $u \neq 0$ be an element in $\mathcal{H}_k^m(M)$ such that $\mathcal{L}_{V_f}^{(k)}u = \lambda u$. As before, we denote by j the index such that, for every i < j, u vanishes in a neighborhood of a_i and such that u does not vanish near a_j . Recall that $\sup(u) \cap V_{a_j}$ is included in $W^u(a_j)$ for some small enough neighborhood V_{a_j} of a_j . Thanks to Proposition 7.3, $-\lambda$ belongs to $\mathcal{I}_k(a_j)$. In order to alleviate notations, we will write $a_j = a$ in the following.

Using Schwartz's Theorem and Lemma 6.6 one more time, we deduce that, in the adapted coordinates of paragraph 3.5, the current u reads as a finite sum :

(30)
$$u(x, y, dx, dy) = \sum_{\alpha', |I'| + |J'| = k} u_{\alpha', I', J'}(y) \partial_x^{\alpha'} \delta_{\{0\}}^{\mathbb{R}^r}(x) dx^{I'} \wedge dy^{J'}$$

where the $u_{\alpha',I',J'}$ are smooth functions in $C^{\infty}(\mathbb{R}^{n-r})$. A direct calculation shows us that, in a small enough neighborhood of a, one has, for every $0 < s \le 1$,

$$(\varphi_f^{\ln s*}u)(x,y,dx,dy) = \sum_{\alpha',|I'|+|J'|=k} u_{\alpha',I',J'}((s^{\chi_j(a)}y_j)_j)\partial_x^{\alpha'}\delta_{\{0\}}^{\mathbb{R}^r}(x)s^{\tilde{\lambda}_{I',J',\alpha'}}dx^{I'}\wedge dy^{J'},$$

where

$$\tilde{\lambda}_{I',J',\alpha'} := \sum_{j=1}^{r} (\alpha'_j + 1)|\chi_j(a)| - \sum_{j \in I'} |\chi_j(a)| + \sum_{j \in J'} |\chi_j(a)|.$$

On the other hand, as u satisfies $\mathcal{L}_{V_f}^{(k)}u = \lambda u$, we know that, for every smooth test form ψ of degree n-k and for every $0 < s \le 1$,

$$\langle \varphi_f^{\ln s*} u, \psi \rangle = s^{\lambda} \langle u, \psi \rangle.$$

Combining this equality to the local form of u, we find

$$s^{\lambda}\langle u,\psi\rangle = \sum_{\alpha',|I'|+|J'|=k} s^{\tilde{\lambda}_{I',J',\alpha'}} \left\langle \partial_x^{\alpha'} \delta_{\{0\}}^{\mathbb{R}^r}(x), u_{\alpha',I',J'}((s^{\chi_j(a)}y_j)_j) dx^{I'} \wedge dy^{J'} \wedge \psi(x,y,dx,dy) \right\rangle.$$

Write now the Taylor expansion of $u_{\alpha',I',J'}$ (which is \mathcal{C}^{∞}). From that, we find that

$$u_{\alpha',I',J'}(y) = c_{\alpha',I',J'} y_{r+1}^{\alpha'_{r+1}} \dots y_n^{\alpha'_n},$$

where $c_{\alpha',I',J'}$ is some fixed constant, α'_i belongs to \mathbb{N} for every $r+1 \leq j \leq n$ and

$$\tilde{\lambda}_{I',J',\alpha'} + \sum_{j=r+1}^{n} \alpha'_{j} |\chi_{j}(a)| = \lambda.$$

Equivalently, one has

$$\lambda = \sum_{j=1}^{r} (\alpha'_j + 1)|\chi_j(a)| + \sum_{j=r+1}^{n} \alpha'_j|\chi_j(a)| - \sum_{j \in I'} |\chi_j(a)| + \sum_{j \in J'} |\chi_j(a)|.$$

To summarize, this shows that the current u reads in the adapted coordinates near a:

(31)
$$u(x, y, dx, dy) = \sum_{\alpha, I, J: (*)} c_{\alpha, I, J} (y \partial_x)^{\alpha} \delta_{\{0\}}^{\mathbb{R}^r} (x) \left(\wedge_{j \notin I} dx_j \right) \wedge \left(\wedge_{j \in J} dy_j \right),$$

where $c_{\alpha,I,J}$ are some fixed constant and where (*) means that (α,I,J) satisfies

- for every $1 \leq j \leq n$, $\alpha_i \in \mathbb{N}$,
- $I \subset \{1, \dots, r\}, J \subset \{r+1, \dots, n\},\$
- |J| |I| = k r, $\lambda = \sum_{j \in I \cup J} (\alpha_j + 1) |\chi_j(a)| + \sum_{j \in (I \cup J)^c} \alpha_j |\chi_j(a)|$,

7.2.2. Extension of the local form to M. Mimicking what was done in the case of the kernel, we will now explain how the local form obtained in (31) can be extended into a natural eigencurrent carried by the closure of $W^u(a)$.

For a fixed triple (α, I, J) satisfying the conditions (*), we define

(32)
$$\tilde{U}_{a}^{\alpha,I,J}(x,y,dx,dy) := \theta(x,y) (y\partial_{x})^{\alpha} \delta_{\{0\}}^{\mathbb{R}^{r}}(x) (\wedge_{j\notin I} dx_{j}) \wedge (\wedge_{j\in J} dy_{j}),$$

where θ is the same smooth function as in paragraph 6.4.2. By construction, one can verify that

$$\mathcal{L}_{V_f}^{(k)} \tilde{U}_a^{\alpha,I,J} = \lambda \tilde{U}_a^{\alpha,I,J}$$

on the open neighborhood $(-\delta_a/4, \delta_a/4)^n$. Moreover, this current belong to the anisotropic space $\mathcal{H}_k^m(M)$ provided that we pick N_0 large enough (compared with $|\alpha|$) in the definition of the order function m. Using the conventions of (17), we then set

$$U_a^{\alpha,I,J} = \sum_{i:\lambda_i^{(k)} = \lambda} \langle \tilde{U}_a^{\alpha,I,J}, v_{i,1}^{(k)} \rangle u_{i,1}^{(k)},$$

which obviously satisfies the eigenvalue equation:

$$\mathcal{L}_{V_f}^{(k)} U_a^{\alpha,I,J} = \lambda U_a^{\alpha,I,J}$$

Let us now describe some properties of this current. First, we let ψ be a smooth n-k form carried outside $\overline{W^u(a)}$. For such a form and for every $0 < s \le 1$, one has $\langle \varphi_f^{-\ln s*} \tilde{U}_a^{\alpha,I,J}, \psi \rangle = 0$. Hence, every term in the asymptotic expansion 18 must vanish. In particular, one has $\langle U_a^{\alpha,I,J}, \psi \rangle = 0$ for every smooth test form supported outside $\overline{W^u(a)}$. Equivalently, one has

$$\operatorname{supp}\left(U_a^{\alpha,I,J}\right) \subset \overline{W^u(a)}.$$

By invariance under the gradient flow, the support is in fact equal to $\overline{W^u(a)}$. Like in the case of the kernel, we would like to verify that $\tilde{U}_a^{\alpha,I,J}$ and $U_a^{\alpha,I,J}$ coincide in a neighborhood of the critical point a. For that purpose, we let $\psi(x,y,dx,dy)$ be a some smooth test form carried in the neighborhood with adapted coordinates. Mimicking the calculation of paragraph 5.4, one finds that, for every $0 < s \le 1$,

$$\left\langle \varphi_f^{\ln s*} \tilde{U}_a^{\alpha,I,J}, \psi \right\rangle = s^{\lambda} (1 + o(1)) \left\langle \left(y \partial_x \right)^{\alpha} \delta_{\{0\}}^{\mathbb{R}^r} (x) \left(\wedge_{j \notin I} dx_j \right) \wedge \left(\wedge_{j \in J} dy_j \right), \psi \right\rangle.$$

Using one more time the spectral expansion of the correlation function (18) and using the fact that there is no Jordan blocks, one can identify the term of order s^{λ} in the asymptotic. In particular, this implies that

$$\langle U_a^{\alpha,I,J}, \psi \rangle = \sum_{i:\lambda_i^{(k)} = \lambda} \langle \tilde{U}_a^{\alpha,I,J}, v_{i,1}^{(k)} \rangle \langle u_{i,1}^{(k)}, \psi \rangle = \langle (y\partial_x)^{\alpha} \delta_{\{0\}}^{\mathbb{R}^r}(x) \left(\wedge_{j \notin I} dx_j \right) \wedge \left(\wedge_{j \in J} dy_j \right), \psi \rangle,$$

for every smooth test form ψ compactly supported in a small enough neighborhood of a. To summarize, we have shown the following:

Proposition 7.5. Let φ_f^t be a Morse-Smale gradient flow all of whose Lyapunov exponents are rationally independent. Let a be a critical point of index r, let $0 \le k \le n$ and let $0 \le \theta \le 1$ be a smooth cutoff function which is compactly supported in a small enough

neighborhood V_a of a, and equal to 1 in an open neighborhood of a. Let I be a subset of $\{1,\ldots,r\}$ and J be a subset of $\{r+1,\ldots,n\}$ satisfying |J|-|I|=k-r. Let α be an element in \mathbb{N}^n . Set $[W^u(a)]_{\alpha,I,J}$ to be the image in the adapted coordinate chart of

$$(y\partial_x)^{\alpha} \delta_{\{0\}}^{\mathbb{R}^r}(x) \left(\wedge_{j \notin I} dx_j \right) \wedge \left(\wedge_{j \in J} dy_j \right).$$

Then, there exists an open neighborhood $\tilde{V}_a \subset V_a$ of a such that the current

$$U_a^{\alpha,I,J} := \sum_{i:\lambda^{(r)} = \lambda} \langle \theta[W^u(a)]_{\alpha,I,J}, v_{i,1}^{(r)} \rangle u_{i,1}^{(r)}$$

satisfies

- $$\begin{split} \bullet \ U_a^{\alpha,I,J}|_{\tilde{V}_a} &= [W^u(a)]_{\alpha,I,J}|_{\tilde{V}_a}, \\ \bullet \ \operatorname{supp}(U_a^{\alpha,I,J}) &= \overline{W^u(a)}, \\ \bullet \ \mathcal{L}_{V_f}^{(k)}(U_a^{\alpha,I,J}) &= \lambda U_a^{\alpha,I,J} \ with \end{split}$$

$$\lambda = \sum_{j \in I \cup J} (\alpha_j + 1) |\chi_j(a)| + \sum_{j \in (I \cup J)^c} \alpha_j |\chi_j(a)|.$$

This Proposition gives a family of natural currents generalizing Laudenbach's currents which were only defined for the eigenvalue 0. Note that they are well defined as soon as (f,g) is a smooth Morse pair generating a Morse-Smale gradient flow. Up to the linearization chart, their expression is more or less explicit. For every λ in $\mathcal{I}_k(a)$, we define the "multiplicity" of λ as

(33)
$$m_k(\lambda) := |\{(\alpha, I, J) \text{ satisfying } (*)\}|,$$

where (*) means that (α, I, J) satisfies

- for every $1 \leq j \leq n$, $\alpha_i \in \mathbb{N}$,
- $I \subset \{1, \dots, r\}, J \subset \{r+1, \dots, n\},\$
- |J| |I| = k r, $\lambda = \sum_{j \in I \cup J} (\alpha_j + 1) |\chi_j(a)| + \sum_{j \in (I \cup J)^c} \alpha_j |\chi_j(a)|$.

Remark 7.6. In order to compute the Weyl's law for our operators, it will be convenient to rewrite things in a slightly different manner. More precisely, for any given α in \mathbb{N}^n , we set

$$m_{k,a}(\alpha) := |\{(I \times J \subset \{1, \dots, r\} \times \{r+1, \dots, n\} : |J| - |I| = k - r, \text{ and } \forall j \in I \cup J, \ \alpha_j \ge 1\}|,$$

where r is the index of a. With these conventions and thanks to the rational independence, any $\alpha.|\chi(a)|$ appears with multiplicity $m_{k,a}(\alpha)$ in $\mathcal{R}_k(f,g)$.

7.2.3. The generation Theorem. We conclude this section by showing that the currents we have just constructed generate a basis of $\operatorname{Ker}(\mathcal{L}_{V_f}^{(k)} + \lambda)$, i.e.

Proposition 7.7. Let φ_f^t be a Morse-Smale gradient flow all of whose Lyapunov exponents are rationally independent. Let $0 \le k \le n$ and let $\lambda \ne 0$ be an element in $\mathcal{R}_k(f,g)$. The family of currents

$$\left\{ U_a^{\alpha,I,J} : \sum_{j \in I \cup J} (\alpha_j + 1) |\chi_j(a)| + \sum_{j \in (I \cup J)^c} \alpha_j |\chi_j(a)| = -\lambda \right\}$$

forms a basis of the kernel of the operator

$$\mathcal{L}_{V_f}^{(k)} + \lambda : \mathcal{H}_k^m(M) \to \mathcal{H}_k^m(M).$$

In particular, the kernel of this operator is of dimension $m_k(\lambda)$.

Proof. The proof is almost the same as in the case $\lambda = 0$, and we briefly adapt it in this context.

Let us first show that it generates the kernel of $\mathcal{L}_{V_f}^{(k)} + \lambda$. This follows from the discussion from paragraph 7.2.1. If we take u in $\mathcal{H}_k^m(M)$ satisfying $\mathcal{L}_{V_f}^{(k)}u = -\lambda u$, then (31) gives us a family of constants $c_{\alpha,I,J}$. We then set $\tilde{u} = u - \sum_{\alpha,I,J} c_{\alpha,I,J} U_a^{\alpha,I,J}$ where a is the smallest critical point where u does vanish in a neighborhood. Note that λ belongs to $\mathcal{I}_k(a)$ from Proposition 7.3. One still has that $\mathcal{L}_{V_f}^{(k)}\tilde{u} = -\lambda \tilde{u}$. From paragraph 7.2.1, we also know that \tilde{u} vanishes near any critical point b satisfying $f(b) \leq f(a)$. Then, combing the rational independence of the Lyapunov exponents with Proposition 7.3, we conclude that $\tilde{u}=0$.

Let us now briefly verify that these elements are independent. Suppose that one can write

$$\sum_{(\alpha,I,J) \text{ satisfying } (*)} \gamma_{\alpha,I,J} U_a^{\alpha,I,J} = 0$$

We write this relation near the critical point a and we use that the germs of current are (from proposition 7.5) linearly independent near this critical point. This implies that $\gamma_{\alpha,I,J} = 0$ for every (α, I, J) associated with a.

7.3. Back to correlations. Mimicking the arguments of paragraph 6.4.4, we obtain a full asymptotic for the correlation function:

Proposition 7.8. Let φ_f^t be a Morse-Smale gradient flow all of whose Lyapunov exponents are rationally independent. Let $0 \le k \le n$.

Then, for every a in Crit(f) and for every α in \mathbb{N}^n , there exist⁷

- $(U_{a,k}^{\alpha,j})_{j=1,\dots,m_{k,a}(\alpha)}$ in $\mathcal{D}'^{,k}(M)$ whose supports is equal to $\overline{W^{u}(a)}$, $(S_{a,n-k}^{\alpha,j})_{j=1,\dots,m_{k,a}(\alpha)}$ in $\mathcal{D}'^{,n-k}(M)$ whose supports is equal to $\overline{W^{s}(a)}$,

such that, for every $\Lambda > 0$ and for every $t \geq 0$,

$$\varphi_f^{-t*} = \sum_{a \in \operatorname{Crit}(f)} \sum_{\alpha \in \mathbb{N}^n: \alpha. |\chi(a)| < \Lambda} e^{-\alpha. |\chi(a)| t} \sum_{j=1}^{m_{k,a}(\alpha)} |U_{a,k}^{\alpha,j}\rangle \langle S_{a,n-k}^{\alpha,j}| + \mathcal{O}_{\Omega^k \to \mathcal{D}', k}(e^{-\Lambda t}).$$

In particular, this implies Theorem 2.2 from the introduction.

⁷See Remark 7.6 for the precise definition of $m_{k,a}(\alpha)$.

7.3.1. The duality map $U_{a,k}^{\alpha,j} \mapsto S_{a,n-k}^{\alpha,j}$ as a generalized Fourier transform. The duality map $U_{a,k}^{\alpha,j} \mapsto S_{a,n-k}^{\alpha,j}$ has in fact a nice interpretation in terms of Fourier transform in the linearizing chart near the critical point a. Recall that the Fourier transform can be defined on the space of tempered currents on \mathbb{R}^n [66, p. 396]. This notion of Fourier transform coincides with the Fourier (sometimes called super Fourier) transform appearing in the theory of Berezin integrals [43, Ch. 7] – see section 5 of [22] for a brief reminder on Berezin integrals. In fact, a tempered distribution u(z, dz) can be identified with an element in the Grassmann algebra $\mathcal{S}'(\mathbb{R}^n)[dz_1,\ldots,dz_n]$. Using the conventions of [22], one can define the Fourier transform as follows:

$$\forall u \in \mathcal{S}'(\mathbb{R}^n)[dz_1, \dots, dz_n], \ \mathcal{F}(u)(z, dz) := \frac{1}{(2\pi)^{\frac{n}{2}}} \int_{\mathbb{R}^{(n|n)}} u(p, P) e^{-i(p.z + P.dz)} dp dP.$$

Thanks to Proposition 7.5, we know exactly the local form of the current $U_{a,k}^{\alpha,j}$ near the critical point a, and we can check that the map $U_{a,k}^{\alpha,j} \mapsto S_{a,n-k}^{\alpha,j}$ is locally near a equal to \mathcal{F} up to some normalizing constant depending only on n and k.

- 7.4. **Asymptotic formulas.** In order to conclude this section, we will give some nice asymptotic formulas that can be easily derived from our description of the spectrum.
- 7.4.1. Weyl asymptotics. Due to the fact that we obtained an explicit expression for the spectrum of the transfer operator, we can easily obtain some Weyl's formula. More precisely,

Proposition 7.9 (Weyl Law). Let $0 \le k \le n$ and let φ_f^t be a Morse-Smale gradient flow all of whose Lyapunov exponents are rationally independent. Then, one has

$$|\{\lambda \in \mathcal{R}_k(f,g) : |\lambda| \le \Lambda\}| = \frac{\Lambda^n}{k!(n-k)!} \sum_{a \in \operatorname{Crit}(f)} \frac{1}{\prod_{j=1}^n |\chi_j(a)|} + \mathcal{O}(\Lambda^{n-1}), \text{ as } \Lambda \to +\infty,$$

where the elements in $\mathcal{R}_k(f,g)$ are counted with their algebraic multiplicities.

Proof. From Remark 7.6 and Propositions 7.7 and 7.2, one knows that

$$|\{\lambda \in \mathcal{R}_k(f,g) : |\lambda| \le \Lambda\}| = \sum_{a \in \text{Crit}(f)} \sum_{\alpha \in \mathbb{N}^n : \alpha, |\gamma(a)| \le \Lambda} m_{k,a}(\alpha).$$

Hence, we can fix a critical point a and compute $\sum_{\alpha \in \mathbb{N}^n: \alpha . |\chi(a)| \leq \Lambda} m_{k,a}(\alpha)$. We write

$$\sum_{\alpha \in \mathbb{N}^n : \alpha. |\chi(a)| \leq \Lambda} m_{k,a}(\alpha) = \left| \left\{ \alpha \in \mathbb{N}^n ; I \subset \{1, \dots, r\}, J \subset \{r+1, \dots, n\} : \alpha. |\chi(a)| \leq \Lambda \text{ and } (**) \right\} \right|,$$

where r is the index of a and where (**) means that |J|-|I|=k-r and $\forall j \in I \cup J$, $\alpha_j \geq 1$. We start by fixing a pair (I,J) where $I \subset \{1,\ldots,r\}$ and $J \subset \{r+1,\ldots,n\}$ subject to the condition |J|-|I|=k-r. We then want to compute

$$|\{\alpha \in \mathbb{N}^n : \forall j \in I \cup J, \ \alpha_j \ge 1 \text{ and } \alpha. |\chi(a)| \le \Lambda\}|.$$

One can verify that

$$|\{\alpha \in \mathbb{N}^n : \forall j \in I \cup J, \ \alpha_j \ge 1 \text{ and } \alpha.|\chi(a)| \le \Lambda\}| = |\{\alpha \in \mathbb{N}^n : \alpha.|\chi(a)| \le \Lambda\}| + \mathcal{O}(\Lambda^{n-1}).$$

Then, one has

$$|\{\alpha \in \mathbb{N}^n : \alpha.|\chi(a)| \le \Lambda\}| = \operatorname{Vol}(\{x \in (\mathbb{R}_+)^n : |\chi(a)| x \le \Lambda\}) + \mathcal{O}(\Lambda^{n-1}),$$

which is the volume of a simplical domain. Hence, one has

$$|\{\alpha \in \mathbb{N}^n : \forall j \in I \cup J, \ \alpha_j \ge 1 \text{ and } \alpha. |\chi(a)| \le \Lambda\}| = \frac{\Lambda^n}{n! |\prod_{j=1}^n \chi_j(a)|} + \mathcal{O}(\Lambda^{n-1}).$$

This is valid for any $I \subset \{1, \ldots, r\}, J \subset \{r+1, \ldots, n\}$ subject to the condition |J| - |I| = k - r. One can remark that the number of such $I \times J$ is equal to the number of $I' \times J \subset \{1, \ldots, r\} \times \{r+1, \ldots, n\}$ subject to the condition |J| + |I'| = k. This is exactly equal to $\binom{n}{k}$. This concludes the proof of the Proposition.

7.4.2. Trace formulas. In this paragraph, we discuss briefly some trace formulas related to our problem. For every $0 \le k \le n$ and every $\lambda \ge 0$, we set

$$C^k(f,\lambda) := \operatorname{Ker}(\mathcal{L}_{V_f}^{(k)} - \lambda),$$

where we mean the kernel of the operator in an appropriate anisotropic Sobolev space as above. We define then the spaces of even (bosonic) and odd (fermionic) eigenstates:

$$C^{\mathrm{even}}(f,\lambda) := \bigoplus_{k \equiv 0 (\bmod 2)} C^k(f,\lambda), \text{ and } C^{\mathrm{odd}}(f,\lambda) := \bigoplus_{k \equiv 1 (\bmod 2)} C^k(f,\lambda)$$

The **fermion number** operator $(-1)^F$ acts on

$$C(f,\lambda) = C^{\text{even}}(f,\lambda) \oplus C^{\text{odd}}(f,\lambda)$$

with eigenvalue ± 1 depending on the parity of the state. Let now $\theta : \mathbb{R} \to \mathbb{C}$. We define the **super**⁸**-trace** as follows:

$$\operatorname{Str}\left(\theta\left(\mathcal{L}_{V_{f}}\right)\right) = \operatorname{Tr}\left((-1)^{F}\theta\left(\mathcal{L}_{V_{f}}\right)\right)$$

$$= \operatorname{Tr}\left(\theta\left(\mathcal{L}_{V_{f}}\right)\right)_{C^{\operatorname{even}}(f,\lambda)} - \operatorname{Tr}\left(\theta\left(\mathcal{L}_{V_{f}}\right)\right)_{C^{\operatorname{odd}}(f,\lambda)}\right)$$

$$:= \sum_{\lambda \in \cup_{k=0}^{n} \mathcal{R}_{k}(f,g)} \theta(\lambda)\left(\operatorname{dim}C^{\operatorname{even}}(f,\lambda) - \operatorname{dim}C^{\operatorname{odd}}(f,\lambda)\right).$$

This allows to define a notion of super-trace as soon as the last quantity is well-defined. In order to avoid too many complications that would be beyond the scope of this article, we take this as a definition of the trace in our framework. We note that this is related to the notion of flat trace – see e.g. [28, Sect. 2.4].

The operators d and i_{V_f} both commute with \mathcal{L}_{V_f} . Hence, $Q = (d + i_{V_f})$ defines an operator

$$Q_{\lambda}: C^{\text{even}}(f, \lambda) \oplus C^{\text{odd}}(f, \lambda) \mapsto C^{\text{odd}}(f, \lambda) \oplus C^{\text{even}}(f, \lambda).$$

⁸One more time, the prefix super just emphasizes the fact that we are considering functions of odd (dz_i) and even (z_i) variables.

which exchanges chiralities. We observe that, for every $\lambda > 0$, Q_{λ} is an isomorphism since $Q_{\lambda}^2 = \mathcal{L}_{V_f} = \lambda \mathrm{Id}$. In particular, for every $\lambda > 0$, one has

$$\dim C^{\text{even}}(f,\lambda) = \dim C^{\text{odd}}(f,\lambda).$$

Combined with Proposition 6.8, this implies

$$\operatorname{Str}\left(\theta\left(\mathcal{L}_{V_f}\right)\right) = \theta(0) \sum_{k=0}^{n} (-1)^k |\{a \in \operatorname{Crit}(f) : \operatorname{ind}(a) = k\}| = \theta(0) \sum_{a \in \operatorname{Crit}(f)} (-1)^{\operatorname{ind}(a)}.$$

By the classical Morse inequalities, the right-hand side of this equality is equal to $\theta(0)\chi(M)$, where $\chi(M)$ is the Euler characteristic of M. We shall prove this property in section 8.

Let us now specialize this result when we take $\theta(\lambda) = e^{-\lambda t} \mathbf{1}_{[0,\Lambda]}(\lambda)$ some fixed $\Lambda > 0$. In that case, we recover the following version of the Atiyah–Bott–Lefschetz fixed point Theorem [3]:

Proposition 7.10. Let φ_f^t be a Morse-Smale gradient flow all of whose Lyapunov exponents are rationally independent. Then, one has, for every $\Lambda > 0$, and for every t > 0,

(34)
$$\sum_{k=0}^{n} (-1)^k \operatorname{Tr} \left(\Pi_{\Lambda}^{(k)} \varphi_f^{-t*} \Pi_{\Lambda}^{(k)} \right) = \sum_{x=\varphi_f^{-t}(x)} \frac{\det(\operatorname{Id} - d_x \varphi_f^{-t})}{|\det(\operatorname{Id} - d_x \varphi_f^{-t})|},$$

where $\Pi_{\Lambda}^{(k)}$ is the spectral projector defined in paragraph 5.3 and Tr is the standard trace.

In the terminology of [3], the left-hand side of (34) is called the **Lefschetz number** of φ_f^{-t*} (more precisely of $\Pi_{\Lambda}^{(k)}\varphi_f^{-t*}\Pi_{\Lambda}^{(k)}$). As was already mentionned, we will verify in the next section that the right-hand side is equal to the Euler characteristic $\chi(M)$ of M. Note that, after integrating the previous equality against $t^{s-1}e^{-zt}$ between 0 and $+\infty$, one can write the following expression for the spectral (super-)zeta function of $\mathcal{L}_{V_f} + z$:

$$\zeta(s,z) := \frac{1}{\Gamma(s)} \sum_{k=0}^n (-1)^k \int_0^{+\infty} t^s e^{-zt} \operatorname{Tr} \left(\Pi_{\Lambda}^{(k)} \varphi_f^{-t*} \Pi_{\Lambda}^{(k)} \right) \frac{dt}{t} = \frac{\chi(M)}{z^s}.$$

Up to the normalizing factors s t and $\Gamma(s)$, this function looks like a (spectral) analogue of the semiclassical zeta function of Faure and Tsujii [33, Sect. 1]. In particular, at s=0, this is formally equal to $\chi(M)$. If we differentiate this expression with respect to s and evaluate it at 0, we find that $e^{-\partial_s \zeta(0,z)} = z^{\chi(M)}$. Equivalently, the super-determinant of $(\mathcal{L}_{V_f} + z)$ verifies:

Corollary 7.11. Let φ_f^t be a Morse-Smale gradient flow all of whose Lyapunov exponents are rationally independent. Then, one has, for every $\Lambda > 0$ and for every z in \mathbb{C}^* ,

(35)
$$\prod_{k=0}^{n} \det \left(\Pi_{\Lambda}^{(k)} (\mathcal{L}_{V_f}^{(k)} + z) \Pi_{\Lambda}^{(k)} \right)^{(-1)^k} = z^{\chi(M)}.$$

⁹In some sense, this allows to take into account the fact that, in our dynamical framework, there are fixed points.

where $\Pi_{\Lambda}^{(k)}$ is the spectral projector defined in paragraph 5.3 and det is the standard determinant

8. Topological considerations

Studying Morse functions has deep connections with the topology of the manifold, and we will now describe some topological consequences of our spectral analysis of the operator $\mathcal{L}_{V_f}^{(*)}$. In all this section, we still suppose that (f,g) is a smooth Morse pair inducing a Morse-Smale gradient flow but we do not suppose a priori that the Lyapunov exponents are rationally independent. The results presented here are in fact related to the description of the Morse complex given by Laudenbach in [53, 54] and to the interpretation of Morse theory given by Harvey and Lawson in [46]. The main novelty here is the spectral interpretation of these results in analogy with Hodge-de Rham theory.

8.1. **De Rham cohomology.** We start with a brief reminder on de Rham cohomology [63, 66]. Recall that, for every $k \geq 0$, the coboundary operator d sends any element in $\Omega^k(M)$ to an element in $\Omega^{k+1}(M)$, and that it satisfies $d \circ d = 0$. In particular, one can define a cohomological complex $(\Omega^*(M), d)$ associated with d:

$$0 \to \Omega^0(M) \to \Omega^1(M) \to \ldots \to \Omega^n(M) \to 0.$$

This complex is also called the de Rham complex. An element ω in $\Omega^*(M)$ such that $d\omega = 0$ is called a **cocycle** while an element ω which is equal to $d\alpha$ for some $\alpha \in \Omega^*(M)$ is called a **coboundary**. We define then

$$Z^k(M) = \operatorname{Ker}(d) \cap \Omega^k(M)$$
, and $B^k(M) = \operatorname{Im}(d) \cap \Omega^k(M)$.

Obviously, $B^k(M) \subset Z^k(M)$, and the quotient space $\mathbb{H}^k(M) = Z^k(M)/B^k(M)$ is called the k-th de Rham cohomology.

According to [66, p. 344-345], the coboundary operator d can be extended into a map acting on the space of currents. This allows to define another cohomological complex $(\mathcal{D}'^{*}(M), d)$:

$$0 \to \mathcal{D}'^{,0}(M) \to \mathcal{D}'^{,1}(M) \to \ldots \to \mathcal{D}'^{,n}(M) \to 0,$$

where we recall that $\mathcal{D}^{\prime,k}(M)$ is the topological dual of $\Omega^{n-k}(M)$. One can similarly define the k-th cohomology of that complex. A remarkable result of de Rham is that these two cohomologies coincide [63, Ch. 4] – see also [66, p.355] for a generalization of this result.

Theorem 8.1 (de Rham). Let u be an element in $\mathcal{D}^{\prime,k}(M)$ satisfying du=0.

- (1) There exists ω in $\Omega^k(M)$ such that $u \omega$ belongs to $\operatorname{Im}(d) \cap \mathcal{D}'^{k}(M)$.
- (2) If u = dv with (u, v) in $\Omega^k(M) \times \mathcal{D}^{\prime,k-1}(M)$, then there exists ω in $\Omega^{k-1}(M)$ such that $u = d\omega$.

Remark 8.2. In the following, we will need something slightly more precise than (i). Namely suppose that u is a cocycle in $\mathcal{H}_k^{m+k}(M)$. We denote by $\Delta_g^{(k)}$ the Laplace-Beltrami operator acting on $L^2(M, \Lambda^k(T^*M))$. We can find a pseudodifferential operator A_k of order -2 such that $u - \Delta_g^{(k)} A_k u$ belongs to $\Omega^k(M)$. As u is a cocycle, one can deduce that $d\Delta_g^{(k)} A_k u \in$

 $\Omega^{k+1}(M)$. From the ellipticity of Δ_g , we find that $dA_k u \in \Omega^{k+1}(M)$. This implies that $d^*dA_k u \in \Omega^k(M)$ and thus $u - dd^*A_k(u)$ belongs to $\Omega^k(M)$. As u belongs to $\mathcal{H}_k^{m+k}(M)$, we get a refinement for point (i) in the sense that $u - \omega = dd^*A_k(u)$ belongs to $d(\mathcal{H}_{k-1}^{m+k-1}(M))$.

8.2. Finite dimensional complexes. Proving that the k-th cohomology is finite dimensional requires more work – see e.g. [63, Ch. 4]. Before deducing that result from our spectral analysis of \mathcal{L}_{V_f} , we start with some general considerations on finite dimensional cohomological complexes. Consider a cohomological complex (C^*, d) associated with the coboundary operator:

$$0 \to C^0 \to C^1 \to \dots \to C^n \to 0,$$

where for every $0 \le k \le n$, C^k is a finite dimensional subspace of $\mathcal{D}^{\prime,k}(M)$.

Remark 8.3. A famous example of such complexes appears in Hodge theory where one considers $C^k(\Delta) = \text{Ker}(\Delta_g^{(k)})$. From the ellipticity of the Laplace-Beltrami operator on that space, one can deduce the fact that $C^k(\Delta)$ is a finite dimensional space inside $\Omega^k(M)$.

Consider now the complexes induced by the operator \mathcal{L}_{V_f} . For that purpose, we pick N_0 and N_1 large enough in the definition of the order function $m(x,\xi)$, and we define

$$C^k(f) := \operatorname{Ker}\left(\mathcal{L}_{V_f}^{(k)}\right)$$

which is a finite dimensional space. Recall one more time from [32, Th. 1.5] that these spaces are intrinsic in the sense that they do not depend on the choice of the order function m. As d commutes with the Lie derivative \mathcal{L}_{V_f} , one can verify that, if $\mathcal{L}_{V_f}^{(k)}u = 0$ with u in $\mathcal{H}_k^m(M)$, then $\mathcal{L}_{V_f}^{(k+1)}(du) = 0$ with du belonging to $\mathcal{H}_{k+1}^{m+1}(M)$. Hence, the coboundary operator d induces a finite dimensional cohomological complex $(C^*(f), d)$

$$0 \to C^0(f) \to C^1(f) \to \dots \to C^n(f) \to 0.$$

We can now apply Proposition 6.8 and we find that $\dim(C^k(f))$ is in fact equal to the number $c_k(f)$ of critical point of f which are of index k.

8.3. Morse type inequalities. Consider a finite dimensional complex (C^*, d) . We briefly recall how to obtain Morse type inequalities in that abstract framework arguing as in [54, Ch. 6]. For that purpose, we define

$$Z^k(C^*) = \operatorname{Ker}(d) \cap C^k$$
, and $B^k(C^*) = \operatorname{Im}(d) \cap C^k$.

As above, we define the quotient space (or the k-th cohomology of the complex):

$$\mathbb{H}^k(C^*) := Z^k(C^*)/B^k(C^*).$$

We denote by $\beta_k(C^*) < \infty$ the dimension of that quotient space. We also introduce

$$b_k(C^*) = \dim B^k(C^*), \ c_k(C^*) = \dim C^k, \ \text{and} \ z_k(C^*) = \dim Z^k(C^*).$$

We observe that

$$\beta_k(C^*) = z_k(C^*) - b_k(C^*)$$
 and $c_k(C^*) = b_{k+1}(C^*) + z_k(C^*)$.

We now write that, for every k > 0,

$$0 \le b_{k+1}(C^*) = (c_k(C^*) - \beta_k(C^*)) - (c_{k-1}(C^*) - \beta_{k-1}(C^*)) + \dots$$

From this expression, we can deduce the following Morse type inequalities associated with the complex (C^*, d) :

(36)
$$\forall 0 \le k \le n, \ \sum_{j=0}^{k} (-1)^{k-j} c_j(C^*) \ge \sum_{j=0}^{k} (-1)^{k-j} \beta_j(C^*),$$

and using $b_{n+1}(C^*) = 0$:

(37)
$$\sum_{j=0}^{n} (-1)^{n-j} c_j(C^*) = \sum_{j=0}^{n} (-1)^{n-j} \beta_j(C^*).$$

In the case where we pick $C^* = C^*(f)$, inequalities (36) and (37) are exactly the Morse inequalities for the complex $C^*(f)$ which is nothing else but the Morse complex (also called Thom-Smale-Witten complex).

- 8.4. The Morse complex is isomorphic to the de Rham complex. Let $0 \le k \le n$. We would like now to give a spectral proof of the fact that the k-th cohomology of the Morse complex is isomorphic to the de Rham cohomology of degree k. A proof of this result based on the theory of currents can be found in [54, Ch. 6] in the case of locally flat metrics. Here, we give an alternative proof of that result based on our spectral analysis of the operator \mathcal{L}_{V_f} . The main idea is just that the Morse complex is the limit as $t \to +\infty$ of the de Rham complex evolved under the semi-group φ_f^{-t*} . This observation is also contained in the works of Harvey and Lawson on finite volume flows [45, 46].
- 8.4.1. Propagating the de Rham complex in finite time. For every t > 0, we can define a complex $(\varphi_f^{-t*}\Omega^*(M), d)$. This complex is homotopic to the de Rham complex $(\Omega^*(M), d)$ via the **cochain homotopy equation**:

$$(38) u - \varphi_f^{-t*} u = dR_t u + R_t du$$

(39)
$$R_t := \int_0^t i_{V_f} \circ \varphi_f^{-s*} ds,$$

where i_{V_f} is the contraction operator by the vector field V_f . This formula can be obtained as follows:

$$u - \varphi_f^{-t*}u = \int_0^t \mathcal{L}_{V_f} \circ \varphi_f^{-s*}uds = \int_0^t (d \circ i_{V_f} + i_{V_f} \circ d) \circ \varphi_f^{-s*}uds,$$

where the first equality is just the Taylor formula while the second equality follows from Cartan's formula [66, p. 351]: $\mathcal{L}_{V_f}^{(k)} = i_{V_f} \circ d + d \circ i_{V_f}$. In particular, one can verify from these relations that the semi-group φ_f^{-t*} induces an **isomorphism** in cohomology $H(\Omega^*(M), d) \simeq H(\varphi_f^{-t*}\Omega^*(M), d)$. Equivalently, the cohomologies of the two complexes are isomorphic via the map φ_f^{-t*} . In order to prove that the Morse complex $(C^*(f), d)$ is isomorphic (in cohomology) to $(\Omega^*(M), d)$, we now have to let t tends to $+\infty$ in the above

homotopy equation. This can be done rigorously thanks to the fact that there 0 is an isolated eigenvalue in the spectrum of \mathcal{L}_{V_f} and thanks to the spectral projector associated with the eigenvalue 0. This will exactly be the content of the next paragraph. Once the limit homotopy equation will be settled, we will still have to verify that it induces an isomorphism in cohomology between the de Rham and Morse complex – see paragraph 8.4.3.

Remark 8.4. Note that, in Hodge theory, the strategy is exactly the same except that we replace our semigroup φ_f^{-t*} by the heat flow $e^{-t\Delta_g}$.

8.4.2. Spectral decomposition. Let $m(x,\xi)$ be an order function with N_0 and N_1 sufficiently large to ensure that 0 is an isolated eigenvalue with finite algebraic multiplicity (eventually equal to 0). Introduce the spectral projector associated with the eigenvalue 0:

$$\mathbb{P}^{(k)} := \int_{\gamma} \frac{dz}{(z - \mathcal{L}_{V_f}^{(k)})} : \mathcal{H}_k^m(M) \to C_k(f),$$

where γ is a small Jordan path which separates 0 from the rest of the spectrum of $\mathcal{L}_{V_f}^{(l)}$ acting on $\mathcal{H}_l^m(M)$ for every $0 \leq l \leq n$ – see [48, App. A]. This operator commutes with $\mathcal{L}_{V_f}^{(k)}$. According to [30, p.244-246], one knows that

$$\mathcal{L}_{V_t}^{(k)}: \left(\operatorname{Id}_{\mathcal{H}_k^m} - \mathbb{P}^{(k)}\right) \mathcal{H}_k^m(M) \to \left(\operatorname{Id}_{\mathcal{H}_k^m} - \mathbb{P}^{(k)}\right) \mathcal{H}_k^m(M)$$

does not contain 0 in its spectrum. In particular, we can write the following decomposition:

$$\operatorname{Id}_{\mathcal{H}_k^m} = \mathbb{P}^{(k)} + \mathcal{L}_{V_f}^{(k)} \circ \left((\mathcal{L}_{V_f}^{(k)})^{-1} \circ \left(\operatorname{Id}_{\mathcal{H}_k^m} - \mathbb{P}^{(k)} \right) \right).$$

By Cartan's formula [66, p. 351], one knows that $\mathcal{L}_{V_f}^{(k)} = i_{V_f} \circ d + d \circ i_{V_f}$. Hence,

(40)
$$\operatorname{Id}_{\mathcal{H}_k^m} = \mathbb{P}^{(k)} + (d \circ i_{V_f} + i_{V_f} \circ d) \circ \left((\mathcal{L}_{V_f}^{(k)})^{-1} \circ \left(\operatorname{Id}_{\mathcal{H}_k^m} - \mathbb{P}^{(k)} \right) \right).$$

Moreover, d commutes with \mathcal{L}_{V_f} hence with $\mathbb{P}^{(k)}$ from the expression of the spectral projector. Hence, for every u in $(\mathrm{Id} - \mathbb{P}^{(k)})\mathcal{H}_k^m(M)$, $du \in (\mathrm{Id} - \mathbb{P}^{(k+1)})\mathcal{H}_{k+1}^{m+1}(M)$, and one has $\mathcal{L}_{V_f}^{-1} \circ du = \mathcal{L}_{V_f}^{-1} \circ d \circ \mathcal{L}_{V_f} \circ \mathcal{L}_{V_f}^{-1}(u) = d \circ \mathcal{L}_{V_f}^{-1}(u)$. From that, we infer that d also commutes with $\mathcal{L}_{V_f}^{-1} \circ (\mathrm{Id}_{\mathcal{H}_k^m} - \mathbb{P}^{(k)})$. Combining this last observation with the fact that d commutes with $\mathbb{P}^{(k)}$ and with (40), we finally find that, for every u in $\mathcal{H}_k^m(M)$, one has

(41)
$$u = \mathbb{P}^{(k)}(u) + d \circ R_{\infty}^{(k)}(u) + R_{\infty}^{(k+1)} \circ d(u),$$

where $R_{\infty}^{(n+1)} = 0$ and, for every $0 \le l \le n$,

$$R_{\infty}^{(l)} := i_{V_f} \circ (\mathcal{L}_{V_f}^{(l)})^{-1} \circ \left(\operatorname{Id}_{\mathcal{H}_l^m} - \mathbb{P}^{(l)} \right).$$

Remark 8.5. Recall that, from our complete description of the spectrum of \mathcal{L}_{V_f} , one has, for every u in $\mathcal{H}_k^m(M)$,

$$\mathbb{P}^{(k)}(u) = \lim_{t \to +\infty} \varphi_f^{-t*} u = \sum_{a \in \operatorname{Crit}(f): \operatorname{ind}(a) = k} \langle S_a, u \rangle U_a \in C^k(f).$$

We note that our cochain homotopy equation should be thought as the limit as $t \to +\infty$ of the cochain homotopy equation between the de Rham complex $(\Omega^*(M), d)$ and the propagated complex $(\varphi_f^{-t*}\Omega^*(M), d)$. Up to its spectral interpretation, this type of "limit homotopy equation" already appears in the works of Harvey and Lawson [46, Th. 2.3].

Remark 8.6. In Hodge theory, the analogue of this decomposition would consist in using the spectral projector associated with the eigenvalue 0 of $\Delta_g^{(k)} = dd^* + d^*d$, this projects currents on **harmonic** forms. Using the same convention for the spectral projector associated with 0, we would get

$$u = \mathbb{P}^{(k)}(u) + d \circ R_{\infty}^{(k)} u + R_{\infty}^{(k+1)} \circ d(u),$$

where, for every $0 \le k \le n$,

$$R_{\infty}^{(l)} := d^* \circ (\Delta_q^{(l)})^{-1} \circ (\operatorname{Id} - \mathbb{P}^{(l)})$$
.

As we shall see below, this relation is the key step to prove that the de Rham cohomology is isomorphic to harmonic forms.

8.4.3. Cohomological consequences. As the coboundary operator d commutes with \mathcal{L}_{V_f} , it also commutes with $\mathbb{P}^{(k)}$. In particular, the map $\mathbb{P}^{(k)}$ induces a map from $Z^k(M)$ to $Z^k(C^*(f))$. We will now show (using our spectral approach) that it induces an isomorphism between the quotient spaces:

Proposition 8.7. Let $0 \le k \le n$. The map

$$\mathbb{P}^{(k)}:\Omega^k(M)\to C^k(f)$$

induces an isomorphism between the vector spaces $\mathbb{H}^k(C^*(f),d)$ and $\mathbb{H}^k(M)$.

From that Proposition, we recover the classical fact that $\mathbb{H}^k(M)$ is a finite dimensional space for every $0 \leq k \leq n$. Its dimension is called the k-th Betti number that we will denote by $b_k(M)$. With the notations of paragraph 8.2, we have $b_k(M) = \beta_k(C^*(f))$ for every $0 \leq k \leq n$. In particular, if we apply (36) and (37) in the case of the complex $(C^*(f), d)$, we recover the classical Morse inequalities:

Corollary 8.8 (Morse inequalities). Let

$$c_k(f) = |\{a \in \operatorname{Crit}(f) \ s.t. \ \operatorname{ind}(a) = k\}|.$$

Then, for all $k \in \{0, ..., n\}$, we have:

$$\sum_{j=0}^{k} (-1)^{k-j} c_j(f) \ge \sum_{j=0}^{k} (-1)^{k-j} b_j(M),$$

with equality in the case¹⁰ k = n.

¹⁰ Recall that in that case, the sum is the Euler characteristic $\chi(M)$ of M.

Proof of Proposition 8.7. Let us start with injectivity. Let u be a cocycle in $\Omega^k(M)$ such that $\mathbb{P}^{(k)}(u) = 0$. We use equality (41), and we find that

$$u = d \circ R_{\infty}^{(k)}(u),$$

which exactly says that u is a coboundary for the complex $(\mathcal{D}'^{,*}(M), d)$. As u is smooth, we know from de Rham Theorem 8.1 that u is a coboundary in $\Omega^k(M)$.

Let us now consider the surjectivity. Fix u a cocycle in $\operatorname{Ker}(\mathcal{L}_{V_f}^{(k)})$. From Remark 8.2, we know that there exists $\omega \in \Omega^k(M)$ and v in $\mathcal{H}_{k-1}^{m+k-1}(M)$ such that $u-\omega=dv$. Writing the cochain homotopy equation (41) for ω , we find that

$$\omega = \mathbb{P}^{(k)}(\omega) + d \circ R_{\infty}^{(k)}(\omega).$$

This implies that

$$u = \mathbb{P}^{(k)}(\omega) + d\left(R_{\infty}^{(k)}(\omega) + v\right).$$

By construction, $R_{\infty}^{(k)}(\omega)+v$ belongs to $\mathcal{H}_{k-1}^{m+k-1}(M)$. Hence, applying the spectral projector to the previous equality and as d commutes with $\mathbb{P}^{(k)}$ (thanks to the integral expression of the spectral projector), we find that

$$u = \mathbb{P}^{(k)}(\omega) + d \circ \mathbb{P}^{(k)} \left(R_{\infty}^{(k)}(\omega) + v \right),$$

which proves the surjectivity.

Remark 8.9. In Hodge theory, the surjectivity is in fact slightly simpler provided we use some results from elliptic theory in a slightly different form. In fact, consider a harmonic form u i.e. such that $\Delta_g u = 0$. From the ellipticity of the operator Δ_g , we know that u is smooth and thus belongs to $\Omega^{(*)}(M)$.

Remark 8.10. We note that it would be tempting to consider the complex associated with a nonzero eigenvalue λ of \mathcal{L}_{V_f} . Yet, this complex is trivial from the point of view of cohomology. In fact, if du = 0 and $(\mathcal{L}_{V_f} - \lambda)u = 0$ for some $\lambda \neq 0$, then we have that

$$\mathcal{L}_{V_f}u = \lambda u.$$

Combining the Cartan formula to the fact that du = 0, we would get

$$d \circ i_{V_f} u = \lambda u.$$

As $\lambda \neq 0$ and as i_{V_f} commutes with \mathcal{L}_{V_f} , we find that u is a coboundary of $(C^*(f), d)$. In particular, if we fix some (finite or not) subset A containing 0 inside $\bigcup_{0 \leq k \leq n} \mathcal{R}_k(f, g)$, then one can define the complex $(C^*(f, A), d)$ such that

$$\forall 0 \leq k \leq n, \ C^k(f,A) := \operatorname{span}\{U_{a,k}^{\alpha,j}: \ a \in \operatorname{Crit}(f), \ \alpha. |\chi(a)| \in A \text{ and } 1 \leq j \leq m_{k,a}(\alpha)\}.$$

From the above observation, one can easily deduce that the complex $(C^*(f, A), d)$ is also isomorphic to the de Rham complex via the map

$$\mathbb{P}_{A}^{(k)}(\psi) := \sum_{a \in \operatorname{Crit}(f)} \sum_{\alpha \in \mathbb{N}^{n}: \alpha . |\chi(a)| \in A} \sum_{j=1}^{m_{k,a}(\alpha)} \langle S_{a,k}^{\alpha,j}, \psi \rangle U_{a,k}^{\alpha,j}.$$

8.5. **Poincaré duality and** $f \mapsto -f$. By construction, one knows that the currents $(S_a)_{a \in \operatorname{Crit}(f)}$ is the dual basis to $(U_a)_{a \in \operatorname{Crit}(f)}$ for the duality bracket between $\mathcal{H}_k^m(M)$ and $\mathcal{H}_{n-k}^{-m}(M)$ which coincides (in the case of smooth forms) with the standard duality bracket between $\mathcal{D}'^{,*}(M)$ and $\Omega^{n-*}(M)$. Moreover, from paragraph 6.4.4, it is in fact a basis of the kernel of the operator $\mathcal{L}_{V_{-f}}^{(*)}$ acting on $\mathcal{H}_{n-*}^{-m}(M)$. We set

$$C^{n-k}(-f) := \operatorname{Ker}(\mathcal{L}_{V_{-f}}^{(n-k)}).$$

We can then define the following complex associated with the coboundary operator d:

$$0 \to C^0(-f) \to \dots \to C^{n-1}(-f) \to C^n(-f) \to 0.$$

As was already explained, the two complexes $(C^*(f), d)$ and $(C^*(-f), d)$ are dual to each other via the duality between $\mathcal{H}_k^m(M)$ and $\mathcal{H}_{n-k}^{-m}(M)$, i.e.

$$\forall (u,v) \in C^k(f) \times C^{n-k}(-f), \ \langle u,v \rangle = \langle u,v \rangle_{\mathcal{H}_k^m(M),\mathcal{H}_{n-k}^{-m}(M)} = \int_M u \wedge v.$$

Introduce now the following **Poincaré isomorphism** between $C^k(f)$ and the dual of $C^{n-k}(-f)$:

$$\mathcal{P}_0^{(k)}: u \in C^k(f) \mapsto \langle u, . \rangle \in C^{n-k}(-f)'.$$

We observe that $\langle u,v\rangle$ does not depend on the cohomology class of u and v. Hence, $\mathcal{P}_0^{(k)}$ induces a linear map between $\mathbb{H}^k(C^*(f),d)$ and $\mathbb{H}^{n-k}(C^*(-f),d)'$. We now follow closely [54, Ch. 6] and verify that this is in fact an isomorphism between the quotient spaces. Suppose that θ is a linear form on $\mathbb{H}^{n-k}(C^*(-f),d)$. This induces a linear form θ on $Z^{n-k}(C^*(-f),d)$ which vanishes on $B^{n-k}(C^*(-f),d)$. By the Hahn-Banach Theorem, we extend this linear form to $C^{n-k}(-f)$. From the duality between $C^k(f)$ and $C^{n-k}(-f)$, there exists a unique u in $C^k(f)$ such that $\theta(v) = \langle u,v\rangle$ for every v in $C^{n-k}(-f)$. As θ vanishes on the image of d, we find that, for every v in $C^{n-k-1}(-f)$, $\langle u,dv\rangle=0$ from which one can deduce that du=0. This shows surjectivity of the linear map induced by $\mathcal{P}_0^{(k)}$. If we intertwine the role of f and -f, we find a linear surjection from $\mathbb{H}^{n-k}(C^*(-f),d)$ to $\mathbb{H}^k(C^*(f),d)'$. This implies that all the spaces have the same dimension. Hence, $\mathcal{P}_0^{(k)}$ induces an isomorphism between $\mathbb{H}^k(C^*(f),d)$ and $\mathbb{H}^{n-k}(C^*(-f),d)'$ for every $0 \leq k \leq n$. Combined with Proposition 8.7 applied to both f and -f, this implies the following well known result:

Proposition 8.11. Let M be a smooth, compact, oriented manifold without boundary. Then, for every $0 \le k \le n$, $\mathcal{P}_0^{(k)}$ induces an isomorphism between $\mathbb{H}^k(C^*(f),d)$ and $\mathbb{H}^{n-k}(C^*(-f),d)'$. In particular, $b_k(M) = b_{n-k}(M)$ for every $0 \le k \le n$.

Remark 8.12. This discussion could be generalized to the complex $(C^k(f,A),d)$ defined in Remark 8.10. For that purpose, we should observe that, for every A inside $\bigcup_{0 \le k \le n} \mathcal{R}_k(f,g)$, $C^*(f,A)$ is a subspace of $\mathcal{D}_{\Gamma_-}^{\prime,*}(M)$ while $C_*(f,A)$ is a subspace of $\mathcal{D}_{\Gamma_+}^{\prime,*}(M)$. The duality pairing $\langle u,v\rangle$ is then given by the duality pairing $\int_M u \wedge v$ between $\mathcal{D}_{\Gamma_-}^{\prime,*}(M)$ and $\mathcal{D}_{\Gamma_+}^{\prime,n-*}(M)$ which is well defined as $\Gamma_- \cap \Gamma_+ = M \times \{0\}$. Note that it coincides with the duality pairing

between the anisotropic Sobolev spaces when we consider elements inside the complexes $C(\pm f, A)$. This would yield a generalized Poincaré isomorphism:

$$\mathcal{P}_A^{(k)}: U_{a,k}^{\alpha,j} \in C^k(f,A) \mapsto \langle U_{a,k}^{\alpha,j}, . \rangle \in C^{n-k}(-f,A)'.$$

Yet, from the point of view of the cohomology, it would not give more informations – see Remark 8.10.

8.6. Koszul complex associated with i_{V_f} . As was already alluded in the introduction, the Cartan formula

$$\mathcal{L}_{V_f} = d \circ i_{V_f} + i_{V_f} \circ d,$$

replaces in our context the formula $\Delta = d \circ d^* + d^* \circ d$ in Hodge theory. Hence, what plays the role in the Morse context of the complex $(\Omega^*(M), d^*)$ from Hodge theory is the Koszul complex induced by the contraction operator i_{V_f} .

Remark 8.13. We underline that the Cartan formula combined with our spectral decomposition yields an analogue of the Hodge decomposition in our framework:

$$u = \mathbb{P}^{(k)}(u) + d\left(i_{V_f} \circ (\mathcal{L}_{V_f}^{(k)})^{-1} \circ (\mathrm{Id} - \mathbb{P}^{(k)})(u)\right) + i_{V_f}\left(d \circ (\mathcal{L}_{V_f}^{(k)})^{-1} \circ (\mathrm{Id} - \mathbb{P}^{(k)})(u)\right).$$

In other words, any u in $\Omega^k(M)$ can be decomposed as the sum of an invariant current, of a coboundary (for d) and of a boundary (for i_{V_f}).

We now consider the Morse-Koszul homological complex $(C^*(f), i_{V_f})$

$$0 \to C^n(f) \to C^{n-1}(f) \to \dots \to C^0(f) \to 0.$$

Again, this is a well defined complex as i_{V_f} commutes with the Lie derivative \mathcal{L}_{V_f} . It is naturally associated with the homological complex $(\Omega^*(M), i_{V_f})$:

$$0 \to \Omega^n(M) \to \Omega^{n-1}(M) \to \ldots \to \Omega^0(M) \to 0.$$

Recall that the Euler characteristic of a homological complex (C^*, i) is given by

$$\chi(C^*, i) = \sum_{j=0}^{n} (-1)^j \dim \left(Z_j(C^*, i) / B_j(C^*, i) \right),$$

where

$$Z_j(C^*,i) := \operatorname{Ker}(i) \cap C^j$$
, and $B_j(C^*,i) := \operatorname{Im}(i) \cap C^j$.

We start our discussion on the Morse-Koszul complex with the following property

Proposition 8.14. Let (f,g) be a smooth Morse pair inducing a Morse-Smale gradient flow. Then, one has

$$\chi(C^*(f), i_{V_f}) = \chi(M),$$

where $\chi(M)$ is the Euler characteristic of the manifold.

Proof. Recall that $C^k(f)$ is equal to the vector space generated by the Laudenbach currents U_a associated with critical points a of index k. According to Proposition 6.7, near a critical point a of index k, U_a can be written in the adapted coordinates of paragraph 3.5 as

$$U_a(x, y, dx, dy) = \delta_0^{\mathbb{R}^k}(x) dx^1 \wedge dx^2 \wedge \dots dx^k.$$

On the other hand, the vector field V_f can be written in this system of coordinates:

$$V_f(x, y, \partial_x, \partial_y) = \sum_{i=1}^r \chi_j(a) x_j \partial_{x_j} + \sum_{j=r+1}^n \chi_j(a) y_j \partial_{y_j}.$$

Hence, locally near a, one has

$$i_{V_f}(U_a)(x, y, dx, dy) = \sum_{j=1}^r \chi_j(a) x_j \delta_0^{\mathbb{R}^k}(x) dx_1 \wedge \dots \widehat{dx_j} \dots \wedge dx_r = 0.$$

As U_a is supported in $\overline{W^u(a)}$, we can deduce that $i_{V_f}(U_a)$ is also carried by $\overline{W^u(a)}$. As we have just shown that it is equal to 0 near a and as $\mathcal{L}_{V_f}(i_{V_f}(U_a)) = 0$, we can deduce that the support of $i_{V_f}(U_a)$ is contained in $\overline{W^u(a)} - W^u(a)$. According to Remark 3.7, we can then deduce that the support of $i_{V_f}(U_a)$ is contained in the union of unstable manifold $W^u(b)$ with $\operatorname{ind}(b) > k$. We now use Proposition 6.8 to write

$$i_{V_f}(U_a) = \sum_{b': \operatorname{ind}(b') = k-1} \alpha_{b'} U_{b'}.$$

Using Proposition 6.7 and the fact that $i_{V_f}(U_a)$ is carried on a union of unstable manifold of index > k, we can deduce that $\alpha_{b'} = 0$ for every critical point b' of index k-1. In other words, $Z_k(C^*(f), i_{V_f}) = C^k(f)$ and $B_k(C^*(f), i_{V_f}) = \{0\}$. In particular, one has

$$\chi(C^*(f), i_{V_f}) = \sum_{j=0}^{n} (-1)^j c_{n-j}(f),$$

from which the result follows thanks to the case of equality in the Morse inequalities.

Remark 8.15. We note that we have implicitely shown that the k-th homology $\mathbb{H}_k(C^*(f), i_{V_f})$ is equal to $C^k(f)$.

As in the case of the Thom-Smale-Witten complex, it would be natural to expect that, as i_{V_f} commutes with $\mathbb{P}^{(k)}$, $\mathbb{P}^{(*)}$ induces an isomorphism between the homological complexes $(C^*(f), i_{V_f})$ and $(\Omega^*(M), i_{V_f})$. In fact, one has also, for every time t > 0, a chain homotopy equation between the complexes $(\Omega^*(M), i_{V_f})$ and $(\varphi_f^{-t*}\Omega^*(M), i_{V_f})$. Precisely, one can write

$$(42) u - \varphi_f^{-t*} u = i_{V_f} R_t u + R_t i_{V_f} u$$

$$(43) R_t := \int_0^t d \circ \varphi_f^{-s*} ds.$$

Then, we could let $t \to +\infty$ and obtain a chain homotopy equation between $(C^*(f), i_{V_f})$ and $(\Omega^*(M), i_{V_f})$. Precisely, as before, we would write

$$u = \mathbb{P}(u) + i_{V_f} \circ R_{\infty}(u) + R_{\infty} \circ i_{V_f}(u),$$

where $R_{\infty} := d \circ \mathcal{L}_{V_f}^{-1} \circ (\operatorname{Id} - \mathbb{P})$. However, it is not the case anymore that the induced morphism is a bijection. In fact, it is neither injective nor surjective as the example on \mathbb{S}^1 below will show. The reason for that phenomenon is that the Morse-Koszul complex is sensitive to the regularity of the coefficients. In fact, compared with the case of the coboundary operator, this complex is not elliptic since the contraction with the vector field V_f is hyperbolic with degenerate points at $\operatorname{Crit}(f)$.

8.6.1. A surprising example with the circle. Let us now discuss the case of a Morse function on \mathbb{S}^1 to illustrate the previous observation. Fix f to be a smooth Morse function on \mathbb{S}^1 and g to be a smooth metric on \mathbb{S}^1 . Note that such a metric necessarily satisfies the Morse-Smale condition. The fact that we have a smooth adapted coordinate chart near any critical point follows from the fact that we can apply the Sternberg-Chen Theorem in such a neighborhood as there is only one Lyapunov exponent. Consider now the Koszul complex $(\Omega^*(\mathbb{S}^1), i_{V_f})$ induced by i_{V_f} :

$$0 \to \Omega^1(\mathbb{S}^1) \to \Omega^0(\mathbb{S}^1) \to 0.$$

Let ω be a cycle in $\Omega^1(\mathbb{S}^1)$, i.e. an element satisfying $i_{V_f}(\omega) = 0$. We find that $\omega(x, dx) = 0$ except at the critical points of a but, as ω is smooth, it implies that $\omega = 0$. In particular, the homology $\mathbb{H}_1(\Omega^*(M), i_{V_f})$ is equal to 0 which is not isomorphic to $\mathbb{H}_1(C^*(f), i_{V_f})$ whose dimension is equal to $c_1(f) \neq 0$ from the proof of Proposition 8.14.

We could also consider the complex $(\mathcal{D}'^{*}(\mathbb{S}^1), i_{V_f})$:

$$0 \to \mathcal{D}'^{,1}(\mathbb{S}^1) \to \mathcal{D}'^{,0}(\mathbb{S}^1) \to 0.$$

In that case, we find that any element u in $\mathcal{D}'^{,1}(\mathbb{S}^1)$ satisfying $i_{V_f}(u) = 0$ must be supported on the critical points of a. In particular, near every critical point a of f, it is of the form

$$u = \sum_{k \le N(a)} \alpha_{a,k} \delta_a^{(k)}(x) dx,$$

where $\alpha_{a,k}$ are constants and N(a) is a finite integer. Using the fact that the vector field is generated by a Morse function, we can verify that N(a) = 0 for every critical point a in order to satisfy $i_{V_f}(u) = 0$. Thus, the first homology $\mathbb{H}_1(\mathcal{D}'^*(f), i_{V_f})$ is of dimension $c_0(f) + c_1(f) \neq c_1(f)$ as there is at least one critical point of index 0. This first calculation shows that the Koszul complex is sensitive to the regularity of the coefficients which was not the case for the de Rham complex.

We have shown the lack of surjectivity between the first homology of $(\Omega^*(M), i_{V_f})$ and $(C^*(f), i_{V_f})$. Let us now prove that the morphism between the homology of order 0 cannot be injective. For that purpose, we need to compute the image of $i_{V_f}: \Omega^1(\mathbb{S}^1) \to \Omega^0(\mathbb{S}^1)$. Let ψ be an element in $\Omega^0(\mathbb{S}^1)$ which is of the form $i_{V_f}(\omega)$ for some ω in $\Omega^1(\mathbb{S}^1)$. Fix now

a critical point a of f. In a system of adapted coordinates near a, we find that ψ must be of the form $x\psi_a(x)$. Hence, if we fix χ_a a smooth cutoff function near a, then we find that

$$\operatorname{Im}(i_{V_f}) \simeq \left(1 - \sum_a \chi_a\right) \Omega^0(\mathbb{S}^1) \oplus \bigoplus_{a \in \operatorname{Crit}(f)} x \chi_a(x) \Omega^0(\mathbb{S}^1).$$

Thus, the dimension of $\mathbb{H}_0(C^*(f), i_{V_f})$ is equal to $c_0(f) + c_1(f) \neq c_0(f)$.

8.6.2. The classical Koszul complex. As a final remark on the comparison between Koszul and Morse–Koszul, we generalize the above discussion on \mathbb{S}^1 to the case of a general manifold M. This is done by recalling classical results on the Koszul complex associated to a smooth vector bundle E over a manifold M. In our work, the relevant bundle is E = TM. Recall that a Koszul complex associated with a smooth vector bundle is defined as follows:

Definition 8.16. Let $E \to M$ be a smooth vector bundle of rank k over M, E^* is the dual bundle and choose a section $s \in \Gamma(M; E)$ and define the contraction operator i_s : $\Gamma(M, \Lambda^{\bullet+1}E^*) \to \Gamma(M, \Lambda^{\bullet}E^*)$. For every open subset $U \subset M$ (U can be taken equal to M), the contraction operator $i_s : \Gamma(U, \Lambda^{\bullet+1}E^*) \to \Gamma(U, \Lambda^{\bullet}E^*)$ is $C^{\infty}(U)$ linear. The Koszul complex $(\Gamma(\Lambda^{\bullet}E^*); i_s)$ is a complex of sheaves of C^{∞} modules:

$$0 \to \Gamma(\Lambda^k E^*) \to \cdots \to \Gamma(\Lambda^0 E^*) = C^{\infty} \to 0.$$

In each degree, $\operatorname{Ker}(i_s)$ and $\operatorname{Im}(i_s)$ are sheaves of C^{∞} modules. It follows that the homology $\mathbb{H}_{\bullet}(\Gamma(\Lambda^{\bullet}E^*);i_s)$ defined as the quotient $\operatorname{Ker}(i_s|_{\Gamma(\Lambda^{\bullet}E^*)})/\operatorname{Im}(i_s|_{\Gamma(\Lambda^{\bullet+1}E^*)})$ are sheaves of C^{∞} modules.

We next give well–known properties of the Koszul complex in the algebraic setting [39, see Proposition 1.4 p. 52], but formulated in the smooth case:

Proposition 8.17. If $s \in \Gamma(U, E)$ does not vanish on U then the complex $(\Gamma(U, \Lambda^{\bullet}E^{*}); i_{s})$ is acyclic i.e. $\mathbb{H}_{i}(\Gamma(U, \Lambda^{\bullet}E^{*}); i_{s}) = 0$ for all $i \in \{0, ..., k\}$.

If the zeros of $s \in \Gamma(M, E)$ are non degenerate then the homology vanishes in all degrees except in degree 0 where $\mathbb{H}_0(\Gamma(M, \Lambda^{\bullet}E^*); i_s) = \mathbb{R}^p$ where p is the **number of zeros** of s.

Beware that the number of zeros of s is counted set theoretically. Therefore, the Koszul complex represents homologically the zeros of our section s.

Proof. Our first step is to show that for all U, $\mathbb{H}_i(\Gamma(U, \Lambda^{\bullet}E^*), i_s) = 0$ for all i > 0. If s has isolated zeros on U, then we choose a moving coframe (e_1, \ldots, e_k) of $E^*|_U$ such that $e_i(s) = \delta_{1i}$ and we find that for every $\alpha = \sum_{i_1 < \cdots < i_p} \alpha_{i_1 \cdots i_p} e_{i_1} \wedge \cdots \wedge e_{i_p} \in \Gamma(U, \Lambda^p E^*)$, $i_s \alpha = 0$ is equivalent to the fact that $\alpha_{1i_2 \cdots i_p}|_{U \setminus \{s=0\}} = 0$ hence $\alpha_{1i_2 \cdots i_p}|_U = 0$ for all $1 < i_2 < \cdots < i_p$ since the zeros of s are isolated. The above means that e_1 does not appear in the decomposition of α hence $\alpha = i_s (e_1 \wedge \alpha)$ and $\operatorname{Ker}(i_s|_{\Gamma(U,\Lambda^i E^*)}) = \operatorname{Im}(i_s|_{\Gamma(U,\Lambda^{i+1}E^*)})$ for all i > 0.

For i = 0, if s does not vanish on U, then it is simple to see that $i_s\Gamma(U, \Lambda^1E^*)$ is onto and therefore the cohomology vanishes in all degree. Otherwise, thanks to the above property,

we can localize the proof over small open subsets U_a centered around each critical point a of s and $\mathbb{H}_0(\Gamma(M, \Lambda^{\bullet}E^*); i_s) = \bigoplus_{a \in \operatorname{zeros}(s)} \mathbb{H}_0(\Gamma(U_a, \Lambda^{\bullet}E^*); i_s)$. On U_a , using the fact that the zeros of s are non degenerate, in some coordinates $(x_i)_i$ where $x_i(a) = 0$, we find that $s = \sum_{i=1}^n x_i s_i, s_i \in \Gamma(U_a, E)$ where $s_i(0) \neq 0, \forall i$. Therefore, an immediate calculation reveals that $\mathbb{H}_0(\Gamma(U_a, \Lambda^{\bullet}E^*); i_s) = \frac{C^{\infty}(U_a)}{\mathcal{I}_a}$ where \mathcal{I}_a is the ideal of functions vanishing on a hence $\mathbb{H}_0(\Gamma(U_a, \Lambda^{\bullet}E^*); i_s) \simeq \mathbb{R}$ and $\mathbb{H}_0(\Gamma(M, \Lambda^{\bullet}E^*); i_s) = \bigoplus_{a \in \operatorname{zeros}(s)} \mathbb{R}$.

Appendix A. Proof of Lemma 4.1

In this appendix, we give the proof of Lemma 4.1, i.e. construct of the escape function $G_m(x,\xi)$. Let $N_0, N_1 > 4||f||_{\mathcal{C}^0}$ be some large parameters. As was already explained, up to some minor differences due to the special form of the dynamics, our construction is the one given in section 2 of [32]. Using the conventions of paragraph 3.4, we recall the following result [32, Lemma 2.1]:

Lemma A.1. Let V^u and V^s be small open neighborhoods of Σ_u and Σ_s respectively, and let $\epsilon > 0$. Then, there exist $\mathcal{W}^u \subset V^u$ and $\mathcal{W}^s \subset V^s$, \tilde{m} in $C^{\infty}(S^*M, [0, 1])$, $\eta > 0$ such that $\tilde{X}_{H_f}.\tilde{m} \geq 0$ on S^*M , $\tilde{X}_{H_f}.\tilde{m} \geq \eta > 0$ on $S^*M - (\mathcal{W}^u \cup \mathcal{W}^s)$, $\tilde{m}(x, \xi) > 1 - \epsilon$ for $(x, \xi) \in \mathcal{W}^s$ and $\tilde{m}(x, \xi) < \epsilon$ for $(x, \xi) \in \mathcal{W}^u$.

Proof. Let us recall the main lines of the proof of this Lemma which relies only on the compactness and on the attracting properties of Σ_u and Σ_s . First, we have to verify that, up to shrinking V_u and V_s a little bit, $V^u \cap V^s = \emptyset$,

(44)
$$\forall t \geq 0, \ \tilde{\Phi}_f^t(V^s) \subset V^s, \text{ and } \tilde{\Phi}_f^{-t}(V^u) \subset V^u.$$

This follows from Lemmas 3.6 and 3.8. Once we have this property, we can follow the proof of [32]. More precisely, we know that

$$\mathcal{I}(x,\xi) := \{ t \in \mathbb{R} : \ \tilde{\Phi}_f^t(x,\xi) \in S^*M - (V^u \cup V^s) \},$$

is a closed, connected interval whose length is uniformly bounded by some constant $\tau > 0$. We then set T > 0 such that $\tau/(2T) < \epsilon$ satisfying

$$\mathcal{W}^u := \tilde{\Phi}_f^{-T}(S^*M - V^s) \subset V^u, \ \mathcal{W}^s := \tilde{\Phi}_f^{-T}(S^*M - V^u) \subset V^s.$$

Once these parameters are fixed, one just has to verify that, if $m_0 \in \mathcal{C}^{\infty}(S^*M, [0, 1])$ is equal to 1 on V^s and to 0 on V^u , then the function

$$m_T(x,\xi) := \frac{1}{2T} \int_{-T}^T m_0 \circ \tilde{\Phi}_f^t(x,\xi) dt$$

satisfies the assumption of the Lemma – see [32] for details.

We now use this Lemma with V^u , V^s and $\epsilon > 0$ small enough (to be precised). Thus, we have a function $\tilde{m}(x,\xi)$ defined on S^*M . We introduce a smooth function m_1 defined on T^*M which satisfies

$$m_1(x,\xi) = N_1 \tilde{m}\left(x, \frac{\xi}{\|\xi\|_x}\right) - N_0\left(1 - \tilde{m}\left(x, \frac{\xi}{\|\xi\|_x}\right)\right), \text{ for } \|\xi\|_x \ge 1,$$

and

$$m_1(x,\xi) = 0$$
, for $\|\xi\|_x \le \frac{1}{2}$.

We set the order function of our escape function to be

$$m(x,\xi) = -f(x) + m_1(x,\xi).$$

Set now

$$\tilde{\Gamma}_{\mp} := \left\{ (x, \xi) \in T^*M : \xi \neq 0 \text{ and } \frac{\xi}{\|\xi\|_x} \in \mathcal{W}^{s/u} \right\}.$$

From the definition of m, $\tilde{\Gamma}_{-}$ (resp. $\tilde{\Gamma}_{+}$) is a small conical neighborhood of Γ_{-} (resp. of Γ_{+}). Moreover, for every (x, ξ) in $\tilde{\Gamma}_{-}$ (resp. $\tilde{\Gamma}_{+}$) satisfying $\|\xi\|_{x} \geq 1$, one has

$$m(x,\xi) \le -N_0(1-\epsilon) + N_1\epsilon + ||f||_{\mathcal{C}^0} \quad (\text{resp.} \ge N_1(1-\epsilon) - N_0\epsilon - ||f||_{\mathcal{C}^0}).$$

If we choose ϵ small enough, then the first items of Lemma 4.1 are proved. We now set the following escape function:

$$G_m(x,\xi) = m(x,\xi)\log(1+||\xi||_x^2),$$

and we have to compute the derivative X_{H_f} . G_m of G_m along the Hamiltonian vector field X_{H_f} associated with H_f . Note that

(45)
$$X_{H_f}.G_m(x,\xi) = \log(1 + \|\xi\|_x^2) X_{H_f}.m(x,\xi) + m(x,\xi) \frac{X_{H_f}.\|\xi\|_x^2}{1 + \|\xi\|_x^2}.$$

Let r > 0 be a small parameter. We shall estimate the derivative of G_m along the Hamiltonian function in $T^*B(a,r)$ for every critical points and in the complementary of this set.

Let us start with the case where (x,ξ) belongs to T^*M_{reg} where M_{reg} is the complementary set of $\bigcup_{a \in \text{Crit} f} B(a,r/2)$. In that case, we fix $R_0 > 1$. Then, there exists C_g depending only on the Riemannian metric and on f such that, for every (x,ξ) in T^*M satisfying $\|\xi\|_x \geq R_0$,

$$X_{H_f}.G_m(x,\xi) \le -X_{H_f}.f(x)\log(1+R_0^2) + C_g(N_0+N_1+||f||_{\mathcal{C}^0}),$$

as $X_{H_f}.m_1 \leq 0$ for $\|\xi\| > 1$ according to Lemma A.1. As x is far from the critical points of f, one knows from (4) that there exists a constant c(r) > 0 (depending only on r > 0) such that $X_{H_f}.f(x) \geq c(r)$. In particular, one has

$$X_{H_f}.G_m(x,\xi) \le -c(r)\log(1+R_0^2) + C_g(N_0+N_1+\|f\|_{\mathcal{C}^0}) \le -\min\{N_0,N_1\},$$

where the last equality holds if we choose $R_0 > 1$ large enough (in a way that depends on r, N_0 and N_1).

It now remains to analyse the behaviour in a "neighborhood" of a critical point a in Crit f. In that case, we can as in [32] make use of the (local) hyperbolic structure of the flow. We fix (x, ξ) in T^*M such that $\|\xi\|_x \ge 1$ and x in B(a, r), and we use (4) to write

$$X_{H_f}.G_m(x,\xi) \le X_{H_f}.m_1(x,\xi)\log(1+\|\xi\|_x^2) + m(x,\xi)\frac{X_{H_f}.\|\xi\|_x^2}{1+\|\xi\|_x^2}.$$

We now distinguish three cases:

• Suppose that we could show that, if (x, ξ) belongs to $\tilde{\Gamma}_-$ and $\|\xi\| > 1$, then, one can find a constant¹¹ $c_- > 0$ depending only on f and g such that

(46)
$$X_{H_f} \cdot (\|\xi\|_x^2) > c_- \|\xi\|_x^2.$$

In particular, one could infer

$$X_{H_f}.G_m(x,\xi) \le -\frac{N_0}{2}c_-,$$

where we used Lemma A.1 to bound $X_{H_f}.m_1(x,\xi)$.

• Suppose that we could show that, if (x,ξ) belongs to $\tilde{\Gamma}_+$ and $\|\xi\| > 1$, then, one can find a constant $c_+ > 0$ depending only on f and g such that

(47)
$$X_{H_f} \cdot (\|\xi\|_x^2) < -c_+ \|\xi\|_x^2.$$

In particular, one could infer

$$X_{H_f}.G_m(x,\xi) \le -\frac{N_1}{2}c_+,$$

where we used again Lemma A.1 to bound $X_{H_f}.m_1(x,\xi)$.

• If (x,ξ) does not belong to $\tilde{\Gamma}_- \cup \tilde{\Gamma}_+$, then $\frac{\xi}{\|\xi\|}$ belongs to $S^*M - (\mathcal{W}^u \cup \mathcal{W}^s)$, and, by Lemma A.1, one finds

$$X_{H_f}.G_m(x,\xi) \le -\eta(N_0+N_1)\log(1+\|\xi\|^2) + C_g(N_0+N_1+\|f\|_{\mathcal{C}^0}).$$

Thus, if we choose $R_0 > 1$ large enough (in a way that depends on N_0, N_1), then one can ensure that $X_{H_f}.G_m(x,\xi) \leq -\min\{N_0,N_1\}$ whenever $\|\xi\| \geq R_0$ on this set.

This concludes the second part of the Lemma except for (46) and (47) that are still to be proved. The proof is similar in both cases and we will only treat the first case. We note that the compactness of Σ_u and Σ_s will one more time play a crucial role in the proof.

We start with the case where $(x,\xi) = (a,\xi)$ belongs to $E_s^*(a)$. In that case, we can make use of the fact that we have a smooth linearizing chart – see paragraph 3.5. Recall also that the linearized vector field $L_g(a)$ is diagonalizable in a basis of eigenvectors which is orthogonal for the metric $g^*(a)$ – see paragraph 3.3. This implies that, in the adapted coordinates (z,ζ) around a,

$$X_{H_f}.(\|\zeta\|_z^2) = 2\sum_{j=1}^r |\chi_j(a)|\zeta_j^2 + \mathcal{O}_{f,g}(\|\zeta\|_a^2),$$

where the constant in the remainder depends only on f and g and $g^*(a)$ is the induced Riemannian metric on T_a^*M . In particular, provided that we take some small enough constant $c_- = c_-(f, g)$ depending on g and f, inequality (46) holds in the case where $(x, \xi) = (a, \xi)$ belongs $E_s^*(a)$. We then define

$$U_r := \{(x,\xi) \in \bigcup_{y \in B(a,r)} T_y^* M : \|\xi\|_x > 1, \text{ and (46) holds with } c_- = 2c_-(f,g)\},$$

¹¹We note that we may have to take V^s small enough.

which is an open set in $T^*M - (M \times \{0\})$. Then, we have to prove that we can choose V_s small enough to ensure that the neighborhood

$$\tilde{V}_s^{(r)} := \{(x,\xi) \in \bigcup_{y \in B(a,r)} T_y^* M : ||\xi|| > 1 \text{ and } (x,\xi/||\xi||_x) \in V_s\}$$

is contained in U_r . We proceed by contradiction, and we suppose that, for every r>0 small enough, there exists $m_0 \geq 1$ such that, for any $m \geq m_0$ and for any neighborhood V_m of Σ_s of size 1/m, one can find $(x_m^{(r)}, \xi_m^{(r)}) \notin U_r$ belonging to $\tilde{V}_m^{(r)}$. Without loss of generality, we can suppose that $\|\xi_m^{(r)}\| \leq 2$. By compactness, we can then extract a subsequence such that $\lim_{m\to+\infty}(x_m^{(r)},\xi_m^{(r)})=(x^{(r)},\xi^{(r)})$. Moreover, as Σ_s is compact, $(x^{(r)},\xi^{(r)})$ belongs to $\bigcup_{y\in\overline{B(a,r)}}(\Gamma_-\cap T_y^*M)$, and, by construction of the sequence, (46) does not hold at this point with the constant $c_-=2c_-(f,g)$. This holds for any r>0 small enough. We now extract a converging subsequence as $r\to 0^+$, and we find a point (x,ξ) in $E_s^*(a)$ where we know that (46) holds with the constant $c_-=c_-(f,g)$. This gives the expected contradiction as $\xi\neq 0$.

Appendix B. Proof of Proposition 4.2

The proof of this Proposition was given in great details in [32, Th. 1.4] for the case k=0. The adaptation to the case $0 \le k \le n$ is almost identical except that we have to deal with pseudodifferential operators with values in $\Lambda^k(T^*M)$. The main point is that the (pseudodifferential) operators under consideration have a diagonal symbol. In fact, given any local basis $(e_j)_{j=1,...J_k}$ of $\Lambda^k(T^*M)$ and any family $(u_j)_{j=1,...J_k}$ of smooth functions $\mathcal{C}^{\infty}(M)$, one has

$$\mathcal{L}_{V_f}^{(k)} \left(\sum_{j=1}^{J_k} u_j e_j \right) = \sum_{j=1}^{J_k} \mathcal{L}_{V_f}(u_j) e_j + \sum_{j=1}^{J_k} \mathcal{L}_{V_f}^{(k)}(e_j) u_j,$$

where the second part of the sum in the right-hand side is a lower order term (of order 0). This diagonal form allows to adapt the proofs of [32] to this vector bundle framework. For completeness, we briefly recall the main lines of the proof and just point a (minor) simplification due to the particular form of our flow. To make the comparison with that reference simpler, we shall consider the operator $-i\mathcal{L}_{V_f}$ instead of $-\mathcal{L}_{V_f}$.

Remark B.1. As was already mentionned, the case of currents was treated by Dyatlov and Zworski in [28] via a slightly different approach. Their method could also probably be adapted to deal with the case of Morse-Smale gradient flows.

The strategy is to consider the equivalent operator

(48)
$$\widehat{\mathcal{L}}_f^{(k)} := \operatorname{Op}(\mathbf{A}_m^{(k)}) \circ (-i\mathcal{L}_{V_f}^{(k)}) \circ \operatorname{Op}(\mathbf{A}_m^{(k)})^{-1},$$

and to begin with, we recall the following result [32, Lemma 3.2]:

Lemma B.2. The operator

$$\operatorname{Op}(\mathbf{A}_m^{(k)}) \circ (-i\mathcal{L}_{V_f}^{(k)}) \circ \operatorname{Op}(\mathbf{A}_m^{(k)})^{-1} + i\mathcal{L}_{V_f}^{(k)}$$

is a pseudodifferential operator in $\Psi^{+0}(M)$ whose symbol in any given system of coordinates is of the form

$$P(x,\xi) = i(X_{H_f}.G_m)(x,\xi)\mathbf{Id} + \mathcal{O}(S^0) + \mathcal{O}_m(S^{-1+0}).$$

In this Lemma, the notation $\mathcal{O}(.)$ means that the remainder is independent of the order function m, while the notation $\mathcal{O}_m(.)$ means that it depends on m. In particular, this Lemma says that $\widehat{\mathcal{L}}_f^{(k)}$ is an element in $\Psi^1(M, \Lambda^k(T^*M))$. Then, combining this Remark to [32, Lemma A.1] which can be adapted directly to the case of operators with values in a vector bundle, one finds that $\widehat{\mathcal{L}}_f^{(k)}$ has a unique closed extension as an unbounded operator on $L^2(M, \Lambda^k(T^*M))$. This shows the first part of the Proposition in the case k=0.

Remark B.3. The proof of this Lemma was given for k = 0 in [32] and the adaptation to the case $1 \le k \le n$ follows from the diagonal structure of the operators involved. Let us recall that the key idea is to observe, by linearizing the exponential,

$$\operatorname{Op}(\mathbf{A}_{m}^{(k)}) \circ (-i\mathcal{L}_{V_{f}}^{(k)}) \circ \operatorname{Op}(\mathbf{A}_{m}^{(k)})^{-1} \simeq (1 + \operatorname{Op}(G_{m}) + \ldots) \circ (-i\mathcal{L}_{V_{f}}^{(k)}) \circ (1 - \operatorname{Op}(G_{m}) + \ldots)$$

$$= -i\mathcal{L}_{V_{f}}^{(k)} + [\operatorname{Op}(G_{m}\mathbf{Id}), -i\mathcal{L}_{V_{f}}^{(k)}] + \ldots,$$

which implies via symbolic calculus

$$\operatorname{Op}(\mathbf{A}_m^{(k)}) \circ (-i\mathcal{L}_{V_f}^{(k)}) \circ \operatorname{Op}(\mathbf{A}_m^{(k)})^{-1} \simeq -i\mathcal{L}_{V_f}^{(k)} + i\operatorname{Op}(X_{H_f}.G_m\mathbf{Id}) + \dots$$

Then, up to the fact that we have to deal with $L^2(M, \Lambda^k(T^*M))$, the second part of Proposition 4.2 is exactly the content of Lemma 3.3 of [32] which only makes use of the properties of the escape function given in Lemma 4.1. We also note that they implicitely shows that, for every z in \mathbb{C} satisfying $\text{Im} z > C_0$, one has

(49)
$$\left\| \left(\widehat{\mathcal{L}}_f^{(k)} - z \right)^{-1} \right\|_{L^2(M,\Lambda^k(T^*M)) \to L^2(M,\Lambda^k(T^*M))} \le \frac{1}{\operatorname{Im}(z) - C_0}.$$

Remark B.4. Combining Proposition 4.2 to the Hille-Yosida Theorem [30, Cor. 3.6, p. 76], one knows that (by conjugation)

(50)
$$(\varphi_f^{-t})^* : \mathcal{H}_k^m(M) \to \mathcal{H}_k^m(M),$$

generates a strongly continuous semigroup which is defined for every $t \geq 0$ and whose norm is bounded by e^{tC_0} .

Finally, the last part of the Proposition is based on results from analytic Fredholm theory. It is in fact the only place where things differ with [32]. The situation is in fact slightly simpler here as we shall now briefly explain it. We write

$$\widehat{\mathbf{V}}_f^{(k)} := \frac{i}{2} \left(\left(\widehat{\mathcal{L}}_f^{(k)} \right)^* - \widehat{\mathcal{L}}_f^{(k)} \right).$$

We denote by $\mathbf{V}_f^{(k)}(x,\xi)$ the symbol of this operator. Note that, from [32, Lemma A.1], $(\widehat{\mathcal{L}}_{V_f}^{(k)})^*$ also has a unique closed extension to $L^2(M,\Lambda^k(T^*M))$. Combining Lemma B.2 to

Lemma 4.1, one knows that, for every (x, ξ) in T^*M ,

$$\mathbf{V}_f^{(k)}(x,\xi) \le (-C_N + C)\mathbf{Id} + \mathcal{O}_m(S^{-1+0}),$$

for some constant C > 0 which is independent of m and for the constant C_N defined in Lemma 4.1. From the sharp Gårding inequality, one can deduce that, for every $0 < \mu < 1$, there exists a constant $C_{\mu,m} > 0$ such that, for every u in $C^{\infty}(M)$

$$\langle (\widehat{\mathbf{V}}_f^{(k)} + C_N - C)u, u \rangle_{L^2(M, \Lambda^k(T^*M))} \leq C_{\mu, m} \|u\|_{H^{\frac{\mu-1}{2}}(M, \Lambda^k(T^*M))}^2,$$

where the remainder $\mathcal{O}_m(S^{-1+0})$ has been absorbed in the RHS thanks to the Calderón-Vaillancourt Theorem. From this inequality, one can deduce that

$$\langle (\widehat{\mathbf{V}}_f^{k)} + C_N - C \rangle u, u \rangle_{L^2(M, \Lambda^k(T^*M))} \leq \left\langle \widetilde{C}_{\mu, m} \left(1 - \Delta_g^{(k)} \right)^{\frac{\mu - 1}{2}} u, u \right\rangle_{L^2(M, \Lambda^k(T^*M))},$$

where $\Delta_g^{(k)}$ is the Laplace-Beltrami operator acting on k differential forms. We define then

$$\widehat{\chi}_k := \widetilde{C}_{\mu,m} \left(1 - \Delta_g^{(k)} \right)^{\frac{\mu - 1}{2}} \in \Psi^{\mu - 1}(M, \Lambda^k(T^*M)),$$

which is a compact operator as $\mu - 1 < 0$. Hence, we can rewrite the last inequality as

$$\langle (\widehat{\mathbf{V}}_f^{(k)} - \widehat{\chi}_k + C_N - C)u, u \rangle_{L^2(M, \Lambda^k(T^*M))} \le 0,$$

from which one can $deduce^{12}$ that the resolvent

$$\left(\widehat{\mathcal{L}}_f^{(k)} - i\widehat{\chi}_k - z\right)^{-1}$$

defines a bounded operator from $L^2(M, \Lambda^k(T^*M))$ to itself as soon as $\operatorname{Im}(z) > -(C_N - C)$. From the compactness of $\widehat{\chi}_k$ we can deduce that $\widehat{\chi}_k \left(\widehat{\mathcal{L}}_f^{(k)} - i\widehat{\chi}_k - z\right)^{-1}$ is also a compact operator which is exactly the content of Lemma 3.4 in [32]. The conclusion then follows by a classical argument from analytic Fredholm theory given in [32, Lemma 3.5].

APPENDIX C. ASYMPTOTIC EXPANSIONS

In this appendix, we review some classical facts on asymptotic expansions (see [51, Chapter 1] for a nice review).

Definition C.1. Let \mathcal{I} be a discrete¹³ countable subset of \mathbb{R} bounded from below. We call \mathcal{I} the **index set**. Then $h \in C^{\infty}((0,1],\mathbb{R})$ has polyhomogeneous asymptotic expansion indexed by \mathcal{I} if

$$\exists (a_{\lambda})_{\lambda \in \mathcal{I}} \text{ such that } \forall \Lambda \in \mathbb{R} \setminus \mathcal{I}, \exists C \geqslant 0, \forall s \in (0, 1]$$

$$\left| h(s) - \sum_{\lambda \in \mathcal{I}, \lambda \leqslant \Lambda} a_{\lambda} s^{\lambda} \right| \leqslant C s^{\lambda_0}$$

where $\lambda_0 = \inf\{\lambda \in \mathcal{I} \cap [\Lambda, +\infty)\}.$

 $^{^{12}}$ The proof of this fact is similar to the proof of Lemma 3.3 in [32].

¹³We mean that it has no accumulation point.

The key property we use in this article is:

Proposition C.2. If such an asymptotic expansion exists then it is unique.

Proof. Assume that h has two asymptotic expansions $\sum_{\lambda \in \mathcal{I}} a_{\lambda} s^{\lambda}$ and $\sum_{\lambda \in \mathcal{J}} b_{\lambda} s^{\lambda}$. Note that, up to taking the union of two index sets, we can make the assumption that both expansions are with respect to the same set, i.e. $\mathcal{I} = \mathcal{J}$. Suppose by contradiction that there exists $\lambda \in \mathcal{I}$ such that $a_{\lambda} \neq b_{\lambda}$, and pick the smallest element enjoying this property. Let us call this element λ_0 . Then the approximation property gives for every $\Lambda \in \mathbb{R} \setminus \mathcal{I}$ sufficiently large and as $s \to 0^+$,

$$\left| \sum_{\lambda \in \mathcal{I}, \lambda \leq \Lambda} a_{\lambda} s^{\lambda} - \sum_{\lambda \in \mathcal{I}, \lambda \leq \Lambda} b_{\lambda} s^{\lambda} \right| = o(s^{\Lambda}).$$

However $|\sum_{\lambda \in \mathcal{I}, \lambda \leqslant \Lambda} a_{\lambda} s^{\lambda} - \sum_{\lambda \in \mathcal{I}, \lambda \leqslant \Lambda} b_{\lambda} s^{\lambda}| = |a_{\lambda_0} - b_{\lambda_0}| s^{\lambda_0} (1 + o(1))$, which contradicts the above upper bound.

Now we have the following

Lemma C.3. Let $(\lambda_i)_{i=1}^n$ be a collection of n positive real numbers. Let \mathcal{I} be the index set defined as

$$\mathcal{I} := \left\{ \sum_{j=1}^{n} k_j \lambda_j : \ \forall 1 \leq j \leq n, \ k_j \in \mathbb{N} \right\} \subset \mathbb{R}.$$

Then, for all $\psi \in C^{\infty}(\mathbb{R}^n)$, the function

$$s \in (0,1] \longmapsto \psi(s^{\lambda_1},\dots,s^{\lambda_n})$$

has polyhomogeneous asymptotic expansion indexed by \mathcal{I} .

Proof. Use the fact that ψ has a Taylor expansion in $\mathbf{s} = (s_1, \dots, s_n)$ with remainder, in multi-index notations:

$$\psi(\mathbf{s}) = \sum_{|\alpha| \le k} \frac{\mathbf{s}^{\alpha}}{\alpha!} \partial_{\mathbf{s}}^{\alpha} \psi(0) + \mathcal{O}_f((|s_1| + \ldots + |s_n|)^{k+1}).$$

Then, making the substitution $s_i = s^{\lambda_i}$ yields an asymptotic expansion of the form

$$\psi(s^{\lambda_1}, \dots, s^{\lambda_n}) = \sum_{|\alpha| \le k} \frac{s^{\lambda_1 \alpha_1 + \dots + \lambda_n \alpha_n}}{\alpha!} \partial_{\mathbf{s}}^{\alpha} \psi(0) + \mathcal{O}_{\psi}((s^{\lambda_1} + \dots + s^{\lambda_n})^{k+1}).$$

If we fix $\Lambda > 0$ and k such that $(k+1)\min_i(\lambda_i) > \Lambda$, one can verify from the above expression that $\psi(s^{\lambda_1}, \ldots, s^{\lambda_n})$ has a polyhomogeneous expansion in \mathcal{I} .

A function $h \in C^{\infty}((0,1],\mathbb{R})$ is said to be **weakly homogeneous** if

$$\exists C > 0, \exists d \in \mathbb{R}, \forall s \in (0, 1], |h(s)| \leqslant Cs^d.$$

Recall that the Mellin transform of $h\mathbf{1}_{[0,1]}$ for f weakly homogeneous is then defined as

(51)
$$\mathcal{M}\left(h\mathbf{1}_{[0,1]}\right)(z) = \int_0^1 h(s)s^z \frac{ds}{s},$$

and that it is holomorphic on the half-plane Re(z) > -d. Finally, we note that the following holds:

Lemma C.4. Under the above conventions, one has:

- (1) For w in \mathbb{C} , the Mellin transform $\mathcal{M}(s^w\mathbf{1}_{[0,1]}(s))(z)$ equals $\frac{1}{w+z}$ and thus, it extends
- meromorphically with a simple pole at z = -w. (2) For every polyhomogeneous h where $h \sim \sum_{\lambda \in \mathcal{I}} a_{\lambda} s^{\lambda}$, the Mellin transform $\mathcal{M}\left(h\mathbf{1}_{[0,1]}\right)(z)$ extends meromorphically to the complex plane with simple poles at $z \in -\mathcal{I}$.

Proof. This is a particular case of [51, Thm 3.1 p. 11]

References

- [1] D.V. Anosov, Geodesic flows on closed Riemannian manifolds of negative curvature, Trudy Mat. Inst. Steklov. **90** (1967)
- [2] J.F. Arnoldi, F. Faure, T. Weich, Asymptotic spectral gap and Weyl law for Ruelle resonances of open partially expanding maps, preprint arXiv:1302.3087, to appear in Erg. Th. Dyn. Syst. (2015)
- [3] M. F. Atiyah, R. Bott, A Lefschetz Fixed Point Formula for Elliptic Complexes I, Ann. of Math., Vol. 86 (1967), 374–407
- [4] V. Baladi, M. Demers, C. Liverani, Exponential Decay of Correlations for Finite Horizon Sinai Billiard Flows, preprint arXiv:1506.02836 (2015)
- [5] V. Baladi, C. Liverani, Exponential decay of correlations for piecewise cone hyperbolic contact flows, Comm. Math. Phys. 314 (2012), no. 3, 689–773
- [6] V. Baladi, M. Tsujii, Anisotropic Hölder and Sobolev spaces for hyperbolic diffeomorphisms, Ann. Inst. Fourier (Grenoble) 57 (2007) 127–154
- [7] V. Baladi, M. Tsujii, Dynamical determinants and spectrum for hyperbolic diffeomorphisms, Contemp. Math. 469 (Amer. Math. Soc.), Volume in honour of M. Brin's 60th birthday (2008), 29-68
- [8] O. Bandtlow, W. Just, J.Slipantschuk, Spectral structure of transfer operators for expanding circle maps, Ann. IHP (C) Non Linear Analysis, in press (2015)
- [9] O. Bandtlow, F. Naud, Lower bounds for the Ruelle spectrum of analytic expanding circle maps, preprint arXiv:1605.06247 (2016)
- [10] J.M. Bismut, W. Zhang, An Extension of a Theorem of Cheeger and Müller, Astérisque 205, Société Math. de France, Paris (1992)
- [11] M. Blank, G. Keller, C. Liverani, Ruelle-Perron-Frobenius spectrum for Anosov maps, Nonlinearity 15 (2002), 1905–1973
- [12] R. Bott, Morse theory indomitable, Publ. Math. IHES 68 (1988), 99–114
- [13] R. Bowen, D. Ruelle, The ergodic theory of Axiom A flows, Invent. Math. 29 (1975), no. 3, 181–202
- [14] R. Bowen *Notebook*, avalable at https://bowen.pims.math.ca/
- [15] C. Brouder, N.V. Dang, F. Hélein, Boundedness and continuity of the fundamental operations on distributions having a specified wave front set. (with a counter example by Semyon Alesker), Studia Mathematica (2016)
- [16] D. Burghelea, L. Friedlander and T. Kappeler, Asymptotic expansion of the Witten deformation of the analytic torsion Journal of Functional Analysis, 137(2), 320-363
- [17] D. Burghelea, Lectures on Witten-Helffer-Sjöstrand theory (Proceedings of the third international workshop in geometry and applications and the first German-Romanian seminar in Geometry, Sibiu, 1997), Astra Association, 85-99, (preprint Schrödinger Institute, No 572)
- [18] D. Burghelea, S. Haller, Dynamics, Laplace transform and spectral geometry, Journal of Topology, LMS, vol 1, 2008, 115-151
- [19] O. Butterley, C. Liverani, Smooth Anosov flows: correlation spectra and stability, J. Mod. Dyn. 1, 301-322 (2007)

- [20] K.T. Chen, Equivalence and decomposition of vector fields about an elementary critical point, Amer. J. Math 85 (1963), 693–722
- [21] N.V. Dang, The Extension of Distributions on Manifolds, a Microlocal Approach, Ann. H. Poincaré 17 (2016), 819–859
- [22] N.V. Dang, G. Rivière, Equidistribution of the conormal cycle of random nodal sets, preprint arXiv:1509.00228, to appear in J. Eur. Math. Soc. (2015)
- [23] K. Datchev, S. Dyatlov, M. Zworski, Sharp polynomial bounds on the number of Pollicott-Ruelle resonances, Ergodic Theory Dynam. Systems 34 (2014), no. 4, 1168–1183
- [24] M. Demers, H.K. Zhang, Spectral analysis of the transfer operator for the Lorentz Gas, Journal of Modern Dynamics 5:4 (2011), 665–709
- [25] D. Dolgopyat, On the decay of correlations in Anosov flows, Ann. of Math. Vol. 147 (1998), 357–390
- [26] S. Dyatlov, F. Faure, C. Guillarmou, Power spectrum of the geodesic flow on hyperbolic manifolds, Anal. PDE 8 (2015), no. 4, 923–1000
- [27] S. Dyatlov, C. Guillarmou, *Pollicott-Ruelle resonances for open systems*, preprint arXiv:1410.5516, to appear Ann. H. Poincaré (2014)
- [28] S. Dyatlov, M. Zworski, Dynamical zeta functions for Anosov flows via microlocal analysis, arXiv:1306.4203, to appear Ann. ENS (2013)
- [29] S. Dyatlov, M. Zworski, *Mathematical theory of scattering resonances*, avalable at http://math.mit.edu/~dyatlov/res/res.pdf (2016)
- [30] K.J. Engel, R. Nagel, One-Parameter Semigroups for Linear Evolution Equations, Grad. Texts in Math. 194, Springer-Verlag New York (2000)
- [31] F. Faure, N. Roy, J. Sjöstrand, Semi-classical approach for Anosov diffeomorphisms and Ruelle resonances, Open Math. Journal, vol. 1, 35–81, (2008)
- [32] F. Faure, J. Sjöstrand, Upper bound on the density of Ruelle resonances for Anosov flows, Comm. in Math. Physics, vol. 308, 325–364, (2011)
- [33] F. Faure, M. Tsujii, The semiclassical zeta function for geodesic flows on negatively curved manifolds, Arxiv arXiv:1311.4932 (2013)
- [34] F. Faure, M. Tsujii, Prequantum transfer operator for symplectic Anosov diffeomorphism, Astérisque Vol. 375, Société Math. de France (2015)
- [35] F. Faure, T. Weich, Asymptotic spectral gap for open partially expanding maps, preprint arXiv:1504.06728 (2015)
- [36] E. Frenkel, A. Losev and N. Nekrasov, *Instantons beyond topological theory. I*, Journal of the Institute of Mathematics of Jussieu, 2011, vol. 10, no 03, p. 463-565
- [37] E. Frenkel, A. Losev and N. Nekrasov, Notes on instantons in topological field theory and beyond, Nuclear Physics B-Proceedings Supplements, 2007, vol. 171, p. 215-230
- [38] C. Gérard, J. Sjöstrand, Semiclassical Resonances Generated by a Closed Trajectory of Hyperbolic Type, Comm. in Math. Physics, vol. 108, 391–421, (1987)
- [39] I.M. Gelfand, M. Kapranov, and A. Zelevinsky, Discriminants, resultants, and multidimensional determinants, Springer, 2008
- [40] P. Giulietti, C. Liverani, M. Pollicott, Anosov flows and dynamical zeta functions, Ann. of Math. (2) 178 (2013), no. 2, 687–773
- [41] S. Gouëzel, C. Liverani Compact locally maximal hyperbolic sets for smooth maps: fine statistical properties, J. Diff. Geometry 79 (2008) 433–477
- [42] S. Gouëzel, Spectre du flot géodésique en courbure négative [d'après F. Faure et M. Tsujii], Séminaire Bourbaki (2015)
- [43] V. Guillemin, S. Sternberg, Supersymmetry and equivariant de Rham theory, Springer-Verlag Berlin Heidelberg (1999)
- [44] P. Hartman, A Lemma in the theory of structural stability of differential equations. Proc. A.M.S. 11 (4) (1960), 610–620

- [45] F.R. Harvey, H.B. Lawson, Morse theory and Stokes Theorem, Surveys in Diff. Geom. VII (2000), 259–311
- [46] F.R. Harvey, H.B. Lawson, Finite volume flows and Morse theory, Ann. of Math. Vol. 153 (2001), 1–25
- [47] B. Helffer, J. Sjöstrand, Points multiples en mécanique semi-classique IV, étude du complexe de Witten, CPDE 10 (1985), 245–340
- [48] B. Helffer, J. Sjöstrand, Résonances en limite semi-classique, Mém. Soc. Math. France 24-25 (1986), 1–228
- [49] G. Henniart, Les inégalités de Morse. Séminaire Bourbaki, 26 (1983–1984), Exposé No. 617
- [50] L. Hörmander, The Analysis of Linear Partial Differential Operators I. Distribution Theory and Fourier Analysis, second ed., Springer Verlag, Berlin, 1990
- [51] J.I. Igusa, Lectures on forms of higher degree, Bombay: Tata Institute of Fundamental Research, 1978.
- [52] P. Koroteev, Morse Theory in Field Theory, Proceedings of the 8th International Conference on Geometry, Integrability and Quantization. Institute of Biophysics and Biomedical Engineering, Bulgarian Academy of Sciences, 2007
- [53] F. Laudenbach, On the Thom-Smale complex, in An Extension of a Theorem of Cheeger and Müller, by J.-M. Bismut and W. Zhang, Astérisque 205, Société Math. de France, Paris (1992)
- [54] F. Laudenbach, Transversalité, courants et théorie de Morse. Un cours de topologie différentielle, Éditions de l'École polytechnique, Palaiseau, 2012. x+182 pp
- [55] C. Liverani, On contact Anosov flows, Ann. of Math., Vol. 159, No. 3 (2004), 1275–1312
- [56] A. Losev, S. Slizovskiy, New observables in topological instantonic field theories, Journal of Geometry and Physics 61.10 (2011): 1868-1880
- [57] A. Losev, I. Polyubin, *Topological quantum mechanics for physicists*, Journal of Experimental and Theoretical Physics Letters 82.6 (2005): 335-342
- [58] G. Minervini, A current approach to Morse and Novikov theories, Rend. di Matematica, Serie VII, Vol. 36 (2015), 95–195
- [59] E. Nelson, *Topics in dynamics. I: Flows.*, Mathematical Notes. Princeton University Press, Princeton, N.J.; University of Tokyo Press, Tokyo (1969) iii+118 pp.
- [60] L.I. Nicolaescu, Tame flows, Memoirs of the AMS 208 (2010)
- [61] S. Nonnenmacher, M. Zworski, Decay of correlations for normally hyperbolic trapping, Invent. Math. 200 (2015), no. 2, 345–438
- [62] M. Pollicott, On the rate of mixing of Axiom A flows, Invent. Math. 81 (1985), no. 3, 413–426
- [63] G. de Rham, Differentiable manifolds: Forms, Currents, Harmonic Forms, Springer (1980)
- [64] D. Ruelle, Resonances for Axiom A flows, J. Differential Geom. 25 (1987), no. 1, 99–116
- [65] D. Ruelle, One-dimensional Gibbs states and Axiom A diffeomorphisms. J. Differential Geom. 25 (1987), no. 1, 117–137
- [66] L. Schwartz, Théorie des distributions, Hermann (1966)
- [67] J.G. Sinai, Gibbs measures in ergodic theory, Uspehi Mat. Nauk 27 (1972), no. 4 (166), 21–64
- [68] S. Smale, Morse inequalities for a dynamical system, Bull. Amer. Math. Soc., 66 (1960), 43–49.
- [69] S. Smale, On Gradient Dynamical Systems, Ann. of Math., Vol. 74, No. 1 (1961), 199–206
- [70] S. Smale, Differentiable dynamical systems, Bull. AMS 73, 747–817 (1967)
- [71] L. Stoyanov, Spectrum of the Ruelle operator and exponential decay of correlations for open billiard flows, Amer. J. Math. 123(4) (2001) 715–759
- [72] R. Thom, Sur une partition en cellules associée à une fonction sur une variété, CRAS 228 (1949), 973–975
- [73] M. Tsujii, Quasi-compactness of transfer operators for contact Anosov flows, Nonlinearity 23, 1495–1545 (2010)
- [74] M. Tsujii, Contact Anosov flows and the Fourier-Bros-Iagolnitzer transform, Erg. Th. Dyn. Sys. 32 (2012), no. 6, 2083–2118

- [75] J. Weber, The Morse-Witten complex via dynamical systems, Expo. Math. 24 (2006), 127–159.
- [76] E. Witten, Supersymmetry and Morse theory, J. Diff. Geom. 17 (1982), 661–692
- [77] W. Zhang, Lectures on Chern-Weil theory and Witten deformation (Vol. 4). World Scientific. (2001)
- [78] M. Zworski, Semiclassical analysis, Graduate Studies in Mathematics 138, AMS (2012).

INSTITUT CAMILLE JORDAN (U.M.R. CNRS 5208), UNIVERSITÉ CLAUDE BERNARD LYON 1, BÂTIMENT BRACONNIER, 43, BOULEVARD DU 11 NOVEMBRE 1918, 69622 VILLEURBANNE CEDEX *E-mail address*: dang@math.univ-lyon1.fr

LABORATOIRE PAUL PAINLEVÉ (U.M.R. CNRS 8524), U.F.R. DE MATHÉMATIQUES, UNIVERSITÉ LILLE 1, 59655 VILLENEUVE D'ASCQ CEDEX, FRANCE

E-mail address: gabriel.riviere@math.univ-lille1.fr